DIAGNOSTIC EXPOSURE OF IONIZING RADIATION AND ITS LONG-TERM EFFECTS

DIAGNOSTIC EXPOSURE OF IONIZING RADIATION AND ITS LONG-TERM EFFECTS

By:

JAMES MCEVOY, BMedSci, BS(Hons)





A Thesis Submitted to Flinders University for the degree of Doctor of Philosophy, Cotutelle de Thesis

College of Medicine and Public Health, Flinders University

Department of Biology, McMaster University

© Copyright by James McEvoy, 28th May 2020, Doctor of Philosophy (Cotutelle)

Ph.D. Thesis – J. McEvoy McMaster University – Biology Flinders University – CoMPH.

Doctor of Philosophy, Cotutelle de Thesis

College of Medicine and Public Health – Flinders University, Adelaide, South Australia Department of Biology – McMaster University, Hamilton, Ontario

TITLE: Low dose radiation and the development of the respiratory system

AUTHOR: James McEvoy, BMedSci Flinders University, BS(Hons) Flinders University

CO-SUPERVISORS:

Associate Professor Dani-Louise Dixon

College of Medicine and Public Health Flinders University Adelaide, South Australia Professor Joanna Wilson

Department of Biology McMaster University Hamilton, Ontario

NUMBER OF PAGES: XXIV, 214

LAY ABSTRACT

Radiation is necessary in medicine to observe the internal structures of the body, but it can sometimes cause unwanted biological changes within the body. This risk is heightened when considering exposure to developing baby because of the dynamic changing it is naturally going through and possible lifetime left to experience effects. This thesis aimed to understand what levels of radiation patients receive in hospital, observing one population predicted to receive high levels (Intensive care patients) and one predicted to receive low levels (pregnant patients). Overall, the majority of patients in the two cohorts received less than the recommended yearly public limit of 1 millisievert (mSv). The second aim was to observe the effects on the growth, lungs and hearts of the babies in an animal model when they are exposed during pregnancy. Radiation had no overall effect on the lungs or heart but can reduce body weight at moderate (100 milligrays (mGy) and high (1000 mGy) exposures.

ABSTRACT

Medical radiation is vital in acquiring a patient diagnosis, but some clinicians are concerned with the perceived risks associated with ionizing radiation. This risk is heightened when incorporating *in utero* exposures due to the risk to the developing foetus. Although other organ systems have been studied, there is a paucity of data on the effects to the respiratory system from *in utero* exposures. The aim of this thesis was to understand the long-term effects on the respiratory system from *in utero* exposures, but as a first step, it was important to determine what levels patients receive whilst admitted to hospital. Two polar populations were chosen based on their predicted exposure levels during hospitalisation; one with high levels, intensive care unit (ICU) patients, and one with low levels, pregnant patients. Most patients cumulatively received < 1 mSv with median exposures of 0.99 mSv (ICU patients) and 0.02 mSv (pregnant patients). However, both cohorts had patients that surpassed 10 mSv. To assess the effects from *in utero* exposures on the respiratory system, two animal models were conducted both exposed during late gestation, one healthy model and one acute lung injury model. In the health animal model, cardiovascular outcomes were also measured, however, ionizing radiation (50, 300, 1000 mGy) did not appear to influence these two organ systems from the outcomes measured. In the acute lung injury model, lipopolysaccharide (3mg/Kg) stimulated an acute lung inflammatory response, however, there was also no overt effect of radiation from the outcomes measured (10, 100, 1000 mGy). In both models, ionizing radiation did cause growth restriction up to 16 weeks of age, but this was only observed from doses above 100 mGy. Overall, the levels of ionizing radiation patients receive is low and from diagnostic exposures during pregnancy, there does not appear to be any strong effects on the developing foetus.

DECLARATION

I certify that this thesis does not incorporate without acknowledgment any material previously submitted for a degree or diploma in any university and that to the best of my knowledge and belief, it does not contain any material previously published or written by another person except where due reference is made in the text.

James McEvoy

Date: 29/8/2019

ACKNOWLEDGEMENT

First and foremost, I would like to thank my parents Sheila and William McEvoy. Without your sacrifice and hard work, I would have never been in the position to begin, let alone complete, this monstrous task. You have supported me, tolerated me, taught me, and only wanted the best for me, for which I am utterly grateful. To my siblings, thank you for grounding me and taking my mind off daunting tasks. The fun and laughter that surrounded Sunday Dinners was a breath of fresh air in the ever-consuming PhD world. To my Sammybear, your behind the scenes support was endless. You have stood tall in every other part of my life keeping them upright and functioning. From making the dinner to driving me around, from organising friends catch up and to organising a wedding, to being the ear that forever listened or the smile the completely changed my day. Over this journey, you have always been there, usually with a glass of wine in your hand ready for me. You have provided the love and support that only a fiancé can provide. To all of my family, I love you and thank you.

To my combined lab family, who witnessed every emotion, thank you. My Flinders JETSS and McMaster Musketeers, you made my PhD what it is. You guys physically reinforced me, mentally supported me, experimentally guided me, fundamentally loved me, and ultimately changed me. As once said by James McEvoy, "We are shaped by those who surround us." A special mention must go to those who check in with me every day, Elena, Lana and Andrea – my fellow brethren. You understood everything that was going on and knew exactly what to say to validate my feelings. Mostly it was summed up with 'Same' or 'who do I need to hit'. This PhD would not, and could not, have been what it is without you three in my life. From onesie parties and late-night feasts to emotional breakdowns, inside jokes and of course Fridays at 4, I thank you from the bottom of my heart.

Thank you to my committee members from across the globe. To Assoc. Prof. Tony hooker and Assoc. Prof. Shailesh Bihari for your guidance in clinical and radiation scientific knowledge whilst in Australia. To Prof Douglas Boreham and Prof Grant McClelland, thank you for your outside perspective and teaching me the Canadian way of life. All of you have put many hours in helping me create this masterpiece and I truly appreciate your guidance and input.

I would like to acknowledge the generosity of the *Australian Government Research Training Program Scholarship* in providing fee-offset and stipend throughout my candidature. This scholarship was crucial in allowing full commitment to this thesis.

Last, but certainly not least, I would like to express my deepest gratitude to Assoc. Prof. Dani-Louise Dixon and Prof Joanna Wilson. These two beacons of light guided me down the garden path, with only a few pushes back on the track. They are the main reason this thesis is what it is, and they have both shaped me into the researcher I am. I am sure they won't miss my 'questions' or misspelt/mispronounced words that can only be found

viii

in the Dictionary of James. But truly, from the bottom of my heart, I really appreciate all and everything you have done for me in getting me here.

To the three simple phrases that got me through the past 3.5 years...

YOU GOT THIS,

WE BELIEVE IN YOU,

WORDS OF ENCOURAGEMENT

TABLE OF CONTENTS

| | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, | •••••• |
|--|--|--------|
| ABSTRACT | | IV |
| Declaratio | N | VI |
| Acknowled | GEMENT | VII |
| TABLE OF CC | NTENTS | x |
| LIST OF TABL | ES | xv |
| LIST OF FIGU | RES | XVI |
| LIST OF ABB | REVIATIONS | XVII |
| Declaratio | N OF ACADEMIC ACHIEVEMENT | xxII |
| 1 Снарт | er 1 Introduction | 1 |
| | | |
| 1.1 RA | DIATION | 1 |
| 1.1 RA 1.1.1 | DIATION Sources of ionizing radiation | 1 |
| 1.1 RAI 1.1.1 1.1.2 | DIATION Sources of ionizing radiation Doses of ionizing radiation | 1 |
| 1.1 RAI 1.1.1 1.1.2 1.2 Lov | DIATION Sources of ionizing radiation Doses of ionizing radiation N DOSE RADIATION | |
| 1.1 RAI 1.1.1 1.1.2 1.2 LOV 1.2.1 | DIATION Sources of ionizing radiation Doses of ionizing radiation V DOSE RADIATION Diagnostic modalities using ionizing radiation | |
| 1.1 RAI 1.1.1 1.1.2 1.2 LOV 1.2.1 1.2.2 | DIATION Sources of ionizing radiation Doses of ionizing radiation V DOSE RADIATION Diagnostic modalities using ionizing radiation Alternative diagnostic procedures | 1 |
| 1.1 RAI 1.1.1 1.2 1.2 LOV 1.2.1 1.2.2 1.3 RIS | DIATION Sources of ionizing radiation Doses of ionizing radiation V DOSE RADIATION Diagnostic modalities using ionizing radiation Alternative diagnostic procedures KS ASSOCIATED WITH LOW DOSES OF IONIZING RADIATION | |
| 1.1 RAI 1.1.1 1.2 1.2 LOV 1.2.1 1.2.2 1.3 Ris 1.3.1 | DIATION Sources of ionizing radiation Doses of ionizing radiation N DOSE RADIATION Diagnostic modalities using ionizing radiation Alternative diagnostic procedures Alternative diagnostic procedures Modelling radiation risk. | |
| 1.1 RAI 1.1.1 1.2 LOV 1.2.1 1.2.2 1.3 RIS 1.3.1 1.3.2 | DIATION Sources of ionizing radiation Doses of ionizing radiation N DOSE RADIATION Diagnostic modalities using ionizing radiation Alternative diagnostic procedures Alternative diagnostic procedures KS ASSOCIATED WITH LOW DOSES OF IONIZING RADIATION Modelling radiation risk. | |
| 1.1 RAI 1.1.1 1.2 Lov 1.2.1 1.2.2 1.3 Ris 1.3.1 1.3.2 1.3.3 | DIATION Sources of ionizing radiation Doses of ionizing radiation N DOSE RADIATION Diagnostic modalities using ionizing radiation Alternative diagnostic procedures Alternative diagnostic procedures Modelling radiation risk Professional understanding of ionizing radiation Decision making with ionizing radiation in clinical practice | |
| 1.1 RAI 1.1.1 1.2 Lov 1.2.1 1.2.2 1.3 Ris 1.3.1 1.3.2 1.3.3 1.4 PRI | DIATION Sources of ionizing radiation Doses of ionizing radiation N DOSE RADIATION Diagnostic modalities using ionizing radiation Alternative diagnostic procedures Alternative diagnostic procedures KS ASSOCIATED WITH LOW DOSES OF IONIZING RADIATION Modelling radiation risk Professional understanding of ionizing radiation Decision making with ionizing radiation in clinical practice | |
| 1.1 RAI 1.1.1 1.2 1.2 Lov 1.2.1 1.2.2 1.3 Ris 1.3.1 1.3.2 1.3.3 1.4 PRI 1.4.1 | DIATION Sources of ionizing radiation Doses of ionizing radiation N DOSE RADIATION Diagnostic modalities using ionizing radiation Alternative diagnostic procedures Alternative diagnostic procedures KS ASSOCIATED WITH LOW DOSES OF IONIZING RADIATION Modelling radiation risk Professional understanding of ionizing radiation Decision making with ionizing radiation in clinical practice GNANCY Prenatal care | |
| 1.1 RAI 1.1.1 1.2 Lov 1.2.1 1.2.2 1.3 Ris 1.3.1 1.3.2 1.3.3 1.4 PRI 1.4.1 1.4.2 | DIATION Sources of ionizing radiation Doses of ionizing radiation N DOSE RADIATION Diagnostic modalities using ionizing radiation Alternative diagnostic procedures Alternative diagnostic procedures Modelling radiation risk Professional understanding of ionizing radiation Decision making with ionizing radiation in clinical practice GNANCY Prenatal care Normal Development of the respiratory system | |

| 1.4.4 Foetal radiation exposure | 24 |
|--|----------|
| | |
| 1.4.5 Deterministic effects | 28 |
| 1.4.6 Stochastic effects | 34 |
| 1.5 Thesis goals | 38 |
| 1.6 CHAPTER SUMMARY | 40 |
| 1.7 References | 43 |
| 2 CHAPTER 2 CUMULATIVE RADIATION IN CRITICALLY ILL PATIENTS: A RETROSPECTIVE AUE | DIT |
| OF IONIZING RADIATION EXPOSURE IN AN INTENSIVE CARE UNIT | 52 |
| | <u></u> |
| | 52 ГЛ |
| | 54 гс |
| 2.2 METHODS | סס בכ |
| 2.2.1 Study conort | 50 |
| 2.2.2 Data collection | 50 |
| 2.2.3 Calculation of effective dose | 57 |
| 2.2.4 Statistics | 59 |
| 2.3 RESULTS. | 59 |
| 2.3.1 Distribution of procedure dose | 61 |
| 2.3.2 Patient cumulative radiation exposure | 63 |
| 2.4 Discussion | 66 |
| 2.5 CONCLUSION | /0 |
| 2.6 REFERENCES | 71 |
| 3 CHAPTER 3 CUMULATIVE RADIATION IN HOSPITALISED PREGNANT PATIENTS: | A |
| RETROSPECTIVE AUDIT OF IONIZING RADIATION EXPOSURE | /3 |
| SUMMARY | 73 |
| 3.1 INTRODUCTION | 75 |
| 3.2 Метнор | 77 |
| 3.2.1 Study cohort | 77 |
| 3.2.2 Data collection | 77 |
| 3.2.3 Calculation of effective dose | 78 |
| 3.2.4 Statistics | 80 |

| 3.3 | Resu | JLTS | 81 |
|--------------------------|----------------|---|-------------|
| 3.3 | 8.1 | Overall patient cohort | |
| 3.3.2 Exposed patient co | | Exposed patient cohort | |
| 3.3 | 3.3 | Distribution of procedure frequency and dose | 85 |
| 3.3 | 8.4 | Pregnant patient cumulative radiation exposure | 85 |
| 3.3 | 8.5 | Foetal cumulative radiation exposure | 90 |
| 3.4 | Disc | USSION | 97 |
| 3.5 | Con | CLUSION | |
| 3.6 | Refe | RENCES | |
| 4 CH | IAPTE VASCL | R 4 <i>IN UTERO</i> EXPOSURE TO LOW DOSE IONIZING RADIATION: JLAR AND RESPIRATORY OUTCOMES IN C57BL/6J MICE | long term |
| SUM | MARY | , | |
| 4.1 | Intr | ODUCTION | 111 |
| 4.2 | Met | HODS | 113 |
| 4.2 | 2.1 | Experimental animals and irradiations | 113 |
| 4.2 | 2.2 | Cardiovascular measurements | 114 |
| 4.2 | 2.3 | Respiratory measurements | 115 |
| 4.2 | 2.4 | Statistical Analysis | 115 |
| 4.3 | Resu | JLTS | 116 |
| 4.3 | 3.1 | Growth rate | 116 |
| 4.3 | 3.2 | Cardiovascular system | 116 |
| 4.3 | 3.3 | Respiratory system | 120 |
| 4.4 | Disc | USSION | 123 |
| 4.5 | Con | CLUSION | 127 |
| 4.6 | Refe | RENCES | 128 |
| 5 CH RESPON | IAPTE SE TO | r 5 <i>In utero</i> exposure to low dose ionizing radiation and the an acute lung injury stimulus at adolescence in balb/c mice | RESPIRATORY |
| | | | |

| PREFA | ACE | 131 |
|-------|--------------|-----|
| SUMN | MARY | 132 |
| 5.1 | Introduction | 134 |

| | Me | THODS | |
|---|---|--|-----------------------|
| 5. | 2.1 | Experimental animals and irradiations | 137 |
| 5. | 2.2 | LPS Model | |
| 5. | 2.3 | Surgery and tissue collection | |
| 5. | 2.4 | Immunoassays | |
| 5. | 2.5 | Tissue homogenisation and western blot | |
| 5. | 2.6 | Statistical Analysis | 141 |
| 5.3 | Res | ULTS | 141 |
| 5. | 3.1 | Transport effect | 141 |
| 5. | 3.2 | Litter size and growth rate | 142 |
| 5. | 3.3 | Respiratory physiology | 147 |
| 5. | 3.4 | Respiratory immunology | 147 |
| 5. | 3.5 | Respiratory antioxidant levels | |
| 5.4 | Dis | CUSSION | |
| 5.5 | Co | NCLUSION | |
| | | | |
| 5.6 | Ref | ERENCES | 170 |
| 5.6 6 CH | Ref H apte | erences R 6 Overall discussion | 170 |
| 5.6 6 Cr 6.1 | Ref HAPTE Clii | erences ER 6 Overall discussion | 170 176 |
| 5.6 6 CH 6.1 6.2 | Ref HAPTE Clii Ani | erences FR 6 OVERALL DISCUSSION NICAL AUDITS COMPARISON MAL STUDIES COMPARISON | |
| 5.6 6 Cr 6.1 6.2 6.3 | Ref HAPTE Clii Ani tra | erences R 6 Overall discussion NICAL AUDITS COMPARISON MAL STUDIES COMPARISON NSLATING ANIMAL FINDINGS TO HUMANS | |
| 5.6 6 CF 6.1 6.2 6.3 6.4 | Ref HAPTE Clii Ani Tra Ovi | erences R 6 Overall discussion NICAL AUDITS COMPARISON MAL STUDIES COMPARISON NSLATING ANIMAL FINDINGS TO HUMANS ERALL CONCLUSIONS | |
| 5.6 6.1 6.2 6.3 6.4 6.5 | Ref HAPTE Clii Ani Tra Ovi Fut | ERENCES ER 6 OVERALL DISCUSSION NICAL AUDITS COMPARISON MAL STUDIES COMPARISON NSLATING ANIMAL FINDINGS TO HUMANS ERALL CONCLUSIONS TURE DIRECTIONS | |
| 5.6 6.1 6.2 6.3 6.4 6.5 6.6 | Ref HAPTE Clii Ani tra Ovi Fut Ref | ERENCES ER 6 OVERALL DISCUSSION NICAL AUDITS COMPARISON MAL STUDIES COMPARISON NSLATING ANIMAL FINDINGS TO HUMANS ERALL CONCLUSIONS URE DIRECTIONS ERENCES | |
| 5.6 6.1 6.2 6.3 6.4 6.5 6.6 7 Cr | Ref HAPTE Clii Ani tra Ovi Fut Ref HAPTE | ERENCES ER 6 OVERALL DISCUSSION NICAL AUDITS COMPARISON MAL STUDIES COMPARISON NSLATING ANIMAL FINDINGS TO HUMANS ERALL CONCLUSIONS URE DIRECTIONS ERENCES ER 7 COTUTELLE PROGRAM EXPERIENCE | |
| 5.6 6.1 6.2 6.3 6.4 6.5 6.6 7 Cr 7.1 | Ref HAPTE CLII ANI TRA OVI FUT REF HAPTE | ERENCES | |
| 5.6 6.1 6.2 6.3 6.4 6.5 6.6 7 Cr 7.1 7.2 | Ref HAPTE CLII ANI TRA OVI FUT REF HAPTE UNI CO | ERENCES | |
| 5.6 6.1 6.2 6.3 6.4 6.5 6.6 7 Cr 7.1 7.2 7.3 | Ref HAPTE CLII ANI TRA OVI FUT REF HAPTE UNI CO ⁻ CO ⁻ | ERENCES | |
| 5.6 6 6.1 6.2 6.3 6.4 6.5 6.6 7 7.1 7.2 7.3 7.3 7.1 | Ref HAPTE CLII ANI TRA OVI FUT REF HAPTE UNI CO ^T CO ^T | ERENCES | |
| 5.6 6.1 6.2 6.3 6.4 6.5 6.6 7.1 7.2 7.3 7.3 7.3 | Ref HAPTE CLII ANI TRA OVI FUT REF HAPTE UNI CO ⁻ CO ⁻ 3.1 3.2 | ERENCES | |

| 7.3.4 | Co-Supervision | 212 |
|-------|-----------------|-----|
| 7.3.5 | Final statement | 212 |

LIST OF TABLES

| Table 1-1: Radiation weighting factors for different forms of radiation | 4 |
|--|--|
| Table 1-2: Tissue weighting factors for different organs of the body used to calculate effective dose | 4 |
| Table 1-3: Effective doses from various diagnostic radiation modalities | 8 |
| Table 1-4: Effective doses from various nuclear medicine exams | 9 |
| Table 1-5: Foetal doses following common diagnostic procedures | 27 |
| Table 1-6: Summary of suspected in utero ionizing radiation-induced deterministic effects | 33 |
| Table 2-1: Conversion factors for each modality divided by body region scanned or radiopharmaceutica | al used 58 |
| Table 2-2: Summary of patient demographics and clinical characteristics | 60 |
| | trihute |
| Table 2-3: Distribution of procedures which use ionizing radiation and the frequency at which they cont number of procedures and total dose | 62 |
| Table 2-3: Distribution of procedures which use ionizing radiation and the frequency at which they cont number of procedures and total dose Table 2-4: Exposures from each modality divided by body region scanned | 62 |
| Table 2-3: Distribution of procedures which use ionizing radiation and the frequency at which they continumber of procedures and total dose Table 2-4: Exposures from each modality divided by body region scanned Table 3-1: Conversion factors for each modality, divided by body region scanned or radiopharmaceutica | 62 62 al used 79 |
| Table 2-3: Distribution of procedures which use ionizing radiation and the frequency at which they continumber of procedures and total dose Table 2-4: Exposures from each modality divided by body region scanned Table 3-1: Conversion factors for each modality, divided by body region scanned or radiopharmaceutica Table 3-2: Distribution of patient and procedure frequency of pregnant patients who received any diag radiation during the length of stay | al used 62 al used 79 gnostic 83 |
| Table 2-3: Distribution of procedures which use ionizing radiation and the frequency at which they continumber of procedures and total dose Table 2-4: Exposures from each modality divided by body region scanned Table 3-1: Conversion factors for each modality, divided by body region scanned or radiopharmaceutica Table 3-2: Distribution of patient and procedure frequency of pregnant patients who received any diag radiation during the length of stay | al used 62 al used 79 gnostic 83 |
| Table 2-3: Distribution of procedures which use ionizing radiation and the frequency at which they cont number of procedures and total dose Table 2-4: Exposures from each modality divided by body region scanned. Table 3-1: Conversion factors for each modality, divided by body region scanned or radiopharmaceutica Table 3-2: Distribution of patient and procedure frequency of pregnant patients who received any diag radiation during the length of stay Table 3-3: Summary of patient demographic and clinical characteristics Table 3-4: Distribution of procedures which use ionizing radiation and the frequency at which they cont to the number of patients, number of procedures and total dose. | al used 62 al used 79 gnostic 83 84 tribute 86 |
| Table 2-3: Distribution of procedures which use ionizing radiation and the frequency at which they contain number of procedures and total dose. Table 2-4: Exposures from each modality divided by body region scanned. Table 3-1: Conversion factors for each modality, divided by body region scanned or radiopharmaceutica and total dose. Table 3-2: Distribution of patient and procedure frequency of pregnant patients who received any diag radiation during the length of stay | al used 62 al used 79 gnostic 83 84 tribute 86 |
| Table 2-3: Distribution of procedures which use ionizing radiation and the frequency at which they cont number of procedures and total dose. Table 2-4: Exposures from each modality divided by body region scanned. Table 3-1: Conversion factors for each modality, divided by body region scanned or radiopharmaceutica radiation during the length of stay Table 3-3: Summary of patient demographic and clinical characteristics Table 3-4: Distribution of procedures which use ionizing radiation and the frequency at which they cont to the number of patients, number of procedures and total dose. Table 3-5: Exposures from each modality divided by body region scanned. Table 3-6: Top 5 highest dose procedures with patient characteristics and diagnoses | al used 62 al used 79 gnostic 83 84 tribute 86 87 88 |

LIST OF FIGURES

| Figure 1-1: Dose quantities and units of ionizing radiation used in radiation research | 5 |
|--|--------------|
| Figure 1-2: Schematic figure for the four dose-response model hypotheses in the low dose range | 15 |
| Figure 1-3: Schematic comparing mouse and human progression times through the 5 stages of respiratory development ¹⁶⁵ | normal 21 |
| Figure 2-1: Distribution of patient exposure grouped by cumulative effective dose | 64 |
| Figure 2-2: Patient factors that contribute to cumulative effective dose | 65 |
| Figure 1-1: Flow chart of patient eligibility criteria for inclusion in retrospective audit | 82 |
| Figure 3-2: Distribution of patient cumulative exposure grouped by cumulative effective dose | 89 |
| Figure 3-3: Factors that affect maternal cumulative doses | 93 |
| Figure 3-4: Factors that affect foetal cumulative effective doses | 96 |
| Figure 4-1: longitudinal weight measurements | 117 |
| Figure 4-2: Longitudinal cardiovascular outcomes | 119 |
| Figure 4-3: Respiratory physiological outcomes | 121 |
| Figure 4-4: Respiratory immunological outcomes | 122 |
| Figure 5-1: Litter size of irradiated mothers | 143 |
| Figure 5-2: longitudinal weight measurements | 144 |
| Figure 5-3: Physiological outcomes of the respiratory system | 145 |
| Figure 5-4: Immunological cell outcomes of the respiratory system | 146 |
| Figure 5-5: Soluble TGF- β concentrations by ELISA in BAL and Plasma | 149 |
| Figure 5-6: Western blot membrane and gel bands for antioxidant proteins | 151 |
| Figure 5-7: Western blot intensities of antioxidant proteins | 153 |
| Figure 5-8: Saturation curves of antioxidant proteins | 155 |

LIST OF ABBREVIATIONS

| ¹²³ | - | lodine-123 |
|-------------------|---|---|
| ¹³⁷ Cs | - | Caesium-137 |
| ¹⁸ F | - | Fluoride-18 |
| ²⁰¹ TI | - | Thallium-201 |
| ^{99m} Tc | - | Technetium-99m |
| a.u. | - | Arbitrary units |
| ALARA | - | As low as reasonably achievable |
| ANOVA | - | Analysis of variance |
| APACHE | - | Acute physiology, age, chronic health evaluation |
| ARPANSA | - | Australian radiation protection and nuclear safety agency |
| Atm | - | atmospheres |
| AUS | - | Australian |
| BAL | - | Bronchoalveolar lavage |
| Bq | - | Becquerel |
| C.I. | - | Confidence interval |
| CAN | - | Canadian |
| CED | - | Cumulative effective dose |
| cm | - | Centimetre |
| CT | - | Computed tomography |
| СТРА | - | Computed tomography pulmonary angiography |
| CXR | - | Chest X-ray |
| DAP | - | Dose area product |
| DBP | - | Diastolic blood pressure |
| dH_2O | - | Demineralised water |
| DLP | - | Dose length product |
| DNA | - | Deoxyribonucleic acid |
| EDTA | - | Ethylenediaminetetraacetic acid |
| ERR | - | Excess relative risk |
| FEV | - | Forced expiratory volume |
| FMC | - | Flinders Medical Centre |
| g | - | Gram |

| GAM | - | Generalised additive model |
|----------|---|---|
| GD | - | Gestational day |
| GLMM | - | Generalised linear mixed model |
| GPx | - | Glutathione peroxidase |
| Gy | - | Gray |
| GYN | - | Gynaecology |
| h | - | Hour |
| H_2O_2 | - | Hydrogen peroxide |
| HELLP | - | Haemolysis, elevated liver enzymes, and low platelet count syndrome |
| HIP | - | Health information portal |
| I.P. | - | Intraperitoneal |
| I.T. | - | Intratracheal |
| I.V. | - | Intravenous |
| ICCU | - | Intensive and critical care unit |
| ICRP | - | International commission on radiation protection |
| ICU | - | Intensive care unit |
| lgG | - | Immunoglobulin G |
| IHC | - | Immunohistochemistry |
| IL | - | Interleukin |
| INF-γ | - | Interferon-gamma |
| IQR | - | Interquartile range |
| J | - | Joules |
| KC | - | Keratinocyte chemokine |
| kDa | - | kilodalton |
| KeV | - | Kiloelectron volts |
| kVp | - | Peak kilovoltage |
| L | - | Litre |
| LAT | - | Lateral image |
| LD | - | Lethal dose |
| LNT | - | Linear no-threshold |
| LOG | - | Logarithm |
| LOS | - | Length of stay |
| LPS | - | Lipopolysaccharide |

| LSS | - | Lifespan study |
|--------|---|--|
| Μ | - | Molarity |
| m | - | Metre |
| MAP | - | Mean arterial pressure |
| mAs | - | Milliampere-seconds |
| mg | - | Milligram |
| mGy | - | Milligray |
| min | - | Minute |
| mL | - | Millilitre |
| mМ | - | Millimolar |
| mm | - | Millimetre |
| MRI | - | Magnetic resonance imaging |
| mSv | - | Millisievert |
| n | - | Number of experiments |
| NaCl | - | Sodium Chloride |
| NCRP | - | National commission on radiation protection |
| NF-κB | - | Nuclear factor kappa-light-chain-enhancer of activated B cells |
| ng | - | Nanogram |
| nm | - | Nanometre |
| Nrf1/2 | - | Nuclear factor erythroid 2-related factor 1 and 2 |
| OACIS | - | Open architecture clinical information system |
| OBS | - | Obstetrics |
| PA | - | Posterior-anterior image |
| PACS | - | Picture archiving and communication system |
| PBS | - | Phosphate buffered saline |
| PBST | - | Phosphate buffered saline + Tween |
| PVDF | - | Polyvinylidene fluoride |
| RNA | - | Ribonucleic acid |
| ROS | - | Reactive Oxygen Species |
| S | - | Second |
| S.E.M | - | Standard Error of the Mean |
| SBP | - | Systolic blood pressure |
| SOD | - | Superoxide dismutase |

| Sv | - | Sievert |
|---------|---|--|
| TBE | - | Tris Borate EDTA |
| TBS | - | Tris Buffered Saline |
| TGF-β | - | transforming growth factor-beta |
| TLR4 | - | Toll-like receptor 4 |
| TNF-α | - | Tumour necrosis factor-alpha |
| UK | - | United Kingdom |
| UNSCEAR | - | United nations scientific committee on the effects of atomic radiation |
| USA | - | United States of America |
| V | - | Volt |
| μg | - | Microgram |
| μL | - | Microliter |
| μm | - | Micrometre |
| μM | - | Micromolar |

| # | - | Number |
|--------------|---|-------------------------------------|
| % | - | Percentage |
| ~ | - | Approximately/no substantial change |
| < | - | Less than |
| > | - | Greater than |
| ≤ | - | Less than or equal to |
| ≥ | - | Greater than or equal to |
| \uparrow | - | Increased/elevated |
| \downarrow | - | Decreased/reduced |
| 0 | - | Degrees |
| °C | - | Degrees Celsius |
| μ | - | Mu |
| α | - | Alpha |
| β | - | Beta |
| γ | - | Gamma |
| к | - | Карра |
| ρ | - | rho |
| | | |

DECLARATION OF ACADEMIC ACHIEVEMENT

This thesis is arranged in a sandwich format approved by McMaster University and with the permission of the supervisory committee. The thesis contains 7 chapters. Chapter 1 describes the background information of ionizing radiation with an overview of the effects during *in utero* exposure. Chapter 2 and 3 are retrospective audits of patient cumulative exposure to diagnostic radiation. Chapter 4 describes the effects of *in utero* exposures to ionizing on the cardiovascular and respiratory system in a healthy animal model. Chapter 5 examines the effect of *in utero* exposures to ionizing radiation on an acute lung injury model. Chapter 6 summarises and concludes the results found in chapters 2-5 and establishes future directions of research. Lastly, Chapter 7 summarises the Cotutelle experience.

This preface is to certify that a chapter within this thesis contains content that is substantially unchanged from the content of multi-author papers which have either been published or are being prepared for publication, which may lead to some repetition of ideas.

CHAPTER 1 INTRODUCTION

CHAPTER 2 CUMULATIVE RADIATION IN CRITICALLY ILL PATIENTS: A RETROSPECTIVE AUDIT OF IONIZING RADIATION EXPOSURE IN AN INTENSIVE CARE UNIT.

Authors: J.H. McEvoy, S. Bihari, A.M. Hooker, D-L. Dixon

xxii

Date accepted: May 4, 2019

Journal: Critical Care and Resuscitation

- Approval: written approval has been obtained to reprint this manuscript (Appendix A)
- Comments: JM, planned and conducted this study under the supervision of SB, AM and DD. JM and SB contributed to collection of data, JM and AH contributed to effective dose calculation, and JM and DD contributed to data analysis. JM, SB, AH and DD contributed to the study design, data interpretation, and the editing of the manuscript.

CHAPTER 3 CUMULATIVE RADIATION IN HOSPITALISED PREGNANT PATIENTS: A RETROSPECTIVE AUDIT OF IONIZING RADIATION EXPOSURE.

Authors: J.H. McEvoy, S. Bihari, A.M. Hooker, D-L. Dixon

Comments: JM, planned and conducted this study under the supervision of SB, AM and DD. JM and SB contributed to collection of data, JM and AH contributed to effective dose calculation, and JM and DD contributed to data analysis. JM, SB, AH and DD contributed to the study design, data interpretation, and the editing of the chapter.

CHAPTER 4 *IN UTERO* EXPOSURE TO LOW DOSE IONIZING RADIATION: LONG TERM CARDIOVASCULAR AND RESPIRATORY OUTCOMES IN C57BL/6 MICE.

Authors: J.H. McEvoy, D. Jones, L. Stoa, D-L. Dixon, T.C Tai, A. Hooker, D. Boreham and J.Y. Wilson

- Comments: JM, planned and conducted this study under the supervision of DD and JW. JM, DJ and LS contributed to animal husbandry and collection of data. JM, DD and JW contributed to data analysis and interpretation. TT, DB and JW contributed to funding. DD, AH and JW contributed to the editing of the chapter.
- CHAPTER 5 *IN UTERO* EXPOSURE TO LOW DOSE IONIZING RADIATION AND THE RESPIRATORY RESPONSE TO AN ACUTE LUNG INJURY STIMULUS AT ADOLESCENCE IN BALB/C MICE.
- Authors:J.H. McEvoy, S. Sreetharan, L. Stoa, D-L. Dixon, T.C Tai, A. Hooker, D.Boreham and J.Y. Wilson
- Comments: JM, planned and conducted this study under the supervision of DD

and JW. JM, SS and LS contributed to animal husbandry and

collection of data. JM, DD and JW contributed to data analysis and

interpretation. DD, AH, TT, DB and JW contributed to funding. DD,

AH and JW contributed to the editing of the chapter.

CHAPTER 6 DISCUSSION

CHAPTER 7 COTUTELLE PROGRAM EXPERIENCE

CHAPTER 1

INTRODUCTION

1.1 RADIATION

Radiation is the emission of energy through space as subatomic particles or waves including electromagnetic, acoustic or gravitational. These forms can travel with a variety of energy levels. Radio waves, microwaves and visible light are all forms of radiation with low energy termed non-ionizing radiation. When the energy level passes a threshold where it is enough to emit an electron from an atom, thereby ionizing the atom, it is considered ionizing radiation, such as X-rays, gamma rays or beta and alpha particles. Because of this effect, ionizing radiation is more biologically relevant, having the effect to inhibit cell function, mutate cells and even cause cell death. However, understanding how this energy can cause these biological effects is complicated and requires an understanding of source and dose of ionizing radiation.

1.1.1 Sources of ionizing radiation

People are continuously exposed to radiation, from naturally occurring sources to artificial man-made sources. We are exposed to natural radiation from cosmic and terrestrial sources. Stars continuously produce cosmic radiation in the form of electromagnetic waves; however, we are mostly protected by the earth's electromagnetic field. Terrestrial exposures come from radioactive material or radioactive isotopes found in our air, food, and soil, including uranium, thorium, radon and potassium. Due to instability, these atoms decay and produce ionizing radiation. Interestingly, humans themselves are radioactive due to radioactive isotopes absorbed within the body, for example, potassium-40 and carbon-14. The dose, however, is relatively small compared to those exposed from terrestrial and cosmic sources¹.

Man-made sources of radiation typically harness natural sources of radiation for other intentions, such as war or terrorism, nuclear power plants, and medical radiation². The atomic bombs dropped in Japan during World War II harnessed the natural radioactive decay of uranium and plutonium in an extremely concentrated way. Similarly, nuclear power plants harness decaying radioactive material for electricity production. Ionizing radiation used in the medical industry harnesses both wave (X-ray and Gamma-ray) and particle (beta, alpha) forms for diagnostics or therapeutic uses¹.

1.1.2 Doses of ionizing radiation

To quantify dose, the energy of ionizing radiation deposited per unit mass is measured and reported in grays (Gy). Termed absorbed dose, 1 J of energy deposition per Kg of tissue = 1000 mGy. However, not all types of ionizing radiation elicit the same biological response, this is the concept of radiation biological effectiveness. Photons (gamma rays or X-rays), beta particles, alpha particles etc. deposit different amounts of energy in space and therefore can produce different biological effects. Due to this, a radiation-specific weighting factor for each radiation type is applied to the absorbed dose to create the equivalent dose, measured in sieverts $(Sv)^2$ (Table 1-1). Finally, radiation doesn't affect each organ in the body similarly, as some are more radiosensitive than others. Tissues such as lung and bone marrow are more radiosensitive than skin or brain and so have more of a risk of radiation causing stochastic effects. These tissues, therefore, relate to a higher tissue weighting factor. This tissue weighting factor is applied to the equivalent dose to create the effective dose, also measured in sieverts (Table 1-2)². The link between absorbed, equivalent and effective dose can be visualised in Figure 1-1.

Additionally, sources such as naturally occurring radon or medical radiopharmaceuticals are expressed in becquerels (Bq); a measurement of radioactivity^{2,3}. Becquerels can, with extensive calculations, be converted into mSv, which generally occurs when considering the effect from multiple forms of radiation to the person, such as yearly background exposures or cumulative patient exposures.

Natural sources of radiation contribute a large amount to the ubiquitous background radiation exposure each year. The average natural background exposure for Australians and Canadians are 1.5 and 1.8 millisieverts (mSv) per year, respectively^{4,5}. However, other places in the world receive higher averages such as Ramsar, Iran (6.0 mSv/y), Yangjiang, China (6.4 mSv/y), Kerala, India (6.9 mSv/y) and Poços de Caldas, Brazil (13 mSv/y), with some inhabited rural areas >100 mSv/y^{6–9}. Table 1-1: Radiation weighting factors for different forms of radiation used to calculate the equivalent dose, with permission from ICRP². Photons relate to X-ray and gamma-ray radiation.

| Radiation type | Radiation weighting factor | |
|----------------------------|----------------------------|--|
| Photons, Beta particle | 1 | |
| Protons | 2 | |
| Alpha particle, heavy ions | 20 | |
| Neutrons | A continuous function of | |
| ine ations | neutron energy | |

Table 1-2: Tissue weighting factors for different organs of the body used to calculate effective dose, with permission from ICRP². Organs can differ in their radio-sensitivity, thus each organ within the scan field must be compensated for.

| Organs | Tissue | weighting |
|-------------------|---------|-----------|
| Orguns | factors | |
| Red Bone Marrow | | 0.12 |
| Colon | | 0.12 |
| Lung | | 0.12 |
| Stomach | | 0.12 |
| Breasts | | 0.12 |
| Gonads | | 0.08 |
| Bladder | | 0.04 |
| Liver | | 0.04 |
| Oesophagus | | 0.04 |
| Thyroid | | 0.04 |
| Skin | | 0.01 |
| Bone surface | | 0.01 |
| Salivary glands | | 0.01 |
| Brain | | 0.01 |
| Remainder of body | | 0.12 |
| Total | | 1.00 |



Figure 1-1: Dose quantities and units of ionizing radiation used in radiation research.

Occupational exposures can also be from natural sources but are present only when the person is in proximity to the occupational source. It can, however, have a large contribution to a person's yearly exposure depending on their occupation⁴. Pilots of international flights can receive up to 4 mSv a year, due to extended periods at high altitude resulting in increased exposure to cosmic radiation¹⁰. On the other hand, some miners working underground are exposed to 3-8 mSv per year due to radioactive uranium, thorium or radon in the rock^{11,12}. With the increase in radiation exposure, comes an increase in the perception of risk and thus strict regulations are enforced. The occupational limit for ionizing radiation exposure is 20 mSv per year averaged over 5 years, with a clausal limit of 50 mSv in a single year¹³.

Medical radiation comprises the dominant exposure from man-made sources. It accounts for 0.9-1.5 mSv/yr bringing the average yearly exposure to 2.4-3 mSv/yr for Australians and 3.6 mSv/yr for Canadians^{4,14}. Radiation is an important tool aiding in the diagnosis and management of disease and as a therapeutic tool to treat disease. While therapeutic exposures use much higher doses of ionizing radiation, diagnostic exposures fall in the low dose radiation range.

1.2 LOW DOSE RADIATION

Diagnostic exposures fall under the category of low dose radiation which the United Nations Scientific Committee on the Effects of Atomic Radiation (UNSCEAR) defines as a single exposure below 100 mSv, in line with the International Council on Radiation Protection (ICRP)^{2,15}. Under this threshold, the biological effects of radiation continue to be debated^{16–19}.

1.2.1 Diagnostic modalities using ionizing radiation

Conventional (plain diagnostic) X-rays, computed tomography (CT), fluoroscopy and nuclear medicine are all forms of ionizing radiation which clinicians use to enable quicker and more accurate patient diagnosis and prognosis.

Conventional radiology, or X-ray radiography, provides a useful non-invasive technique of visualising the internal tissues and structures of the body. Dense tissues, such as bone, absorb X-rays and appear white on photographic films and less dense tissues allow X-rays to pass through resulting in dark areas on the films²⁰. X-rays can be taken of any part of the body and can be used for diagnostic assessment and monitoring. Mettler *et al.* collected published data between 1980-2007 from around the world and compiled the adult effective doses for each procedure as given in Table 1-3 and Table 1-4²¹. Although frequent, the amount of radiation received from X-ray procedures is relatively small compared to other procedures. The average chest X-Ray (CXR) accounts for 0.02 mSv, equating 5 days background radiation in Australia (1.5 mSv)^{4,21}.

Table 1-3: Effective doses from various diagnostic radiation modalities including conventional X-rays, computed tomography and fluoroscopy, adapted from Mettler *et al* with permission from RNSA²¹.

| Examination | Effective dose (mSv) |
|--|----------------------|
| NUCLEAR MEDICINE | |
| Brain (¹⁸ F or ^{99m} Tc) | 5.7-14.1 |
| Thyroid scan (^{99m} Tc or ¹²³ I) | 1.9-4.8 |
| Cardiac Stress (^{99m} Tc or ²⁰¹ Tl) | 9.4-40.7 |
| Lung ventilation/perfusion (^{99m} Tc) | 0.2-2 |
| Renal (^{99m} Tc) | 1.8-6.3 |
| Bone (^{99m} Tc) | 6.3 |

 18 F = fluorine 18, 99m Tc = Technetium 99m, 123 I = Iodine 123, 201 Tl = Thallium 201.

| Examination | Average effective dose (mSv) | Values reported in Literature (mSv) |
|-------------------------------|---------------------------------|--|
| CONVENTIONAL RADIOGRAPHY | | |
| Skull | 0.1 | 0.03-0.22 |
| Cervical spine | 0.2 | 0.07-0.3 |
| Thoracic spine | 1 | 0.6-1.4 |
| Lumbar spine | 1.5 | 0.5-1.8 |
| LAT chest | 0.1 | 0.05-0.24 |
| AP chest | 0.02 | 0.007-0.05 |
| Abdomen | 0.7 | 0.04-1.1 |
| Pelvis | 0.6 | 0.2-1.2 |
| Hip | 0.7 | 0.18-2.71 |
| Other extremities | 0.001 | 0.0002-0.1 |
| COMPUTED TOMOGRAPHY | | |
| Head | 2 | 0.9-4.0 |
| Neck | 3 | |
| Head and neck angiography | 5 | 0.8-19.6 |
| Chest | 7 | 4-18 |
| Thoracic angiography | 15 | 13-40 |
| Abdomen | 8 | 3.5-25 |
| Pelvis | 7 | 3.3-10 |
| Abdominal angiography | 12 | 4-48 |
| Spine | 6 | 1.5-10 |
| Coronary angiography | 15 | 7-57 |
| FLUOROSCOPY | | |
| Intravenous urography | 3 | 0.7-3.7 |
| Upper gastrointestinal series | 6 | 1.5-12 |
| Barium Enema | 8 | 2-18 |

Table 1-4: Effective doses from various nuclear medicine exams, adapted from Mettler *et al* with permission from RNSA²¹.

mSv = *millisieverts*, *LAT* = *lateral position*, *AP* = *anteroposterior position*

CT is a combination of many X-rays taken from different angles to create crosssectional images. It was defined as a high dose technique in 1990 due to its slice by slice, axial imaging, and although several improvements have reduced the dose, compared to general radiology, CT procedures are still a high dose technique²⁰. These doses can vary significantly based on procedure type and body location of scan, for example, an abdominal CT = 8 mSv, but can vary 5 fold (Table 1-3)²¹. Technological developments and increases in the amount of information received from the exam have led to an increase in the use of CT scans over recent decades. In 1989, the National Radiation Protection Board in the UK estimated CT represent 2% of radiological procedures and contribute 20% to the cumulative effective dose (CED)²². Less than a decade later in 1998, the figures were 5% and 40% respectively²³. In 2014, CT made up 11% of procedures and 82% CED²⁴. The significant benefit of CT to healthcare has resulted in continued improvement in scanner technology and clinical application²⁵. Currently, a CT scan can compile a 3D reconstruction of the internal organisation of the body within a few seconds, thereby significantly decreasing the dose.

Fluoroscopy uses a continuous X-ray beam. Unlike general radiology, the X-rays are transmitted to a monitor allowing for real-time visualisation of tissue or contrast dye movements throughout the body²⁰. Fluoroscopy is used for assessing swallowing or upper and lower gastrointestinal tract movements. Additionally, it is useful for arthrography, angiography and aiding in placement of intravenous catheters and stents^{26,27}. Fluoroscopic examinations are high in individual exposures, similar to lower-end CT scans, but are relatively infrequently prescribed²⁴. The dose is highly depended on the time taken thus careful consideration is needed as extended skin exposure to radiation can result in skin burns and ulcerations²⁷.

Nuclear Medicine is the use of radiopharmaceuticals to visualise internal structures in the place of an X-ray beam. Harnessing gamma (γ) waves, which are very similar to Xrays, radioisotopes are injected, inhaled or swallowed and visualised by recording the emitted radiation²⁰. Procedures can include myocardial, lung or renal perfusion tests, and bone density scans and can range from 0.2-40 mSv as seen in Table 1-4²¹. Choice of radioisotope depends on the requirements for the procedure, including the state of isotope, decay rate, or uptake in the target organ, but nuclear medicine procedures are the least frequent modality prescribed^{21,24}.

1.2.2 Alternative diagnostic procedures

Some diagnostic procedures, such as magnetic resonance imaging and ultrasound, use other types of radiation to visualise the internal structures of the body without harnessing ionizing radiation. Magnetic resonance imaging (MRI) is also a non-invasive technique that harnesses magnetic fields and radio waves. MRI spectroscopically measures the magnetic properties of the nucleus, particularly protons. Beginning in the 1940s, its applications take advantage of its high contrast in soft tissue differentiation, including brain and skeletomuscular scans, contrast perfusion and diffusion imaging and angiography. MRI drawbacks include significant imaging time, image artefact production and cost²⁰.
Ultrasound is another non-invasive diagnostic technique but rather harnesses soundwaves and the acoustic properties of the body to produce an image. These sounds waves, which exceed the frequencies audible to humans, are pulsed into the tissue and depending on the acoustic properties of different tissues, sound waves are reflected as an echo. This process is repeated, and the collected echoes are transformed into an image. Ultrasound is used during pregnancy to visualise the foetus but can also be used for needle biopsy guidance and cardiovascular disease. Disadvantages of this method include poor image clarity/detail, artefact creation and requirement of significant training to establish and maintain competency^{20,28}.

1.3 RISKS ASSOCIATED WITH LOW DOSES OF IONIZING RADIATION

The observation that environmental factors can influence the occurrence of mutation, and thus cancer, has been known since the late 18th century²⁹. However, the recognition of mutations caused by X-rays came almost 50 years later from Muller³⁰. Although failing to cite similar work published earlier that year, Muller was credited with the observation that X-rays cause mutations in Drosophila, which induced transgenerational, phenotypic changes providing a plausible mechanism of evolution. This finding very quickly transformed the radiation field and later gained him the Nobel prize. Indeed, it was from a combination of Muller's pioneering observation, target theory, and some mathematical modelling that the first mechanism based, risk model (linear no-threshold model - LNT) was created³¹. However, as understanding in mutations has

improved, some scientists have begun to rebut the interpretations of his results, as reviewed by Calabrese^{32–34}.

A cornerstone observation made by Caspari and Stern found that dose rate played an integral role in the effect of radiation³⁵. This challenged the LNT model and hypothesised that either a threshold dose does exist or that multiple independent actions are required to initiate a mutation. Over the decades since this observation, numerous studies have not supported the LNT model, which has given rise to other models to calculate risks in the low dose range.

1.3.1 Modelling radiation risk

Of the many current hypothesised models currently debated, Figure 1-2 shows four examples. The low-dose-high-sensitivity hypothesis assumes there are people who are highly sensitive to radiation and thus differ in their DNA damage response to radiation exposure deleteriously³⁶. Contrary to this, the hormesis hypothesis assumes that there is a health benefit at low exposures, through stimulation of adaptive responses, compared to non-exposed persons, as reviewed by Vaiserman³⁷. However, the top two debated models are the Linear No-Threshold (LNT) model and the threshold model. LNT is a linear extrapolation from high dose radiation and infers no "safe" level of radiation. Although this model is the current international standard enforced by most, if not all, radiation protection agencies, including the International Commission of Radiation Protection (ICRP), International Atomic Energy Agency (IAEA), National Commission of Radiation Protection (NCRP), and Australasian Radiation Protection and Nuclear Safety Agency (ARPANSA)^{2,38,39}, the support is foremost as a radiation protection model and not as a cancer risk assessment model due to the difficulty in accurately measuring detrimental effects at low doses. Therefore, it is impractical to produce cancer risk estimates for diagnostic radiation, which is all below 100 mGy. The contradicting evidence to the LNT is substantial, as suggested by Tubiana *et al*, Averbeck *et al* and Calabrese *et al*^{34,40–42}. The majority that opposes the LNT hypothesis support the threshold hypothesis, which indicates that under a "threshold" dose there are no radiation-related health detriments⁴³. This hypothesis would make it easier to ignore perceived risks associated with very low dose radiation exposure, however as the major governing bodies support the LNT model, the current paradigm follows the 'as low as reasonably achievable' (ALARA) principle which aims at reducing any exposure unless an absolute necessity and thus reducing the perceived risk to the person.



Figure 1-2: Schematic figure for the four dose-response model hypotheses in the low dose range, reprinted with the permission from Wolters Kluwer Health Inc.¹⁶⁴. (a) the low-dose-high-sensitivity hypothesis, (b) the linear no-threshold (LNT) hypothesis, (c) the threshold hypothesis, and (d) the hormesis hypothesis. *ERR = excess relative risk*

1.3.2 Professional understanding of ionizing radiation

The perception of risk changes with the level of knowledge of radiation. Typically, professionals with more experience in the field of radiation have less fear of low dose radiation whereas those with limited radiation understanding exacerbate risk perception⁴⁴. This, unfortunately, leads to some health care workers being unsure of the doses associated with ionizing radiation procedures or the actual risks associated with those doses⁴⁵. In a survey collected by Lee *et al*, 91% of doctors did not understand the lifetime risk associated with an abdominal CT, or if it contributed to cancer incidence. Therefore, they could not, and did not, impart this information to the patients. Additionally, radiologists (77%), clinicians (74%) and patients (100%) did not know the dose of an abdominal CT (8 mSv) compared to a chest X-ray (0.02 mSv)^{21,45}. Another study asked clinicians to compare a fluoroscopic exam to a chest X-ray and demonstrated a similar lack of understanding⁴⁶. Although these studies have small-medium cohorts from single centres, the results imply a limited understanding of radiation dose and the associated risks, which is an essential requirement for examining the risk-to-benefit ratio of diagnostic radiology^{3,45–50}. In nearly all cases, the benefit of the diagnostic imaging will outweigh the relatively small risks associated with the exam, however, the lack of understanding, and thus inversely related levels of fear, might lead the clinician away from diagnostic radiation and potential misdiagnoses. In contrast, the scientific evidence does not suggest that diagnostic radiation should be overused as this would increase cumulative exposures of radiation, financial burden and patient time mismanagement but instead prescribe clinically indicated procedures with a firm understanding of radiation, its effects and risks. It is therefore in the interest of the patient if the clinician has an accurate understanding of all factors relating to the risk to benefit ratio of prescribing any action/intervention including diagnostic radiation.

1.3.3 Decision making with ionizing radiation in clinical practice

In the instance of prescribing diagnostic radiation, the clinical indications and the risks associated with the procedure are important in the decision-making process, but the patient's individual circumstance can also have a huge effect on the outcome of the decision. From this understanding, two polar populations can be chosen based on their predicted levels of ionizing radiation exposures.

Amongst hospitalised patients, those in the Intensive Care Unit (ICU) have a high severity of illness and are most constantly and intensively monitored by equipment. As the patient's conditions are severe with dynamic changes, diagnostic radiation is frequently used, with general radiology the predominant procedure²⁴. Up until recent years, it was common for ICUs to administer chest X-rays as daily routine care until it was found to have no benefit in identifying a change in treatment^{51–54}. When faced with acute and severe disease, the attention is on addressing the immediate problems and so the stochastic risks of radiation are usually not the focus. Thus, this population can be predicted to receive high levels of ionizing radiation.

Conversely, pregnant patients are commonly admitted with a variety of illness severities, with some requiring acute medical care that may include diagnostic radiation. However, the effects of radiation will not only be to the mother but also to the unborn foetus, thus the clinician's risk assessment must extend to include that of the developing foetus. This shifts the focus to long term outcomes, as there is a lifetime of risk still remaining, meaning that the stochastic effects of radiation are greater. As such, it is predicted that clinicians may reduce exposure to pregnant patients, to reduce the risk to the foetus. There is no dose limit to pregnant patients, instead, the prescription must be clinically justified, however strict care is taken to reduce the foetal dose as much as possible⁵⁵.

1.4 PREGNANCY

The normal gestation can vary significantly depending on the species, mice have a 19-day gestation which is much small than the 38-40 weeks (280 days) for humans^{56,57}. In this time a single-cell zygote progresses through embryogenesis and foetal development until a newborn offspring is birthed. The first stage of development is the preimplantation period, which encompasses the first two weeks of development. In this period the single-celled zygote goes through cleavage, blastocyst formation and implantation, and ends in the development of a bilaminar disk. Next is the embryonic period, which lasts until week 8 and is the beginning of organogenesis. Each of the germ layers proceeds to differentiate into various tissues and by the end of this period, the main organs and major features of

the body are beginning to be established. The period of development from here (week 9) until birth is the foetal period, characterised by the maturation of organs and rapid growth⁵⁶.

1.4.1 Prenatal care

Prenatal care is a very important and influential period with small changes in the intrauterine environment having the potential to cause significant and long-lasting changes to the offspring. The premise that prenatal care can have a significant implement to the offspring's phenotype is known as the 'Barker Hypothesis' or foetal programming. It was first discussed in the early 1990s by David Barker, a physician and epidemiologist at the University of Southampton. He originated this term to describe his observations that infants of lower birthweight had more health-related problems later in life⁵⁸. Clinically, these infants had a higher risk of heart disease and diabetes⁵⁹. He suggested that deprivation prenatally, due to poor nutrition, alcohol or drug abuse, or typical all-around poverty, caused changes to the foetus and its organ development, which led to a higher risk of certain diseases later in life. It is now known that there are various stressors, which can induce foetal programming and affect many different organs including the lung.

1.4.2 Normal Development of the respiratory system

In mice, as reviewed by Maeda et al. and Pinkerton et al. and seen in Figure 1-3^{60,61}, the embryonic formation of the lung buds begins 9 days post-conception. The development of the major bronchi and division of the tracheal-oesophageal tube commences first. At gestational day 11, lung development reaches the pseudoglandular stage. Branching and budding begin to give the stereotypic structure of the lungs with the development of the bronchial and bronchiolar tubules. Smooth muscle builds around the airway branches and the pulmonary circulation is initiated by neighbouring blood vessel neovascularisation. Day 15 begins the movement into the canalicular phase, where the general structure of the lungs is complete with intertwining airway and vessel branches. The future conducting and respiratory zones are easily distinguishable creating the framework for region-specific differentiation. From the saccular phase until birth, the respiratory epithelium continues cellular differentiation. The peripheral lung spaces become smaller as saccules develop, the lung fluid begins reabsorption and synthesis of surfactant is initiated. Postnatally, the final phase of lung development occurs with alveolarization of saccular structures increasing surface area and pulmonary gas exchange. This concludes the morphogenesis of the lung giving rise to the fully formed and functional pulmonary system.



Figure 1-3: Schematic comparing mouse and human progression times through the 5 stages of normal respiratory development¹⁶⁵. Copyright with permission from LifeMap Discovery. The generation of lung tissue in humans follows a similar pattern as described for the mouse except with respect to timing. As reviewed by Smith *et al*⁶², lung buds begin to develop 3 weeks post-conception during the embryonic period. At roughly 6 weeks, lung development moves into the pseudoglandular stage where branching and budding give rise to the stereotypic structure of the lung. The canalicular phase, with conducting and alveolar epithelial cell differentiation, begins at week 16 and lasts until week 26 where the saccular phase begins. Finally, the alveolar period of lung development begins right before birth and continues postnatally. This has been accepted to be up to 2-3 years postnatally, however current studies suggest the lung could still be developing through childhood and adolescence, although at a much slower rate^{63–65}.

1.4.3 Foetal programming of the respiratory system

Lung development can be influenced by foetal programming through several stimuli, including smoking and malnutrition. Smoking during pregnancy has been linked to foetal programming disrupting normal growth patterns and causing respiratory restructuring. Maternal smoking may have many inducers of foetal programming, including inhaled chemicals crossing the placental barrier, but it also causes an increase in carboxyhemoglobin and reduced placental blood flow leading to foetal hypoxia^{66–68}. In a rat model of maternal smoking, offspring had foetal growth restriction, lung hypoplasia with increased indications of emphysema⁶⁹. Similar outcomes were witnessed in humans with prenatal smoking associated with airway remodelling and the development of asthma^{70–74}.

Inadequate maternal nutrition during pregnancy can also cause significant disruption to growth patterns and lung development. In a rat model of maternal malnutrition, starved pups had reduced body weight, lung cell differentiation and immature air-blood barrier at birth. Enzymatic activity of alveolar macrophages was also reduced posing implications for post-birth response to infection⁷⁵. Malnutrition can also alter lung lipid differentiation, lung morphology, and lung function that can persist into adulthood^{76,77}. However, it may be possible that foetal programming stems from the reduced growth rate and not the stimuli *per se*.

In a multivariant analysis, adjusting for multiple confounding factors, it was low birth weight and not smoking status, maternal height or family history of asthma, that was associated with reduced lung function (forced expiratory volume, FEV, in 0.4 seconds)⁷⁸. Thus, regardless of the stimuli, if it is potent enough to cause intrauterine growth restriction then respiratory development could be affected. This supports Barker *et al* in that low birth weight and low weight at 1 year of age were associated with a worse respiratory function in adulthood, with lower birthweights having lower FEV at 1 second (FEV₁), regardless of smoking status, social status, or early childhood respiratory infection⁷⁹. Turner *et al* provided additional data in that reduced body size was associated with reduced FEV₁, forced vital capacity and expiratory flow rate⁸⁰. Adult asthma, wheezing and bronchial hyperreactivity have also been linked to low birthweight^{81,82} and, counterintuitively, asthma severity is associated with an increased head circumference⁸³. It could be expected that reduced body size would equate to reduced head circumference however, the foetus may adapt to preserve cerebral blood and head circumference by redistributing blood flow and nutrients to the brain, at the cost of the rest of the body's development, resulting in asymmetric growth restriction⁸⁴. Twin control cohorts have strengthened these studies by identifying that there is a relationship between reduced birth weight and the development of asthma in both monozygotic and dizygotic twins. These results reinforce the foetal programming hypothesis that reduced foetal growth negatively impacts lung development independent of environment or genetic factors⁸⁵.

There are therefore many factors that may cause respiratory foetal programming, with associated intrauterine growth restriction, including environmental, disease, dietary, or stress-related stimuli^{66–72,75–77,85–88}. One environmental stimulus that has yet to be explored is ionizing radiation during pregnancy to assess if it causes intrauterine growth restriction and thereby possible respiratory changes. Unlike other methods of foetal programming, like maternal hypotension or dietary restrictions, the effect of radiation may not only occur indirectly through the mother's exposure and foetal programming but also directly on the foetus itself. As X-rays and gamma rays penetrate the body, they will inevitably also penetrate the foetus. Thus, the effects of ionizing radiation during pregnancy may encompass both foetal programming and direct effects.

1.4.4 Foetal radiation exposure

When the foetus is in the field of view of medical procedures it will be directly exposed, however even when outside the field of view, the foetus may still be exposed via scatter radiation, but this dose would be minimal in comparison. Therefore, procedures of the abdomen and pelvis carry larger doses than those of the chest or head (Table 1-5). In addition, the mother's skin and tissues provide some level of shielding, with increased maternal perimeter and foetal depth in the body associated with decreased foetal dose⁸⁹

Table 1-5 outlines foetal doses following common diagnostic procedures; however, these numbers were calculated in the 1990s using different technology to current modalities. The present technologies allow for better dose reduction and control and so the current exposures would presumably be less⁹⁰. Although these doses from medical radiation typically fall below the threshold of deterministic effects, such as lethality or malformation development, other factors can influence the effect the radiation dose has on the foetus including the type of radiation, as mentioned previously, and timing of exposure⁹¹.

The effect of the radiation on the foetus differ significantly based on the length post-conception and therefore developmental stage of the foetus. In the pre-implantation period, death to the foetus is the dominant effect from large radiation doses, typically from cytogenetic damage. Exposures early in gestation of > 1000 mGy significantly induce embryonic death, whereas exposures later in gestation required higher levels of radiation to attain LD_{50/30}, showing the sensitivity to radiation at the preimplantation stage⁹². Interestingly, the LD_{50/30} varies significantly (1000 – 5500 mGy) and not in a linear fashion⁹². Exposures during organogenesis and foetal development periods tend to produce

malformations or growth retardation but lessen with higher gestational age⁹³. Data from survivors of the Hiroshima and Nagasaki bombing demonstrate that there is a significant shift in effect with respect to the timing of exposure and formation of microcephaly, small head circumference. Most microcephaly cases were induced from foetal doses >100 mGy, but the frequency reduces from 1st trimester to 2nd to 3rd, independent of dose exposure, demonstrating the importance of timing in regards exposure⁹⁴.

In clinical settings, the physical parameters that change with gestation can also influence exposure. For indirect clinical exposures of diagnostic radiation, through a chest CT, the foetal dose from scattered radiation increases with gestational age⁹⁶, presumably due to a larger foetus being closer to the field of view during the procedure. For direct exposures, such as abdominal or pelvic examinations, a larger mass means a decrease in foetal depth and therefore less shielding from the mother's facia, increasing the dose received by the foetus⁸⁹.

The importance of gestational timing during exposure to ionizing radiation is clear but what are the actual effects of such exposures during the three main stages of development: pre-implantation, embryogenic and foetal? These effects can be either deterministic, occurring relatively close after exposure, or stochastic, taking many years to come to fruition.

| | Mean (mGy) | Maximum (mGy) |
|---------------------|------------|---------------|
| CONVENTIONAL X-RAY | | |
| Skull | <0.01 | <0.01 |
| Thoracic spine | <0.01 | <0.01 |
| Chest | <0.01 | <0.01 |
| Lumbar spine | 1.7 | 10 |
| Abdomen | 1.4 | 4.2 |
| Pelvis | 1.1 | 4 |
| FLUOROSCOPY | | |
| Barium meal | 1.1 | 5.8 |
| Barium enema | 6.8 | 24 |
| Intravenous urogram | 1.7 | 10 |
| COMPUTED TOMOGRAPHY | | |
| Head | <0.01 | <0.01 |
| Chest | 0.06 | 0.96 |
| Lumbar spine | 2.4 | 8.6 |
| Abdomen | 8 | 49 |
| Pelvis | 25 | 79 |

Table 1-5: Foetal doses following common diagnostic procedures taken from UK surveys of diagnosticradiology and adapted from Sharp, Shrimpton and Bury with copywrite permission grantedMean (mGv)Maximum (mGv)

mSv = millisieverts

1.4.5 Deterministic effects

The deterministic effects of radiation have been known for 90 years from when Goldstein and Murphy observed a high incidence of neonatal cranial malformations in pregnant women who received radiotherapy, estimated >1000 mGy, for uterine cancer⁹⁷. Many studies identifying the deterministic effects of *in utero* exposure to ionizing radiation typically look at outcomes including lethality, growth restrictions, malformations, and neurological function, but these outcomes can vary in severity depending on the timing of exposure during development.

1.4.5.1 Experimental models – Preimplantation

The preimplantation stage of development can often go unrecognised in clinical settings due to this very early stage of pregnancy being missed during diagnosis and thus there is a scarcity of human epidemiological data with most conclusions coming from animal studies. From 0-8 days post-conception of rodents, the foetus appears to be less radiosensitive to deterministic effects. In this time, the effect of radiation is typically reserved to lethality, a failure of embryo implantation or early abortion, or no long term effect at all: this is known collectively as the all-or-nothing theory⁹⁸. This idea presumes the effects either cause lethality or repair/replacement of damaged cells via a repair mechanism or totipotent stem cell differentiation^{98,99}. Doses above 500 mGy result in embryonic fatality and reduced litter size, but the likelihood of malformation during this time is low^{92,93,100,101}. This may be because, during preimplantation exposures, only the extraembryonic tissues are exposed, as the primitive streak doesn't develop until the 2nd

week. The structural features of the body are yet to develop and thus malformation effects are unlikely to arise. The contradictory evidence to this statement comes from animal studies where there are species and sex variations, and only occurs for doses higher than 100 mGy, but nonetheless, fatality remains the greatest risk associated during the preimplantation stage of development^{102–104}.

1.4.5.2 Experimental models – Embryonic

Fatality is still a prominent effect of radiation during the embryonic development stage, however, the incidence of intrauterine growth restriction and malformation, particularly neuropathological, are more likely. One study with swiss albino mice showed exposures that 50 mGy had increased mortality from 10% to 16%, possibly due to high dose rate (900 mGy/min)¹⁰⁵ however, most other studies have a much higher dose threshold (>500 mGy)^{100,101,106,107}. Birth weight was reduced for rats and mice exposed to > 450 mGy at both early and late time periods in this developmental stage, suggesting intrauterine growth restriction had occurred^{100,108–111}, but only doses > 1000 mGy result in a sustained lighter weight later in life¹¹⁰.

Malformations and physiological changes appear to be variable and inconsistent. A rat study found exposures up to 800 mGy, far above diagnostic exposures, 9 days post-conception resulted in no physiological or preweaning reflex changes but a mouse studies with 500 mGy 9 days post-conception did for similar outcomes^{112,113}. Likewise, other studies in rodents > 500 mGy show physical abnormalities of the face or appendages, and

also some motor-neurological abnormalities, such as loss of limb control^{92,100,101}. In human, Kinlen and Acheson conducted a case-control study of 605 children with malformations and found no association with prenatal diagnostic radiation¹¹⁴, which was supported by Ornoy *et al*¹¹⁵. Similarly, a prospective study showed no higher incidence of malformation, or mortality, compared to the general population¹¹⁶. A recent study identified that maternal exposure to ionizing radiation during conception and early gestation resulted in birth defects in 2/19 categories however lacked the power to remain after more scrutinous statistical analysis¹¹⁷.

In humans, as with rodents: some studies have found significant increases in cranial malformations, including microcephaly; decreased cognitive function; or a change in behaviours, including anxiety and nervousness, with a threshold dose > 100 mGy^{92,94,100,101,111,115,118}. The neurological changes are not just short term and can vary based on the dose received or the timing of the exposure^{119–121}. For example, mice irradiated with 100 or 200 mGy during late embryonic development can have behavioural changes that last up to 13 months post-birth, but those exposed to 500 or 1000 mGy only manifest changes after 20 months^{120,121}. Therefore, although there are inconsistencies with the report of malformations and neurobiological defects, it appears there is a consensus that these events only occur from doses above 100 mGy, which is far greater than received from diagnostic procedures.

1.4.5.3 Experimental models – Foetal Development

The effects of radiation during the foetal development stage (2nd and 3rd trimesters) are observed less, with a higher dose threshold required to elicit similar responses to the previous stages. Postnatal fatality in rodents does not increase with 500 mGy exposures but does with exposures > 1000 mGy^{100,122–124}. Human data available in the early to mid-20th century demonstrated radiation-induced fatality but was not 100% effective. Using a higher dose in the region of 5000 mGy, clinicians would use irradiation as a method to induce abortion¹²⁵. In humans and rodents, intrauterine growth restriction from exposures to radiation is still possible at < 1000 mGy, but they usually recover 1-2 weeks postnatally, – the reduced weight only persists when the foetus is exposed to > 1000 mGy^{107,111,122–124}.

Similar to the embryonic period, exposures during the foetal development period have variable and inconsistent effects on formation or physiological outcomes. Exposures of 500 mGy in swiss albino mice or 2000 mGy in ICR mice during foetal development showed no malformations^{100,101}, however, a threshold dose of 0.5 mGy in Swiss albino mice in another study caused cranial and appendage malformations¹²³. In Wistar rats, doses above 400 mGy resulted in physiological changes, including cranial malformations, and reduced preweaning reflexes^{112,124}. Preweaning and motor reflexes and motor strength in Sprague-Dawley rats were not different from controls when exposed up to 800 mGy¹²⁶. Other neurological outcomes including cognitive ability, learning memory, and other behaviour changes have a threshold effect of 250 mGy, with the persistence of effects only seen > 500 mGy^{127–129}. In humans, exposures above 100 mGy early in this period can cause microcephaly, but later gestational exposures appear not to have any effect regardless of dose⁹⁴. Additionally, Otake *et al* re-evaluated a threshold dose for severe mental retardation in the atomic bomb survivors to 60 - 310 mGy (95% Cl) for early foetal development and 250 – 280 mGy for later gestations¹³⁰. Although there are variations in malformations and neurobiological defects, it appears there is a consensus that there is a threshold dose > 100 mGy to which these effects occur. For ease of view, the American College of Radiology tabulated suspected effects from ionizing radiation dictated from ICRP reports¹³¹, Table 1-6.

Amongst all periods of exposure, preimplantation, embryonic and foetal development, it is apparent there is a large variation in outcomes based on species, strain and sex. For instance, Nash *et al.* found a large variation in results when comparing inbred mice and hybrid genotypes indicating varying levels of radioresistance among different strains as well as sex¹⁰⁶. For a wider comparison, the LD_{50/30} can vary from 1730 – 8000 mGy also based on the differences in sex, species and strain^{132–141}. This variation in radiosensitivity is most likely the reason for conflicting results above which it is difficult to draw meaningful and translatable-to-human conclusions. It is therefore important that more research is conducted, with species-specific radiosensitivities in mind, to fill these gaps.

| Gestational Age | <50 mGy | 50 – 100 mGy | >100 mGy |
|--------------------|---------|--|---|
| 0 – 2 wks | None | None | None |
| 3 – 4 wks | None | Probably none | Possibly spontaneous abortion |
| 5 – 10 wks | None | Potential effects are scientifically uncertain and probably too subtle to be clinically detectable | Possible malformations increasing in likelihood as dose increases. |
| 11 – 17 wks | None | Potential effects are scientifically uncertain and probably too subtle to be clinically detectable. | Risk of diminished IQ or of mental retardation, increasing in frequency and severity with increasing dose. |
| 18 – 27 wks | None | None | IQ deficits not detectable at diagnostic doses. |
| >27 wks | None | None | Nonapplicable to diagnostic medicine |

Table 1-6: Summary of suspected *in utero* ionizing radiation-induced deterministic effects, adapted with copywrite permission from ACR¹³¹.

mGy = milligray, wks = weeks

1.4.6 Stochastic effects

The long-term effects of ionizing radiation arise from stochastic effects. These originate from damage to the genetic material of a cell in terms of mutations. The level of DNA damage increases with radiation dose and the levels of mutations are predicted to be proportional to the level of DNA damage^{142–144}. When these mutations originate in germline cells, there is an association with hereditary effects, and when these mutations originate in somatic cells there is an association to cancer development¹⁴⁴. As stated by Kamiya¹⁴⁵, it is important to delineate that 'the probability, and not the severity, of stochastic effects increase with radiation dose.'

The Life-Span Study (LSS) is the largest epidemiology study of the atomic bomb survivors allowing for the best understanding of the effects of radiation on the human population. The study examined 94,000 survivors who were <10 km from the hypocentres of either Hiroshima or Nagasaki and 26,000 controls who were out of the cities at the time of exposure. Of the 92% of exposed individuals who had their exposure successfully estimated, the majority received doses <100 mGy (79%), while 55% had less than 5 mGy; only 2.8% received a dose => 1000 mGy¹⁴⁶. The risk of hereditary effects and cancer from ionizing radiation in humans has largely been estimated from epidemiological studies of the atomic bomb survivors, however, the gestational timing of pregnant women during these exposures is much harder to elucidate as the incidence of effects are small.

The risk of hereditable effects during *in utero* exposures from ionizing radiation is difficult to calculate as there has not been a study to investigate the F2 generation. However, the children (F1) born after parental atomic bomb exposure had no increase in malformations, stillbirths or perinatal death post-birth, and no increased risk of death caused by cancer, death by non-cancer disease, or common adult-onset multifactorial diseases including hypertension, diabetes, heart disease or stroke^{147–152}. The mutation rate in children born from exposed survivors is also not different and as reviewed by Neel et al. there is no evidence of any genetic effects from radiation occurring in these children^{153,154}. For the F1 generation born after parental exposure, the association between these diseases and radiation could only come from the radiation's effect on the germline cells, indicating that the exposures the parents received had no adverse effects on their germline cells. An assumption could then be made that *in utero* exposed foetuses, who carry their germ cells or progenitors at that time of exposure, would also have no increase in stochastic events in their offspring, however, there is no evidence to support this notation currently.

Cancer risk during *in utero* exposure is also difficult to calculate but estimates come from both the LLS and clinical cohorts as well. Data from the 807 *in utero* exposures of the LSS showed only 1 cancer death in the first 15 years of life and 10 more up to 46 years old, most being female^{155,156}. These studies from the LSS conclude a slight increase in solid cancer rates, but not leukemia, in those exposed *in utero*. However, caution was recommended in the interpretation of results due to numerous limiting factors including low cancer death incidence, sex differences, and no apparent dose-response variation¹⁵⁶. One large clinical study collated data from abdominal and pelvic irradiations during the 1950s and reported an increased rate of childhood cancers and leukemia in their offspring¹⁵⁷. Recent reviews have stated, with limited certainty, that doses >10 mGy to the foetus may cause a small increase in childhood carcinogenesis^{158,159}, however, another study found no statistical increase¹⁶⁰. This variation could be due to the decade's difference between studies in which presumably increased awareness of radiation risk resulted in a reduction in the number of radiological scans received during pregnancy, including in the controls. Thus, the later study would have reduced foetal exposure and consequently saw no significant increase in cancer risk. It is important to recognise that, as reviewed and reported by Boice and Miller¹⁶¹, the majority of the evidence supporting an association comes from case-controlled studies whereas cohort studies typically find no association. Case-controlled studies observe the outcome (cancer) and then identify which patients received the exposure (radiation), repetitively seen in clinical studies/observations, whereas cohort studies reverse this order; observe the exposure (radiation) and follow to see the outcome (cancer), such as the LSS. Cohort studies are powerful but require large patient numbers to observe rare events. In contrast, case-control studies are good at assessing rare events with small patient numbers but can typically have a reporting bias in patients recalling if they had received the exposure decades ago. This bias is not as evident in cohort studies which may play some role in the difference between results from these methods¹⁶².

A final point to note is that these two populations, pregnant patients who received diagnostic radiation and those that don't, resemble two subpopulations that are inherently different. One medically requires radiation for diagnostic purposes, thus there is clinical justification. Totter and MacPherson argue that the selection factor of the medical decision to prescribe diagnostic radiation, and not the radiation itself, might be responsible for the associated risks¹⁶³. Therefore, it is relatively difficult to epidemiologically delineate the effect of radiation and thus interpretation of such data should be met with caution.

Overall, the integrity and the reliability of the low dose data used in epidemiological studies causes uncertainty in the robustness of the risk estimates, but all who conclude some level of association infer that the risk is not zero. Furthermore, across all periods of development, there appears to be a paucity of research conducted on the deterministic and stochastic effect of radiation on respiratory development, thus there is a clear gap in the literature that needs to be investigated.

1.5 THESIS GOALS

The overall objective of this thesis was to understand what level of exposures patients are receiving whilst admitted to hospital and the potential effects these levels of radiation can have, particularly for the respiratory system. To investigate this goal, clinical levels of diagnostic radiation were first recorded then used in animal models to understand more about its effect.

Clinically, it is important that clinicians have a firm understanding of the risks and benefits to any test or intervention they prescribe. General radiology has to potential to be lifesaving as a diagnostic tool, but clinicians are not as firm with the understanding of its risks. However, before considering the risks of diagnostic radiation, it is necessary to understand the doses being prescribed. For the reason that doses are very patient-centric and clinically justified, two populations at polar ends of the patient spectrum were selected to understand how much radiation admitted patients would receive; those who are the sickest and require constant diagnosis and management of disease, ICCU patients, vs those who are typically restricted in radiographic prescription due to the fear of deterministic and stochastic effects, pregnant patients and their unborn foetus.

The convention surrounding irradiations of pregnant patients is to limit the dose to the foetus to as low as reasonably achievable due to the risks from ionizing radiation, however, these risks are interpreted from much higher doses of ionizing radiation than that which comes from diagnostic radiation. Research into low dose radiation during pregnancy is emerging and showing little effect on the development of the foetus, however, there is a paucity of data on the respiratory system. To understand these effects, healthy and acute lung injury mouse models were used to investigate if *in utero* exposure to diagnostic levels of ionizing radiation altered the long-term respiratory physiology and immunology.

Together, this thesis wants to explore low dose ionizing radiation during development and its long-term effect on the respiratory system, with specific goals to

- 1. Examine the cumulative levels of ionizing radiation from diagnostic radiation received by patients admitted to the ICCU.
- 2. Examine the cumulative levels of ionizing radiation by diagnostic radiation received by pregnant patients admitted to hospital and the resulting foetal exposures.
- 3. Examine the impact of diagnostic levels of ionizing radiation during late gestation pregnancy on the long-term development of the respiratory system in a healthy mouse model.
- 4. Examine the impact of diagnostic levels of ionizing radiation during late gestation pregnancy on the physiological and immunological response to an acute lung injury stimulus at adolescence in a mouse model.

39

1.6 CHAPTER SUMMARY

With a focus on medical uses, this thesis will highlight the clinical apprehensiveness between ionizing radiation and pregnancy, with the intent of informing clinicians to improve risk to benefit analysis of prescribing these procedures. Through the use of mouse models, this work significantly contributed to the current understanding of the long-term effects and aligns with the ideology that the risks associated with low doses ionizing radiation, such as diagnostic radiation, are negligible compared to the benefit of performing the procedure and possibly saving the patient's life.

Chapter 1 is a general introduction to the themes covered in this thesis including ionizing radiation and pregnancy. This chapter also summarises the current knowledge of ionizing radiation during pregnancy highlighting the paucity of data specifically relating to the respiratory system.

Chapter 2 investigated the levels of ionizing radiation received by patients admitted to the ICCU. Radiological records from 526 patients were assessed to calculate the cumulative effective dose for each patient. Clinical and demographic characteristics were also recorded to assess association with cumulative effective doses. This study helped comprehend the levels of ionizing radiation received by the sickest patients who frequently required these procedures. Chapter 3 investigated the levels of ionizing radiation received by pregnant patients admitted to hospital. Radiological records from 557 patients were assessed to calculated cumulative effective doses. The procedure exposures were then used to calculate foetal exposures. Again, clinical and demographic characteristics were investigated for an association to dose. This study identified the levels of exposures received by patients who sparingly received ionizing radiation.

Chapter 4 used a healthy animal model to examine the effect of *in utero* exposure to ionizing radiation during late gestation. The diagnostic levels of ionizing radiation used, replicate clinical settings and late gestation was chosen to avoid mortality and major malformations and focus on foetal programming effects and functional changes. This study highlighted the temporal effects on the cardiovascular system as well as the long-term response to the respiratory system.

Chapter 5 also used a mouse model but induced acute lung injury, through intratracheal instillation of a respiratory stimulus, to examine if *in utero* exposures to ionizing radiation altered physiological and immunological responses during inflammation. Similar diagnostic levels of ionizing radiation were used but the introduction of acute sickness allows for investigation into the immunomodulatory response of ionizing radiation, and if this response is long term.

Chapter 6 summarises the key findings from the experimental chapters (chapter 2-5) and discusses how these data support the negligible effects, and thus risk, diagnostic levels of radiation during pregnancy have on the respiratory system. It also discusses perspectives on the findings and potential routes for future research.

Chapter 7 summarises the experience received from embarking on a Cotutelle de Thesis PhD, highlighting the struggles and benefits from participating in this crossinstitutional arrangement. This chapter also emphases the value this involvement has added, both academically and professionally.

1.7 REFERENCES

- 1. United Nations Scientific Committee on the Effects of Atomic Radiation. *Sources and Effects of Ionizing Radiation: Vol 1 Annex B.*; 2008.
- The International Commission on Radiological Protection. The 2007 Recommendations of the International Commission on Radiological Protection. ICRP Publication 103. Ann ICRP. 2007;37(2-4). doi:10.1177/ANIB_37_2-4.
- 3. Lin EC. Radiation Risk From Medical Imaging. *Mayo Clin Proc.* 2010;85(12):1142-1146. doi:10.4065/mcp.2010.0260.
- 4. Australian Radiation Protection and Nuclear Safety Agency. Ionising Radiation and Health What about artificial sources of radiation? *Aust Radiat Prot Nucl Saf Agency*.
- 5. Grasty RL, LaMarre JR. The annual effective dose from natural sources of ionising radiation in Canada. *Radiat Prot Dosimetry*. 2004;108(3):215-226. doi:10.1093/rpd/nch022.
- 6. Sohrabi M, Esmaili A. R. New public dose assessment of elevated natural radiation areas of Ramsar (Iran) for epidemiological studies. *Int Congr Ser*. 2002;1225:15-24. doi:10.1016/S0531-5131(01)00528-3.
- 7. Zou J, Tao Z, Sun Q, Akiba S, Zha Y, Sugahara T, Wei L. Cancer and non-cancer epidemiological study in the high background radiation area of Yangjiang, China. *Int Congr Ser*. 2005;1276:97-101. doi:10.1016/j.ics.2004.11.167.
- Veiga LH., Koifman S, Melo VP, Sachet I, Amaral EC. Preliminary indoor radon risk assessment at the Poços de Caldas Plateau, MG – Brazil. J Environ Radioact. 2003;70(3):161-176. doi:10.1016/S0265-931X(03)00101-2.
- 9. Hendry JH, Simon SL, Wojcik A, Sohrabi M, Burkart W, Cardis E, Laurier D, Tirmarche M, Hayata I. Human exposure to high natural background radiation: what can it teach us about radiation risks? *J Radiol Prot*. 2009;29(2A):A29-42. doi:10.1088/0952-4746/29/2A/S03.
- 10. Rafnsson V, Hrafnkelsson J, Tulinius H. Incidence of cancer among commercial airline pilots. *Occup Environ Med*. 2000;57(3):175-179. doi:10.1136/oem.57.3.175.
- 11. Marušiaková M, Gregor Z, Tomášek L. A review of exposures to radon, long-lived radionuclides and external gamma at the Czech uranium mine. *Radiat Prot Dosimetry*. 2011;145(2-3):248-251. doi:10.1093/rpd/ncr065.
- 12. Ministry of Health R. 2014 Draft MCH Annual Report (July 2013 to June 2014). 2014;(June).
- 13. Internation Atomic Energy Agency. Radiation protection and safety of radiation sources: international basic safety standards general safety: No GSR part 3. *IAEA Saf Stand*. 2014;3:436. doi:STI/PUB/1578.
- 14. United Nations Scientific Committee on the Effects of Atomic Radiation. *Sources and Effects of Ionizing Radiation*. Vol 2.; 2018. doi:10.18356/f48d655d-en.
- 15. United Nations Scientific Committee on the Effects of Atomic Radiation. *Summary of Low-Dose Radiation Effects* on Health. UNSCEAR 2010.; 2010.
- 16. Mothersill C, Rusin A, Seymour C. Towards a New Concept of Low Dose. *Health Phys.* 2019;117(3):1. doi:10.1097/hp.00000000001074.
- 17. Little MP, Wakeford R, Tawn EJ, Bouffler SD, Berrington de Gonzalez A. Risks Associated with Low Doses and Low Dose Rates of Ionizing Radiation: Why Linearity May Be (Almost) the Best We Can Do. *Radiology*. 2009;251(1):6-12. doi:10.1148/radiol.2511081686.
- 18. Calabrese EJ, O'Connor MK. Estimating Risk of Low Radiation Doses A Critical Review of the BEIR VII Report and its Use of the Linear No-Threshold (LNT) Hypothesis. *Radiat Res.* 2014;182(5):463-474. doi:10.1667/rr13829.1.
- 19. Prasad KN, Cole WC, Hasse GM. Health Risks of Low Dose Ionizing Radiation in Humans: A Review. *Exp Biol Med*. 2004;229(5):378-382. doi:10.1177/153537020422900505.

- 20. Bushberg JT, Seibert JA, Leidholdt EMJ, Boone JM. *The Essential Physics of Medical Imaging*.; 2011.
- 21. Mettler FA, Huda W, Yoshizumi TT, Mahesh M. Effective doses in radiology and diagnostic nuclear medicine: a catalog. *Radiology*. 2008;248(1):254-263. doi:10.1148/radiol.2481071451.
- 22. Royal College of Radiologist and National Radiation Protection Board. *Patient Dose Reduction in Diagnostic Radiology*. Vol 1.; 1990.
- 23. Shrimpton PC, Jones DG, Hillier MC, Wall BF, Le Heron JC, Faulkner K. Survey of CT practice in the UK. Pt. 2. 1991.
- Lutterman AC, Moreno CC, Mittal PK, Kang J, Applegate KE. Cumulative Radiation Exposure Estimates of Hospitalized Patients from Radiological Imaging. J Am Coll Radiol. 2014;11(2):169-175. doi:10.1016/j.jacr.2013.08.028.
- 25. Shrimpton PC, Hillier MC, Meeson S, Golding SJ. *Doses from Computed Tomography (CT) Examinations in the UK* 2011 Review.; 2011.
- 26. Wambani JS, Korir GK, Tries MA, Korir IK, Sakwa JM. Patient radiation exposure during general fluoroscopy examinations. *J Appl Clin Med Phys.* 2014;15(2):262-270. doi:10.1120/JACMP.V15I2.4555.
- 27. Mahesh M. Fluoroscopy: Patient Radiation Exposure Issues. *RadioGraphics*. 2001;21(4):1033-1045. doi:10.1148/radiographics.21.4.g01jl271033.
- 28. Abramowicz JS. Benefits and risks of ultrasound in pregnancy. *Semin Perinatol.* 2013;37(5):295-300. doi:10.1053/j.semperi.2013.06.004.
- 29. Pott P. Chirurgical Observations : Relative to the Cataract, the Polypus of the Nose, the Cancer of the Scrotum, the Different Kinds of Ruptures, and the Mortification of the Toes and Feet. London: Printed by T.J. Carnegy, for L. Hawes, W. Clarke, and R. Collins; 1775.
- 30. Muller HJ. Artificial transmutation of the elements. *Nature*. 1933;131(3319):832. doi:10.1126/science.66.1699.84.
- 31. Timoféeff-Ressovsky N, Zimmer E, Delbrück M. Über die Natur der Genmutation und der Genstruktur. *Nachricht Biol Gesell Wissen*. 1935;1:189-245.
- 32. Stadler LJ. On the Genetic Nature of Induced Mutations in Plants. *Proc Sixth Int Congr Genet*. 1932;1:274-294.
- 33. McClintock B. Induction of instability at selected loci in Maize. *Genetics*. 1953;38(6):579-599.
- 34. Calabrese EJ. Flaws in the LNT single-hit model for cancer risk: An historical assessment. *Environ Res.* 2017;158:773-788. doi:10.1016/j.envres.2017.07.030.
- 35. Caspari E, Stern C. The Influence of Chronic Irradiation with Gamma-Rays at Low Dosages on the Mutation Rate in Drosophila Melanogaster. *Genetics*. 1948;33(1):75-95.
- 36. Burlakova EB. Low Intensity Radiation: Radiobiological Aspects. *Radiat Prot Dosimetry*. 1995;62(1-2):13-18. doi:10.1093/oxfordjournals.rpd.a082810.
- 37. Vaiserman AM. Radiation hormesis: Historical perspective and implications for low-dose cancer risk assessment. *Dose-Response*. 2010;8(2):172-191. doi:10.2203/dose-response.09-037.Vaiserman.
- 38. National Council on Radiation and Measurements NCRP. *Limitations of Exposure to Ionizing Radiation. NCRP Report No. 116.*; 2012.
- 39. Australian Radiation Protection and Nuclear Safety Agency. *Position Statement on the Use of the Linear No-Threshold Model in Radiation Protection.*; 2017.
- 40. Tubiana M, Feinendegen LE, Yang C, Kaminski JM. The linear no-threshold relationship is inconsistent with radiation biologic and experimental data. *Radiology*. 2009;251(1):13-22. doi:10.1148/radiol.2511080671.
- 41. Averbeck D. Does scientific evidence support a change from the Int model for low-dose radiation risk extrapolation? In: *Health Physics*. Vol 97. ; 2009:493-504. doi:10.1097/HP.0b013e3181b08a20.
- 42. Calabrese EJ. The threshold vs LNT showdown: Dose rate findings exposed flaws in the LNT model part 1. The

Russell-Muller debate. Environ Res. 2017;154:435-451. doi:10.1016/J.ENVRES.2016.12.006.

- 43. Beral V, Gilbert E, Mabuchi K, Sinclair W, Tao Z. Low-dose extrapolations of radiation related cancer risk. *ann ICRP*. 2005.
- 44. Seong KM, Kwon TW, Seo S, Lee D, Park S, Jin YW, Lee SS. Perception of low dose radiation risks among radiation researchers in Korea. *PLoS One*. 2017;12(2):e0171777. doi:10.1371/journal.pone.0171777.
- 45. Lee CI, Haims AH, Monico EP, Brink JA, Forman HP. Diagnostic CT Scans: Assessment of Patient, Physician, and Radiologist Awareness of Radiation Dose and Possible Risks. *Radiology*. 2004;231(2):393-398. doi:10.1148/radiol.2312030767.
- 46. Correia MJ, Hellies A, Andreassi MG, Ghelarducci B, Picano E. Lack of radiological awareness among physicians working in a tertiary-care cardiological centre. *Int J Cardiol.* 2005;103(3):307-311. doi:10.1016/j.ijcard.2004.08.070.
- 47. Heyer C, Peters S, Lemburg S, Nicolas V. Einschätzung der Strahlenbelastung radiologischer Thorax-Verfahren: Was ist Nichtradiologen bekannt? *RöFo - Fortschritte auf dem Gebiet der Röntgenstrahlen und der Bildgeb Verfahren*. 2007;179(3):261-267. doi:10.1055/s-2006-927366.
- Rice HE, Frush DP, Harker MJ, Farmer D, Waldhausen JH, APSA Education Committee. Peer assessment of pediatric surgeons for potential risks of radiation exposure from computed tomography scans. *J Pediatr Surg*. 2007;42(7):1157-1164. doi:10.1016/j.jpedsurg.2007.02.007.
- 49. Soye JA, Paterson A. A survey of awareness of radiation dose among health professionals in Northern Ireland. *Br J Radiol*. 2008;81(969):725-729. doi:10.1259/bjr/94101717.
- 50. Shiralkar S, Rennie A, Snow M, Galland RB, Lewis MH, Gower-Thomas K. Doctors' knowledge of radiation exposure: questionnaire study. *BMJ*. 2003;327(7411):371-372. doi:10.1136/bmj.327.7411.371.
- 51. Oba Y, Zaza T. Abandoning Daily Routine Chest Radiography in the Intensive Care Unit: Meta-Analysis. *Radiology*. 2010;255(2):386-395. doi:10.1148/radiol.10090946.
- 52. Ganapathy A, Adhikari NK, Spiegelman J, Scales DC. Routine chest x-rays in intensive care units: a systematic review and meta-analysis. *Crit Care*. 2012;16(2):R68. doi:10.1186/cc11321.
- 53. Cruz J, Ferra M, Kasarabada A, Gasperino J, Zigmund B. Evaluation of the Clinical Utility of Routine Daily Chest Radiography in Intensive Care Unit Patients With Tracheostomy Tubes. J Intensive Care Med. 2016;31(5):333-337. doi:10.1177/0885066614538393.
- 54. Amorosa JK, Bramwit MP, Mohammed T-LH, Reddy GP, Brown K, Dyer DS, Ginsburg ME, Heitkamp DE, Jeudy J, Kirsch J, MacMahon H, Ravenel JG, Saleh AG, Shah RD. ACR Appropriateness Criteria Routine Chest Radiographs in Intensive Care Unit Patients. J Am Coll Radiol. 2013;10(3):170-174. doi:10.1016/j.jacr.2012.11.013.
- 55. The International Commission on Radiological Protection. *Pregnancy and Medical Radiation. ICRP Publication* 84. Vol 30. Published for the International Commission on Radiological Protection by Pergamon; 2000.
- 56. Sadler TW. Langman's Essential Medical Embryology.; 2012. doi:10.1016/j.biocel.2015.02.002.
- 57. Theiler K. The House Mouse: Atlas of Embryonic Development.; 1989. doi:10.1007/978-3-642-88418-4.
- 58. Barker D. The fetal and infant origins of adult disease. *BMJ Br Med J.* 1990;301(156):1111. doi:10.1136/bmj.301.6761.1111.
- 59. Barker DJP, Godfrey KM, Gluckman PD, Harding JE, Owens JA, Robinson JS. Fetal nutrition and cardiovascular disease in adult life. *Lancet*. 1993;341(8850):938-941. doi:10.1016/0140-6736(93)91224-A.
- 60. Maeda Y, Davé V, Whitsett JA. Transcriptional control of lung morphogenesis. *Physiol Rev.* 2007;87(1):219-244. doi:10.1152/physrev.00028.2006.
- 61. Pinkerton KE, Joad JP. The mammalian respiratory system and critical windows of exposure for children's health. *Environ Health Perspect*. 2000;108(SUPPL. 3):457-462. doi:10.1289/ehp.00108s3457.
- 62. Smith LJ, McKay KO, van Asperen PP, Selvadurai H, Fitzgerald DA. Normal development of the lung and

premature birth. Paediatr Respir Rev. 2010;11(3):135-142. doi:10.1016/j.prrv.2009.12.006.

- 63. Thurlbeck WM. Postnatal human lung growth. *Thorax*. 1982;37(8):564-571. doi:10.1136/thx.37.8.564.
- 64. Herring MJ, Putney LF, Wyatt G, Finkbeiner WE, Hyde DM. Growth of alveoli during postnatal development in humans based on stereological estimation. *Am J Physiol Cell Mol Physiol*. 2014;307(4):L338-L344. doi:10.1152/ajplung.00094.2014.
- 65. Narayanan M, Owers-Bradley J, Beardsmore CS, Mada M, Ball I, Garipov R, Panesar KS, Kuehni CE, Spycher BD, Williams SE, Silverman M. Alveolarization continues during childhood and adolescence: New evidence from helium-3 magnetic resonance. *Am J Respir Crit Care Med*. 2012;185(2):186-191. doi:10.1164/rccm.201107-13480C.
- 66. de B Machado J, Filho PVM, Petersen GO, Chatkin JM. Quantitative effects of tobacco smoking exposure on the maternal-fetal circulation. *BMC Pregnancy Childbirth*. 2011;11(1):24. doi:10.1109/CISP.2008.339.
- 67. Ko TJ, Tsai LY, Chu LC, Yeh SJ, Leung C, Chen CY, Chou HC, Tsao PN, Chen PC, Hsieh WS. Parental smoking during pregnancy and its association with low birth weight, small for gestational age, and preterm birth offspring: A birth cohort study. *Pediatr Neonatol.* 2014;55(1):20-27. doi:10.1016/j.pedneo.2013.05.005.
- 68. Jordanov JS. Cotinine concentrations in amniotic fluid and urine of smoking, passive smoking and non-smoking pregnant women at term and in the urine of their neonates on 1st day of life. *Eur J Pediatr*. 1990;149(10):734-737. doi:10.1007/BF01959534.
- 69. Collins MH, Moessinger AC, Kleinerman J, Bassi J, Rosso P, Collins AM, James SL, Blanc WA. Fetal lung hypoplasia associated with maternal smoking: A morphometric analysis. *Pediatr Res.* 1985;19(4):408-412. doi:10.1203/00006450-198519040-00018.
- 70. Alati R, Al Mamun A, O'Callaghan M, Najman JM, Williams GM. In utero and postnatal maternal smoking and asthma in adolescence. *Epidemiology*. 2006;17(2):138-144. doi:10.1097/01.ede.0000198148.02347.33.
- Burke H, Leonardi-Bee J, Hashim A, Pine-Abata H, Chen Y, Cook DG, Britton JR, McKeever TM. Prenatal and Passive Smoke Exposure and Incidence of Asthma and Wheeze: Systematic Review and Meta-analysis. *Pediatrics*. 2012;129(4):735-744. doi:10.1542/peds.2011-2196.
- 72. Cheraghi M, Salvi S. Environmental tobacco smoke (ETS) and respiratory health in children. *Eur J Pediatr*. 2009;168(8):897-905. doi:10.1007/s00431-009-0967-3.
- 73. Gilliand FD, Li Y, Peters JM. Effects of Maternal Smoking during Pregnancy and Environmental Tobacco Smoke on Asthma and Wheezing in Children. *Am J Respir Crit Care Med.* 2001;163(2):429-436. doi:10.1164/ajrccm.163.2.2006009.
- 74. Lannerö E, Wickman M, Pershagen G, Nordvall SL. Maternal smoking during pregnancy increases the risk of recurrent wheezing during the first years of life (BAMSE). *Respir Res.* 2006;7(1):3. doi:10.1186/1465-9921-7-3.
- 75. Curle DC, Adamson IYR. Retarded development of neonatal rat lung by maternal malnutrition. *J Histochem Cytochem*. 1978;26(5):401-408. doi:10.1177/26.5.659840.
- 76. Karadag A, Sakurai R, Wang Y, Guo P, Desai M, Ross MG, Torday JS, Rehan VK. Effect of maternal food restriction on fetal rat lung lipid differentiation program. *Pediatr Pulmonol*. 2009;44(7):635-644. doi:10.1002/ppul.21030.
- 77. Rehan VK, Sakurai R, Li Y, Karadag A, Corral J, Bellusci S, Xue YY, Belperio J, Torday JS. Effects of maternal food restriction on offspring lung extracellular matrix deposition and long term pulmonary function in an experimental rat model. *Pediatr Pulmonol*. 2012;47(2):162-171. doi:10.1002/ppul.21532.
- 78. Dezateux C. Low birth weight for gestation and airway function in infancy: exploring the fetal origins hypothesis. *Thorax*. 2004;59(1):60-66.
- 79. Barker DJ, Godfrey KM, Fall C, Osmond C, Winter PD, Shaheen SO. Relation of birth weight and childhood respiratory infection to adult lung function and death from chronic obstructive airways disease. BMJ. 1991;303(6804):671-675. doi:10.1136/bmj.303.6804.671.
- 80. Turner SW, Campbell D, Smith N, Craig LCA, McNeill G, Forbes SH, Harbour PJ, Seaton A, Helms PJ, Devereux GS.

Associations between fetal size, maternal α -tocopherol and childhood asthma. *Thorax*. 2010;65(5):391-397. doi:10.1136/thx.2008.111385.

- 81. Shaheen SO, Sterne JAC, Montgomery SM, Azima H. Birth weight, body mass index and asthma in young adults. *Thorax*. 1999;54(5):396-402. doi:10.1136/thx.54.5.396.
- 82. Bjørke-Monsen AL, Vollsæter M, Ueland PM, Markestad T, Øymar K, Halvorsen T. Increased bronchial hyperresponsiveness and higher asymmetric dimethylarginine levels after fetal growth restriction. *Am J Respir Cell Mol Biol*. 2017;56(1):83-89. doi:10.1165/rcmb.2016-02100C.
- Moldenhauer JS, Lai Y, Ph D, Schatz M, Wise R, Landon MB, Newman RB, Rouse DJ, Hyagriv N, Leveno KJ, Miodovnik M, Marshall D. HHS Public Access. J Asthma. 2015;47(2):145-149. doi:10.3109/02770900903483774.Influence.
- 84. Al-Ghazali W, Chita SK, Chapman MG, Allan LD. Evidence of redistribution of cardiac output in asymmetrical growth retardation. *BJOG An Int J Obstet Gynaecol.* 1989;96(6):697-704. doi:10.1111/j.1471-0528.1989.tb03285.x.
- Ortqvist AK, Lundholm C, Carlstrom E, Lichtenstein P, Cnattingius S, Almqvist C. Familial Factors Do not Confound the Association Between Birth Weight and Childhood Asthma. *Pediatrics*. 2009;124(4):e737-e743. doi:10.1542/peds.2009-0305.
- 86. Rozance PJ, Seedorf GJ, Brown A, Roe G, O'Meara MC, Gien J, Tang J-R, Abman SH. Intrauterine growth restriction decreases pulmonary alveolar and vessel growth and causes pulmonary artery endothelial cell dysfunction in vitro in fetal sheep. *Am J Physiol Lung Cell Mol Physiol*. 2011;301(6):L860-L871. doi:10.1152/ajplung.00197.2011.
- 87. Nogueira PJ, Ferreira HHA, Antunes E, Teixeira NA. Chronic mild prenatal stress exacerbates the allergen-induced airway inflammation in rats. *Mediators Inflamm*. 1999;8(2):119-122. doi:10.1080/09629359990621.
- Adams Waldorf KM, McAdams RM. Influence of infection during pregnancy on fetal development. *Reproduction*. 2013;146(5):R151-62. doi:10.1530/REP-13-0232.
- 89. Angel E, Wellnitz C V, Goodsitt MM, Yaghmai N, DeMarco JJ, Cagnon CH, Sayre JW, Cody DD, Stevens DM, Primak AN, McCollough CH, McNitt-Gray MF. Radiation Dose to the Fetus for Pregnant Patients Undergoing Multidetector CT Imaging: Monte Carlo Simulations Estimating Fetal Dose for a Range of Gestational Age and Patient Size. *Radiology*. 2008;249(1):220-227. doi:10.1148/radiol.2491071665.
- 90. Hart D, Shrimpton PC, Brown G. Fourth review of the UK national patient dose database. *Br J Radiol.* 2012;85(1018):e957-8. doi:10.1259/bjr/28778022.
- 91. Steenvoorde P, Pauwels EKJ, Harding LK, Bourguignon M, Marière B, Broerse JJ. Diagnostic nuclear medicine and risk for the fetus. *Eur J Nucl Med*. 1998;25(2):193-199. doi:10.1007/s002590050215.
- 92. Rugh R, Wohlfromm M. Prenatal X-Irradiation and Postnatal Mortality. *Radiat Res.* 1965;26(4):493-506. doi:10.2307/3571860.
- 93. The International Commission on Radiological Protection. *Biological Effects after Prenatal Irradiation (Embryo and Fetus). ICRP Publication 90.* Vol 90.; 2003.
- 94. Otake M, Schull WJ. Radiation-related small head sizes among prenatally exposed a-bomb survivors. *Int J Radiat Biol*. 1993;63(2):255-270. doi:10.1080/09553009314550341.
- 95. Sharp C, Shrimpton JA, Bury RF. *Diagnostic Medical Exposures: Advice on Exposure to Ionising Radiation during Pregnancy.*; 1998.
- 96. Winer-Muram HT, Boone JM, Brown HL, Jennings SG, Mabie WC, Lombardo GT, Mabie WC, Jennings SG, Boone JM. Pulmonary Embolism in Pregnant Patients: Fetal Radiation Dose with Helical CT. *Radiology*. 2002;224(2):487-492. doi:10.1148/radiol.2242011581.
- 97. Goldstein L, Murphy DP. Microcephalic idiocy following radium therapy for uterine cancer during pregnancy. *Am J Obstet Gynecol*. 1929;18(2):189-195. doi:10.1016/S0002-9378(29)91057-7.
- 98. Russell L, Russel W. The effect of radiation on the preimplantation stages of the mouse embryo. *Anat Rec.* 1950;108.
- 99. Moore NW, Adams CE, Rowson LE. Developmental potential of single blastomeres of the rabbit egg. J Reprod Fertil. 1968;17(3):527-531.
- 100. Uma Devi P, Baskar R. Influence of gestational age at exposure on the prenatal effects of gamma-radiation. *Int J Radiat Biol.* 1996;70(1):45-52.
- 101. Kim SH, Kim SR, Lee YS, Kim TH, Jo SK, Lee CS. Influence of gestational age at exposure on the prenatal effects of gamma-radiation. *J Vet Sci (Suwon-si, Korea)*. 2001;2(1):37-42. doi:10.1080/095530096145319.
- 102. Jacquet P, de Saint-Georges L, Vankerkom J, Baugnet-Mahieu L. Embryonic death, dwarfism and fetal malformations after irradiation of embryos at the zygote stage: studies on two mouse strains. *Mutat Res Fundam Mol Mech Mutagen*. 1995;332(1-2):73-87. doi:10.1016/0027-5107(95)00156-4.
- 103. Müller W -U, Streffer C. Lethal and teratogenic effects after exposure to X-rays at various times of early murine gestation. *Teratology*. 1990;42(6):643-650. doi:10.1002/tera.1420420609.
- 104. Streffer C, Müller WU. Malformations after radiation exposure of preimplantation stages. *Int J Dev Biol*. 1996;40(1):355-360. doi:10.1387/IJDB.8735948.
- 105. Prakash Hande M, Uma Devi P, Jagetia GC. Effect of "in utero" exposure to low doses of low energy X-rays on the postnatal development of mouse. *J Radiat Res.* 1990;31(4):354-360. doi:10.1269/jrr.31.354.
- 106. Nash DJ. Effects of Prenatal X-Irradiation on Development and Postnatal Viability of Inbred and Hybrid Mice. *Biol Bull*. 1971;140(2):230-241. doi:10.2307/1540071.
- 107. Nash DJ, Gowen JW. Effects of X-irradiation upon postnatal growth in the mouse. *Biol Bull*. 1962;122(1):115-136. doi:10.2307/1539326.
- 108. Sikov MR, Resta CF, Lofstrom JE. The effects of prenatal x-irradiation of the rat on postnatal growth and mortality. *Radiat Res.* 1969;40(1):133-148. doi:10.2307/3572990.
- 109. Li HP, Miki T, Gu H, Satriotomo I, Mastumoto Y, Kuma H, Gonzalez D, Bedi KS, Suwaki H, Takeuchi Y. The effect of the timing of prenatal X-irradiation on Purkinje cell numbers in rat cerebellum. *Dev Brain Res.* 2002;139(2):159-166. doi:10.1016/S0165-3806(02)00542-4.
- 110. Norton S. Behavioral changes in preweaning and adult rats exposed prenatally to low lonizing radiation. *Toxicol Appl Pharmacol.* 1986;83(2):240-249. doi:10.1016/0041-008X(86)90301-7.
- 111. Kimler BF, Norton S. Behavioral changes and structural defects in rats irradiated in utero. *Int J Radiat Oncol Biol Phys.* 1988;15(5):1171-1177. doi:10.1016/0360-3016(88)90200-3.
- 112. Jensh RP, Brent RL. The Effects of Prenatal X Irradiation on the Appearance of Reflexes and Physiologic Markers in the Neonatal Rat. *Radiat Res.* 1988;116(3):416. doi:10.2307/3577385.
- 113. Zhong DZ, Pei C, Xiu-Qin LI. Neurobehavioral study of prenatal exposure to hyperthermia combined with irradiation in mice. *Neurotoxicol Teratol*. 1996;18(6):703-709. doi:10.1016/S0892-0362(96)00127-4.
- 114. Kinlen LJ, Acheson ED. Diagnostic irradiation, congenital malformations and spontaneous abortion. *Br J Radiol.* 1968;41(489):648-654. doi:10.1259/0007-1285-41-489-648.
- 115. Ornoy A, Patlas N, Schwartz L. The effects of in utero diagnostic X-irradiation on the development of preschoolage children. *Isr J Med Sci*. 1996;32(2):112-115. doi:10.1016/S0140-6736(05)66736-0.
- 116. Osei EK, Faulkner K. Fetal doses from radiological examinations. *Br J Radiol*. 1999;72(AUG.):773-780. doi:10.1259/bjr.72.860.10624343.
- 117. Lim H, Beasley CW, Whitehead LW, Emery RJ, Agopian AJ, Langlois PH, Waller DK. Maternal exposure to radiographic exams and major structural birth defects. *Birth Defects Res Part A Clin Mol Teratol*. 2016;106(7):563-572. doi:10.1002/bdra.23496.
- 118. Baskar R, Uma Devi P. Influence of gestational age to low-level gamma irradiation on postnatal behavior in mice.

Neurotoxicol Teratol. 2000;22(4):593-602. doi:10.1016/S0892-0362(00)00076-3.

- 119. Sienkiewicz ZJ, Haylock RGE, Saunders RD. Differential learning impairments produced by prenatal exposure to ionizing radiation in mice. *Int J Radiat Biol*. 1999;75(1):121-127. doi:10.1080/095530099140889.
- 120. Minamisawa T, Hirokaga K. Long-Term Changes in Open Field Activity of Male Mice Irradiated with Low Levels of Gamma Rays at Late Stage of Development. *J Radiat Res.* 1996;37(2):117-124. doi:10.1269/jrr.37.117.
- 121. Minamisawa T, Hirokaga K. Long-Term Effects of Prenatal Exposure to Low Levels of Gamma Rays on Open-Field Activity in Male Mice. *Radiat Res.* 1995;144(2):237. doi:10.2307/3579264.
- 122. Murphree RL, Pace HB. The Effects of Prenatal Radiation on Postnatal Development in Rats. *Radiat Res.* 1960;12(5):495. doi:10.2307/3570971.
- 123. Hossain M, Uma Devi P, Bisht KS. Effect of prenatal gamma irradiation during the late fetal period on the postnatal development of the mouse. *Teratology*. 1999;59(3):133-138. doi:10.1002/(SICI)1096-9926(199903)59:3<133::AID-TERA4>3.0.CO;2-T.
- 124. Jensh RP, Eisenman LM, Brent RL. Postnatal neurophysiologic effects of prenatal x-irradiation. *Int J Radiat Biol.* 1995;67(2):217-227. doi:10.1080/09553009514550271.
- 125. Mayer MD, Harris W, Wimpfheimer S. Therapeutic Abortion by Means of X-Ray. *Am J Obstet Gynecol*. 1936;32(6):945-957. doi:10.1016/S0002-9378(16)41156-7.
- 126. Wereboff J, Goodman I, Havlena J, Sikov MR. Effects of prenatal x-irradiation on motor performance in the rat. *Am J Physiol*. 1961;201(4):703-706. doi:10.1016/j.medcli.2014.04.029.
- 127. Hossain M, Uma Devi P. Effect of irradiation at the early foetal stage on adult brain function of mouse: Learning and memory. *Int J Radiat Biol.* 2001;77(5):581-585. doi:10.1080/09553000110037394.
- 128. Sienkiewicz ZJ, Haylock RGE, Saunders RD. Prenatal irradiation and spatial memory in mice: Investigation of doseresponse relationship. *Int J Radiat Biol*. 1994;65(5):611-618. doi:10.1080/09553009414550701.
- 129. Hossain M, Uma Devi P. Effect of irradiation at the early fetal stage on adult brain function in the mouse: locomotor activity. *Int J Radiat Biol.* 2000;76(10):1397-1402. doi:10.1080/09553000050151673.
- 130. Otake M, Schull WJ, Lee S. Threshold for radiation-related severe mental retardation in prenatally exposed Abomb survivors: A re-analysis. *Int J Radiat Biol*. 1996;70(6):755-763. doi:10.1080/095530096144644.
- 131. American College of Radiologists. *Practice Parameter for Imaging Pregnant or Potentially Pregnant Adolescents and Women With Ionizing Radiation*. Vol 1076.; 2018.
- 132. Moroni M, Lombardini E, Salber R, Kazemzedeh M, Nagy V, Olsen C, Whitnall MH. Hematological changes as prognostic indicators of survival: Similarities between gottingen minipigs, humans, and other large animal models. *PLoS One*. 2011;6(9):e25210. doi:10.1371/journal.pone.0025210.
- 133. Mandel L, Travnicek J, Talafantova M, Zahradnickova M. The LD50/30 and the survival time in whole-body gamma-irradiated conventional and germfree Minnesota miniature piglets. *Z Versuchstierkd*. 1980;22(2):96-100.
- Carsten AL. Acute Lethality The Hemopoietic Syndrome in Different Species. In: *Response of Different Species to Total Body Irradiation*. Vol 10. Dordrecht: Springer Netherlands; 1984:59-86. doi:10.1007/978-94-009-6048-0_5.
- 135. Norris WP, Fritz TE, Rehfeld CE, Poole CM. The response of the beagle dog to cobalt-60 gamma radiation: determination of the LD50(30) and description of associated changes. *Radiat Res.* 1968;35(3):681-708. doi:10.2307/3572525.
- 136. Singh VK, Newman VL, Berg AN, MacVittie TJ. Animal models for acute radiation syndrome drug discovery. *Expert Opin Drug Discov*. 2015;10(5):497-517. doi:10.1517/17460441.2015.1023290.
- 137. Mole RH. The LD50 for uniform low LET irradiation of man. *Br J Radiol*. 1984;57(677):355-369. doi:10.1259/0007-1285-57-677-355.

- Plett PA, Sampson CH, Chua HL, Jackson W, Vemula S, Sellamuthu R, Fisher A, Feng H, Wu T, MacVittie TJ, Orschell CM. The H-ARS dose response relationship (DRR): Validation and variables. *Health Phys.* 2015;109(5):391-398. doi:10.1097/HP.00000000000354.
- 139. Health Protection Agency. Human Radiosensitivity Report of the Independent Advisory Group on Ionising Radiation.; 2013.
- 140. Grahn D, Hamilton KF. Genetic Variation in the Acute Lethal Response of Four Inbred Mouse Strains to Whole Body X-Irradiation. *Genetics*. 1957;42(3):189-198.
- 141. Nunamaker EA, Artwohl JE, Anderson RJ, Fortman JD. Endpoint refinement for total body irradiation of C57BL/6 mice. *Comp Med*. 2013;63(1):22-28.
- 142. Wang Y, Xu C, Du LQ, Cao J, Liu JX, Su X, Zhao H, Fan FY, Wang B, Katsube T, Fan SJ, Liu Q. Evaluation of the comet assay for assessing the dose-response relationship of DNA damage induced by ionizing radiation. *Int J Mol Sci.* 2013;14(11):22449-22461. doi:10.3390/ijms141122449.
- 143. Vignard J, Mirey G, Salles B. Ionizing-radiation induced DNA double-strand breaks: A direct and indirect lighting up. *Radiother Oncol.* 2013;108(3):362-369. doi:10.1016/j.radonc.2013.06.013.
- 144. Robert Weinberg. *The Biology of Cancer*. 2nd ed.; 2013.
- 145. Kamiya K, Ozasa K, Akiba S, Niwa O, Kodama K, Takamura N, Zaharieva EK, Kimura Y, Wakeford R. Long-term effects of radiation exposure on health. *Lancet*. 2015;386(9992):469-478. doi:10.1016/S0140-6736(15)61167-9.
- 146. Cullings HM, Fujita S, Funamoto S, Grant EJ, Kerr GD, Preston DL. Dose Estimation for Atomic Bomb Survivor Studies: Its Evolution and Present Status. *Radiat Res*. 2006;166(1):219-254. doi:10.1667/RR3546.1.
- 147. Grant EJ, Furukawa K, Sakata R, Sugiyama H, Sadakane A, Takahashi I, Utada M, Shimizu Y, Ozasa K. Risk of death among children of atomic bomb survivors after 62 years of follow-up: A cohort study. *Lancet Oncol.* 2015;16(13):1316-1323. doi:10.1016/S1470-2045(15)00209-0.
- 148. Neel J V., Schull WJ, McDonald DJ, Morton NE, Kodani M, Takeshima K, Anderson RC, Wood J, Brewer R, Wright S, Yamazaki J, Suzuki M, Kitamura S. The effect of exposure to the atomic bombs on pregnancy termination in Hiroshima and Nagasaki: Preliminary report. *Science (80-)*. 1953;118(3071):537-541. doi:10.1126/science.118.3071.537.
- 149. Fujiwara S, Suyama A, Cologne JB, Akahoshi M, Yamada M, Suzuki G, Koyama K, Takahashi N, Kasagi F, Grant EJ, Lagarde F, Hsu WL, Furukawa K, Ohishi W, Tatsukawa Y, Neriishi K, Takahashi I, Ashizawa K, Hida A, Imaizumi M, Nagano J, Cullings HM, Katayama H, Ross NP, Kodama K, Shore RE. Prevalence of Adult-Onset Multifactorial Disease among Offspring of Atomic Bomb Survivors. *Radiat Res*. 2008;170(4):451-457. doi:10.1667/RR1392.1.
- 150. Izumi S, Koyama K, Soda M, Suyama A. Cancer incidence in children and young adults did not increase relative to parental exposure to atomic bombs. *Br J Cancer*. 2003;89(9):1709-1713. doi:10.1038/sj.bjc.6601322.
- 151. Aisner J, Wiernik PH, Pearl P. Pregnancy outcome in patients treated for hodgkin's disease. *J Clin Oncol.* 1993;11(3):507-512. doi:10.1200/JCO.1993.11.3.507.
- 152. Hawkins MM. Is there evidence of a therapy-related increase in germ cell mutation among childhood cancer survivors? *J Natl Cancer Inst.* 1991;83(22):1643-1650. doi:10.1093/jnci/83.22.1643.
- 153. Kodaira M, Ryo H, Kamada N, Furukawa K, Takahashi N, Nakajima H, Nomura T, Nakamura N. No Evidence of Increased Mutation Rates at Microsatellite Loci in Offspring of A-Bomb Survivors. *Radiat Res.* 2010;173(2):205-213. doi:10.1667/RR1991.1.
- 154. Neel JV, Schull WJ, Awa AA, Satoh C, Kato H, Otake M, Yoshimoto Y. The children of parents exposed to atomic bombs: estimates of the genetic doubling dose of radiation for humans. *AmJHumGenet*. 1990;46:1053-1072. doi:10.1269/jrr.32.SUPPLEMENT 347.
- 155. Delongchamp RR, Mabuchi K, Yoshimoto Y, Preston DL. Cancer mortality among atomic bomb survivors exposed in utero or as young children, October 1950-May 1992. *Radiat Res*. 1997;147(3):385-395.

- 156. Preston DL, Cullings H, Suyama A, Funamoto S, Nishi N, Soda M, Mabuchi K, Kodama K, Kasagi F, Shore RE. Solid cancer incidence in atomic bomb survivors exposed in utero or as young children. J Natl Cancer Inst. 2008;100(6):428-436. doi:10.1093/jnci/djn045.
- 157. Stewart A, Webb J, Giles D, Hewitt D. Malignant disease in childhood and diagnostic irradiation in utero. *Lancet*. 1956;268(6940):447. doi:10.1093/jac/23.2.247.
- 158. Doll R, Wakeford R. Risk of childhood cancer from fetal irradiation. *Br J Radiol*. 1997;70(FEB.):130-139. doi:10.1259/bjr.70.830.9135438.
- 159. Wakeford R, Little MP. Risk coefficients for childhood cancer after intrauterine irradiation: A review. *Int J Radiat Biol*. 2003;79(5):293-309. doi:10.1080/0955300031000114729.
- 160. Rajaraman P, Simpson J, Neta G, De Gonzalez AB, Ansell P, Linet MS, Ron E, Roman E. Early life exposure to diagnostic radiation and ultrasound scans and risk of childhood cancer: Case-control study. *BMJ*. 2011;342(7794):424. doi:10.1136/bmj.d472.
- 161.Boice JD, Miller RW. Childhood and adult cancer after intrauterine exposure to ionizing radiation. In: Teratology.
Vol 59. ; 1999:227-233. doi:10.1002/(SICI)1096-9926(199904)59:4<227::AID-TERA7>3.0.CO;2-E.
- 162. Song JW, Chung KC. Observational studies: Cohort and case-control studies. *Plast Reconstr Surg*. 2010;126(6):2234-2242. doi:10.1097/PRS.0b013e3181f44abc.
- 163. Totter JR, MacPherson HG. Do childhood cancers result from prenatal x-rays? *Health Phys.* 1981;40(4):511-524.
- 164. Nakashima E. Radiation Dose Response Estimation with Emphasis on Low Dose Range Using Restricted Cubic Splines. *Health Phys.* 2015;109(1):15-24. doi:10.1097/HP.00000000000293.
- 165. LifeMap Discovery. Development of the Respiratory System infographic. https://discovery.lifemapsc.com/library/images/development-of-the-respiratory-system. Accessed June 26, 2019.

CHAPTER 2

CUMULATIVE RADIATION IN CRITICALLY ILL PATIENTS: A RETROSPECTIVE AUDIT OF IONIZING RADIATION EXPOSURE IN AN INTENSIVE CARE UNIT.

Reproduced with kind permission from Critical Care and Resuscitation. McEvoy J.H., Bihari S., Hooker A.M., Dixon D-L., 2019, Crit Care Resus, 21(3): Appendix A

SUMMARY

Ionizing radiation is a valuable tool in modern medicine including for patients in an intensive care unit (ICU). However, clinicians are faced with a trade-off between benefit of information received from procedure versus risks associated with radiation. As a first step to understanding the risk and benefits of radiation exposure to ICU patients, we aimed to assess the cumulative levels of ionizing radiation patients receive during their ICU stay. This audit included 526 patients admitted to the ICU, Flinders Medical Centre, South Australia, for longer than 120hrs (long-stay) over a 12-month period from April 2015 to 2016. The 526 patients audited underwent 4331 procedures totalling 5688.45 mSv of ionizing radiation. The most frequent procedure was a chest X-ray (82%) which contributed 1.2% to cumulative effective dose (CED). Although only 3.6% of total procedures, abdominal/pelvic computed tomography (CT) contributed the most to CED (68%). Over

50% of patients received less than 1 mSv CED during their stay in the ICU. However, 6% received >50 mSv and 2% >100 mSv CED. Trauma patients received significantly higher CED compared to other admission diagnoses and CED increased with length of stay. Most ICU patients received low CED during their stay, with the majority less than the recommended public limit (1 mSv). These results could educate clinicians regarding radiation exposures in ICU settings highlighting the relatively low exposures and thus low risk to the patients.

2.1 INTRODUCTION

lonizing radiation is continuous and ubiquitous, coming from both natural and artificial sources and can be expressed in millisieverts (mSv), which is the measurement of the biological effect of the absorbed radiation dose. At natural background levels, Australians are exposed to 1.5 millisieverts of radiation annually, while residents in the United States of America (USA) receive approximately 3.1 mSv^{1,2}. This level increases with artificial exposures, another 1.7 mSv for Australia and 3.1 mSv for USA, totalling an average yearly exposure of 3.2 mSv and 6.2 mSv respectively^{2,3}. Most artificial exposures derive from medical sources. Ionizing radiation is a crucial tool in medicine, aiding both diagnosis and therapy. Among hospitalised patients, diagnostic and interventional radiology is potentially most frequently used for patients admitted to Intensive Care Units (ICU).

Diagnostic radiation is used to aid ICU clinicians in both disease diagnosis and management. Previous studies attempting to calculate patient's cumulative effective dose (CED) over their stay in the ICU have been limited to exclusive populations within the ICU and thus overestimate cumulative exposure^{4–12}. The real burden of radiological exposure in ICU patients remains unknown.

Conventional X-rays, computed tomography (CT), fluoroscopy and nuclear medicine are all forms of ionizing radiation that clinicians use to enable quicker and more accurate patient diagnosis. Mettler *et al.* collated average effective doses of each radiological exam worldwide between 1980 and 2008¹³. An anterior-posterior chest x-ray,

the most common radiological procedure prescribed in ICUs, was reported as 0.02 mSv, approximating to 5 days background radiation^{1,13}. Generally, higher exposures of radiation are received from CT where median chest and abdominal exposures are reported as 7 mSv and 8 mSv respectively^{9,13}. However, when surveyed, it was highlighted that some health care workers have a limited understanding of the doses associated with procedures utilising ionizing radiation and therefore have a misunderstanding of the associated risks^{14–19}.

Under the current international regulatory framework, it is considered that there is no 'safe' level of radiation and that all radiation can be considered harmful. Currently, the member of public limit has been set at 1 mSv/yr, whilst occupationally exposed workers can receive up to 20 mSv/yr above background²⁰. For medical exposures, all treatments must be clinically justified on an individual's needs for successful diagnosis and treatment. The International Commission on Radiation Protection (ICRP) states a cancer risk of 5.5% per Sievert (Sv)²⁰. Thus, when extrapolated down to diagnostic levels, a whole-body 10 mSv CT scan would equate to an approximate 0.06% increase (1/1800) in cancer risk. When added to the average risk of cancer by the age of 65, the cancer risk from a CT would increase from 40% to 40.06%²¹. However, it is important to recognise that the ICRP, as well as other governing bodies, state that epidemiological methods do not have the power to isolate cancer risks for exposures below 100 mSv^{20,22,23}.

55

As a first step, it is important that we accurately determine the level of radiation exposure of patients in the general ICU cohort, including an assessment of the exposure due to each procedure, to inform clinicians and thereby aid their assessment of the risk-tobenefit ratio when prescribing diagnostic radiation.

The aim of this study was to conduct a retrospective audit of daily and cumulative radiation exposure of all patients admitted for >120 hours to a 32-bed mixed surgical and medical, metropolitan, tertiary level ICU, over a 12-month period.

2.2 METHODS

2.2.1 Study cohort

The study cohort included patients admitted to the ICU, Flinders Medical Centre, Adelaide, Australia, between 1st April 2015 and 1st April 2016 with ICU length of stay >120 hours. Patients with shorter stays were excluded to remove routine post-surgery stays. This audit was reviewed and approved by the South Adelaide Clinical Human Research Ethics Committee (OFR # 131.16) in line with the requirements of the *National Statement on Ethical Conduct in Human Research* and the requirement of informed consent was waived.

2.2.2 Data collection

Hospital records from the Open Architecture Clinical Information System, the Australian Outcomes Research Tool for Intensive Care (AORTIC) database and the radiological database picture archiving and communications system were used to collect demographic information, clinical data and radiology reports for each patient. All ionizing forms of radiation were recorded including conventional X-rays, CT, fluoroscopy and nuclear medicine.

2.2.3 Calculation of effective dose

Effective doses for each procedure were calculated using values from patient radiology reports and reported conversion factors by Deak *et al* and Hart and Wall^{24,25}. In short, values for Dose Area Product (DAP) or Dose Length Product (DLP) produced from the radiology reports were collected and converted to effect dose (mSv) via multiplication by tissue-specific conversion factors (Table 2-1). For conventional X-rays, accepted diagnostic reference levels were used for each individual exam due to lack of information in reports, except in cases where sufficient information enabled calculation of effective doses.

As this study focuses on initial management and intensive care treatment, radiological examinations were only included if they were performed during the time between admission and discharge from the ICU. No radiological procedures were recorded pre or post ICU stay.

57

| | Conversion factor | |
|----------------------|-------------------|--|
| CONVENTIONAL X-RAY | 0.000028 | |
| Head/neck | 0.00012 | |
| Chest | 0.00027 | |
| Abdominal and pelvic | 0.00001 | |
| Extremities | 0.0019 | |
| COMPUTED TOMOGRAPHY | 0.0051 | |
| Head | 0.0145 | |
| Neck | 0.0153 | |
| Chest | 0.0015 | |
| Abdominal and pelvic | 0.0120 | |
| Extremities | 0.000028 | |
| Spine | 0.00003 | |
| FLUOROSCOPY | 0.00012 | |
| Cerebral | 0.00012 | |
| Oesophageal | 0.00026 | |
| Pulmonary | 0.00018 | |
| Cardiovascular | 0.00001 | |
| Gastrointestinal | 0.011 | |
| Urinary | 0.007 | |
| Peripheral | 0.0057 | |
| NUCLEAR MEDICINE | 0.0017 | |
| 99mTc MAA | 0.015 | |
| 99mTc MAG3 | 0.000028 | |
| 99mTc MDP | 0.00012 | |
| 99mTc DISIDA | 0.00027 | |
| 99mTc Tgas | 0.00001 | |

Table 2-1: Conversion factors for each modality divided by body region scanned or radiopharmaceutical used adopted from Deak *et al* and Hart and Wall^{24,25}.

CT= computed tomography; abdo/pelv= abdominal and pelvic; 99mTc = Technetium-99m; MMA = macro aggregated albumin; MAG3 mercaptoacetyltriglycine; MDP = methyl diphosphonate; DISIDA = diisopropyliminodiacetic acid; Tgas = Technegas.

2.2.4 Statistics

All outcomes measured in this audit were not normally distributed, therefore data was expressed as median (interquartile range, IQR) and tested using non-parametric tests. The dichotomous variable sex was assessed by Mann-Whitney U test and categorical variables, including patient ethnicity, length of stay, and admission diagnosis, were assessed by Kruskal-Wallis analysis with *post hoc* Mann-Whitney U test. Scale variables, including age and APACHE III score, were assessed using Spearman's rank-order correlations. Statistical significance was dictated by p < 0.05

2.3 RESULTS

This cohort included 526 patients that underwent 4331 procedures totalling 5688.45 mSv of ionizing radiation. The median (IQR) age was 65 (51 – 76) years old with a median (IQR) length of ICU stay of 7 (6 – 13) days and overall mortality of 10.3% (Table 2-2). Patients were admitted with either non-operative (medical) or post-operative (surgical) diagnoses with respiratory or cardiovascular complications the main reason for admission.

Table 2-2: Summary of patient demographics and clinical characteristics.

| Demographic and clinical characteristics | Values (%) | |
|--|-------------|--|
| TOTAL NUMBER OF PATIENTS | 526 | |
| AGE (years), median (IQR) | 65 (51–76) | |
| SEX | | |
| Male | 306 (58.2%) | |
| ETHNICITY | | |
| Aboriginal | 31 (5.9%) | |
| Caucasian | 477 (90.7%) | |
| Asian | 5 (1%) | |
| Other | 3 (0.6%) | |
| Unknown | 10 (1.9%) | |
| APACHE III SCORE, median (IQR) | 72 (57–86) | |
| ICU STAY (days), median (IQR) | 7 (6–13) | |
| ICU mortality | 54 (10.3%) | |
| ADMISSION DIAGNOSIS | | |
| Medical | | |
| Cardiovascular | 91 (17.3%) | |
| Respiratory | 94 (17.9%) | |
| Gastrointestinal | 45 (8.6%) | |
| Neurological | 46 (8.7%) | |
| Sepsis | 52 (9.9%) | |
| Trauma | 24 (4.6%) | |
| Metabolic | 25 (4.8%) | |
| Haematological | 6 (1.1%) | |
| Renal/genitourinary | 14 (2.7%) | |
| Other disorders | 1 (0.2%) | |
| Musculoskeletal | 5 (1.0%) | |
| Surgical | | |
| Cardiovascular | 50 (9.5%) | |
| Respiratory | 7 (1.3%) | |
| Gastrointestinal | 34 (6.5%) | |
| Neurological | 13 (2.5%) | |
| Sepsis | 8 (1.5%) | |
| Renal/genitourinary | 2 (0.4%) | |
| Gynaecological | 1 (0.2%) | |
| Musculoskeletal | 8 (1.5%) | |
| Haematological | 0 (0.0%) | |
| Metabolic | 0 (0.0%) | |

APACHE = Acute Physiology, Age, Chronic Health Evaluation;

ICU = intensive care unit; IQR = interquartile range

All patients who received some diagnostic radiography (98.5%) received at least one conventional X-ray. Of those, 248 patients (48%) received only conventional X-rays. A CT scan was received by 255 patients (49.3%), fluoroscopic exam by 74 (14.3%) and only 6 patients (1.2%) underwent nuclear medicine. The median (IQR) CED of the cohort was 0.91 mSv (0.08 – 11.37mSv). There was no difference in median CED between sexes (p = 0.131), or ethnicities (p = 0.964).

2.3.1 Distribution of procedure dose

As dictated in Table 2-3, the most common modality performed in this ICU was a conventional X-rays, accounting for 85.2% of the 4331 procedures of which the chest X-ray was the most frequent procedure accounting for 82% of all procedures performed. However, because conventional X-rays produce a relatively low dose of radiation, they only accounted for 2.3% contribution to CED. Conversely, CTs contributed the most to CED (93.1%) with the highest contribution coming from abdominal/pelvic, representing 68.8% of the overall 5688.45 mSv. Both the fluoroscopic exams and nuclear medicine accounted for minimal percentages of both number of procedures and total CED.

The highest median effective dose was produced from abdominal/pelvic CTs (21.88 mSv), which also gave the highest individual exposure of any procedure (78.82 mSv) (Table 2-4). There was large variation between doses for the same procedure sometimes up to 1000-fold, for example with gastrointestinal fluoroscopy.

Table 2-3: Distribution of procedures which use ionizing radiation and the frequency at which they contribute number of procedures and total dose.

| | Contribution to frequency | Contribution to total dose |
|---------------------|---------------------------|----------------------------|
| | (n = 4311 procedures) | (n = 5688.45 mSv) |
| Conventional X-ray | 85.2% | 2.3% |
| Computed tomography | 12.5% | 93.1% |
| Fluoroscopy | 2.2% | 4.3% |
| Nuclear medicine | 0.1% | 0.3% |

mSv = millisieverts

Table 2-4: Exposures from each modality divided by body region scanned.

| | Median effective dose (mSv) | Range of effective dose |
|----------------------|-----------------------------|-------------------------|
| | Median (IQR) | (mSv) |
| CONVENTIONAL X-RAY | | |
| Head/neck | 0.01 (0.01–0.03) | 0.01-0.03 |
| Chest | 0.02 (0.02–0.02) | 0.02-0.06 |
| Abdominal and pelvic | 0.75 (0.75–1.50) | 0.14-3.75 |
| Extremities | 0.00 (0.00–0.00) | 0.00-0.02 |
| COMPUTED TOMOGRAPHY | | |
| Head | 1.64 (1.59–1.79) | 1.06-5.20 |
| Neck | 4.14 (3.29–5.76) | 1.61-12.83 |
| Chest | 7.53 (5.07–11.64) | 0.98-36.30 |
| Abdominal and pelvic | 21.88 (14.05–31.72) | 1.55-78.82 |
| Extremities | 0.97 (0.31–1.79) | 0.25-2.14 |
| Spine | 9.66 (5.76–21.70) | 5.76-21.70 |
| FLUOROSCOPY | | |
| Cerebral | 2.67 (2.05–3.25) | 0.11-13.95 |
| Oesophageal | 0.02 (0.01–0.26) | 0.01-1.14 |
| Pulmonary | 5.30 (0.70–11.16) | 0.23-12.89 |
| Cardiovascular | 0.02 (0.00–0.23) | 0.00-28.70 |
| Gastrointestinal | 3.86 (0.94–7.36) | 0.24-27.78 |
| Urinary | 0.02 (0.02–0.02) | 0.02-0.02 |
| Peripheral | 0.00 (0.00–0.02) | 0.00-0.06 |
| NUCLEAR MEDICINE | | |
| Gastrointestinal | 2.34 (2.32–2.52) | 2.32-2.52 |
| Urinary | 5.12 (5.06–5.19) | 5.06-5.19 |

mSv = *millisieverts*, *IQR* = *Interquartile range*

2.3.2 Patient cumulative radiation exposure

In this cohort, 50.5% of patients received under 1 mSv CED during their ICU stay, with 9 patients receiving no ionizing radiation (Figure 2-1). Conversely, 33 patients (6.3%) received more than 50 mSv, of which 7 (1.3%) received more than 100 mSv. The highest CED was 199.89 mSv.

Admission diagnosis also contributed to CED. A patient admitted with trauma received significantly more radiation compared to medical and surgical admissions (Figure 2-2A). Finally, patients who had longer length of stay in ICU had greater CED (Figure 2-2B). There was no correlation between age in years ($\rho = -0.059$, p = 0.175) or APACHE III score ($\rho = 0.008$, p = 0.859) to CED.



Figure 2-1: Distribution of patient exposure grouped by cumulative effective dose. Majority of patients received less than 1 mSv exposure to ionizing radiation. Bars represent absolute count of total cohort.



Figure 2-2: Patient factors that contribute to cumulative effective dose. Tukey box and whisker plots of (a) admission diagnosis and (b) length of stay against cumulative effective dose, with italicised superscripts denoting statistically significant differences (p < 0.05) by Kruskal–Wallis analysis with post hoc Mann–Whitney U test. Patients with an admission diagnosis of trauma had significantly higher cumulative effective dose (CED) compared with medical or surgical diagnoses. CED was higher for longer intensive care unit stays. Box plots represent median and interquartile range (IQR), and whisker plots represent 1.5 × IQR below and above the lower and upper quartiles respectively. Single data points represent outliers.

2.4 DISCUSSION

This study is the first to assess cumulative radiation exposures for a large, critically ill cohort without exclusions, based on significantly extended ICU stays, specific diagnosis categorisation or requirements of at least 1 CT or fluoroscopy procedure^{4–11}. Although this study excluded stays shorter than five days to remove routine post-surgical stays and thereby focus on those patients intentionally in the ICU, this study did include all radiological exams which used ionizing radiation, not just high dose procedures, and included all admission diagnoses, not just trauma or emergency patients. Therefore, this study is more representative of a typical ICU population and its ionizing radiation exposure than some previous reports. Among previous studies, the median CED was largely overestimated due to these inclusion criteria and ranged from 1.5-104 mSv. However, most patients from this study received less than 1 mSv CED during their stay in the ICU. These exposures are less than both the internationally-recommended exposure limit for members of public (1 mSv) and the man-made portion of the average Australian's yearly radiation exposure (1.7 mSv)^{20,26}. However, these patients would have received this exposure over a shorter time period (median stay, 7 days) compared to the year-long period of the public limit exposure and the average Australian's exposure.

Only 6% of patients received more than 50 mSv and 2% above 100 mSv, the upper limit of the low dose radiation spectrum. The health-related risks associated with exposures at these levels are negligible with the large majority of studies showing no harmful effects below 100 mSv. The most accepted risk assessment model uses the Linear No-Threshold model and predicts that for every 10mSv of radiation, the excess relative risk of cancer incidence increases 0.06%²⁰. This is on top of the baseline cancer incidence of 40-42% Australia and USA and 50% for UK^{21,27,28}. Thus, these patients, receiving above 100 mSv cumulatively, have a postulated increased risk of cancer from 40-42% to 40.56-42.56%. However, these radiation protection guidelines overcompensate this risk by using modelling based on the atomic bomb survivors and misleading epidemiological studies, which inaccurately extrapolate the risk of low dose radiation from high doses^{29,30}. Therefore, although some guidelines calculate increased cancer and other health-related risks, through harnessing the radiation protection modelling, it is the position of many governing bodies to refrain from making such bold statements when dealing with exposures below 100 mSv^{20,22,23}, to which most diagnostic radiography exposures are found¹³. Long-term, population-based follow up of patients will be required to further decipher this perceived risk.

From this cohort, both trauma categorisation and length of ICU stay were associated with a larger cumulative effective dose (CED). An admission categorisation of trauma typically encompasses patients with multiple sites of major injury, including fractured bones and internal bleeding, leading to scans of most regions of the body. Due to the severity of the injury, secondary follow up scans are also likely leading to increased CED. The length of ICU stay is predominately dictated by the health status of the patient; with sicker or more critical patients associated with longer ICU stays. Continuous monitoring and re-diagnosis occur in these patients, typically using diagnostic radiation, which additively contributes to higher CED. These, along with a variety of other factors, have also been previously reported in the literature as having an associations to larger CED in ICUs including, active malignancies, readmission to ICU, number of radiology exams, CT scans, and fluoroscopy minutes^{6–9}.

Consistent with ICUs and hospitals around the world, chest X-rays were the most frequent individual procedure but contributed very little to CED^{6,9,11,31–33}. The universal high frequency is likely due to clinical practice and ease of access, with most ICUs having portable X-ray machines. While daily routine chest X-rays were common practice, it was reported than only 2.3% result in a change of management, most often adjustment of antibiotic treatment, implanted devices or central lines³⁴. Therefore, practice is moving to clinically indicative administration, saving hospital and patient costs, radiation exposure, and time^{35,36}.

The median dose of each modality calculated in this cohort is largely consistent with previously published literature, except for the abdominal CT¹³. Typically, abdominal scans result in the highest effective dose of this modality, due to the large number of radiosensitive organs within this region^{6,9–11,13,20,31–33}. However, the average abdominal CT in this audit was almost 3x higher than reported in Mettler *et al*¹³. This variation could be due to the defining regions of the scans. It was clinical practice in this ICU to scan both

68

abdominal and pelvic regions together rather than separately, thereby increasing scan time and radiation exposure to those organs.

This retrospective audit is of a single, medium-sized tertiary centred study and thus these results are limited only to this centre. No conclusions can be made regarding ionizing radiation exposures to ICU patients in other centres without completion of a multicentred approach. It is therefore difficult to discuss these results as being universal across centres in Australia or elsewhere, however, in this centre, majority of patients received negligible doses of ionizing radiation.

Another limitation was that a CT scan was counted as per prescribed regardless of number of actual scans during the procedure. Some CT procedures, such as CT angiograms, require multiple phases to observe how the contrast is circulated around the body. At different time periods different organs are better visualised based on where the contrast is during circulation. Therefore, these procedures can contain multiple scans — for example, a pre-contrast scan, an arterial scan, a venous scan, a delayed scan etc — that are completed at various time intervals after contrast injection, causing the patient to receive a higher dose. Given that some dose reports additively counted the scans as one procedure, separation of these multiphase scans was not possible and, thus, all multiphase CT procedures were reported as one, which explains the large variation in CT dose

2.5 CONCLUSION

Medical radiography is a necessary component of medicine, perhaps particularly in ICU settings. This audit demonstrates that the exposure these patients receive from radiography in a western, tertiary hospital, mixed ICU is relatively small, posing a negligible risk against the potential life-saving benefits. However, further study should be undertaken to identify if these results can be replicated in other tertiary care centres, around Australia and internationally, using similar study criteria.

2.6 REFERENCES

- 1. Webb D V, Solomon SB, Thomson DEM. Background radiation levels and medical exposure levels in Australia. *Radiat Prot Aust.* 1999;16(2):25-32.
- 2. National Council on Radiation Protection and Measurements. *Ionizing Radiation Exposure of the Population of the United States. NCRP Report No. 93.* National Council on Radiation Protection and Measurements; 2009.
- 3. Hayton A, Wallace A, Marks P, Edmonds K, Tingey D, Johnston P. Australian per caput dose from diagnostic imaging and nuclear medicine. *Radiat Prot Dosimetry*. 2013;156(4):445-450. doi:10.1093/rpd/nct101.
- 4. Slovis BH, Shah KH, Yeh DD, Seethala R, Kaafarani HMA, Eikermann M, Raja AS, Lee J. Significant but reasonable radiation exposure from computed tomography-related medical imaging in the ICU. *Emerg Radiol*. 2016;23(2):141-146. doi:10.1007/s10140-015-1373-y.
- Kim PK, Gracias VH, Maidment AD, O'Shea M, Reilly PM, Schwab CW. Cumulative radiation dose caused by radiologic studies in critically ill trauma patients. J Trauma. 2004;57(3):510-514. doi:10.1097/01.TA.0000141028.97753.67.
- 6. Rohner DJ, Bennett S, Samaratunga C, Jewell ES, Smith JP, Gaskill-Shipley M, Lisco SJ. Cumulative total effective whole-body radiation dose in critically ill patients. *Chest*. 2013;144(5):1481-1486. doi:10.1378/chest.12-2222.
- 7. Salottolo K, Bar-Or R, Fleishman M, Maruyama G, Slone DS, Mains CW, Bar-Or D. Current utilization and radiation dose from computed tomography in patients with trauma. *Crit Care Med.* 2009;37(4):1336-1340. doi:10.1097/CCM.0b013e31819d6739.
- 8. Sudhir K, Shradda N, Moghekar A, Changarath Vijayan A, Dong F, Adhi F, Yella J, Mehta A, Duggal A, Guzman J. Annual Cumulative Radiation Dose Exposure in a Cohort of Young Medical Intensive Care Unit (MICU) Survivors-Characteristics and Predictors. In: Am J Respir Crit Care Med; 2016:193. doi:10.1164/AJRCCM-CONFERENCE.2016.193.1 MEETINGABSTRACTS.A2760.
- 9. Moloney F, Fama D, Twomey M, O'Leary R, Houlihane C, Murphy KP, O'Neill SB, O'Connor OJ, Breen D, Maher MM. Cumulative radiation exposure from diagnostic imaging in intensive care unit patients. *World J Radiol*. 2016;8(4):419. doi:10.4329/wjr.v8.i4.419.
- 10. Tien HC, Tremblay LN, Rizoli SB, Gelberg J, Spencer F, Caldwell C, Brenneman FD. Radiation exposure from diagnostic imaging in severely injured trauma patients. *J Trauma Inj Infect Crit Care*. 2007;62(1):151-156. doi:10.1097/TA.0b013e31802d9700.
- 11. Leeson A, Adiotomre E, Mannings A, Kotnis N, Morrison G, Wiles M. Cumulative radiation dose due to diagnostic investigations in seriously injured trauma patients admitted to critical care. *J Intensive Care Soc.* 2015;16(1):12-17. doi:10.1177/1751143714551250.
- 12. Krishnan S, Moghekar A, Duggal A, Yella J, Narechania S, Ramachandran V, Mehta A, Adhi F, Vijayan AKC, Han X, Wang X, Dong F, Martin C, Guzman J. Radiation Exposure in the Medical ICU: Predictors and Characteristics. *Chest*. 2018;153(5):1160-1168. doi:10.1016/j.chest.2018.01.019.
- 13. Mettler FA, Huda W, Yoshizumi TT, Mahesh M. Effective doses in radiology and diagnostic nuclear medicine: a catalog. *Radiology*. 2008;248(1):254-263. doi:10.1148/radiol.2481071451.
- 14. Shiralkar S, Rennie A, Snow M, Galland RB, Lewis MH, Gower-Thomas K. Doctors' knowledge of radiation exposure: questionnaire study. *BMJ*. 2003;327(7411):371-372. doi:10.1136/bmj.327.7411.371.
- 15. Ditkofsky N, Shekhani HN, Cloutier M, Chen ZN, Zhang C, Hanna TN. Ionizing Radiation Knowledge Among Emergency Department Providers. *J Am Coll Radiol*. 2016;13(9):1044-1049.e1. doi:10.1016/j.jacr.2016.03.011.
- 16. Soye JA, Paterson A. A survey of awareness of radiation dose among health professionals in Northern Ireland. *Br J Radiol.* 2008;81(969):725-729. doi:10.1259/bjr/94101717.
- 17. Brown N, Jones L. Knowledge of medical imaging radiation dose and risk among doctors. J Med Imaging Radiat

Oncol. 2013;57(1):8-14. doi:10.1111/j.1754-9485.2012.02469.x.

- 18. Babaloui S, Parwaie W, Refahi S, Abrazeh M, Afkhami Ardekani M. Awareness Assessment of Nurses in the OR, ICU, CCU, and PICU About Radiation Protection Principles of Portable Radiography in Hospitals of Bandar Abbas, Iran. *Journal of Radiology Nursing*. February 1, 2018.
- 19. Lam DL, Larson DB, Eisenberg JD, Forman HP, Lee Cl. Communicating potential radiation-induced cancer risks from medical imaging directly to patients. *Am J Roentgenol*. 2015;205(5):962-970. doi:10.2214/AJR.15.15057.
- The International Commission on Radiological Protection. The 2007 Recommendations of the International Commission on Radiological Protection. ICRP Publication 103. Ann ICRP. 2007;37(2-4). doi:10.1177/ANIB_37_2-4.
- 21. Australian Bureau of Statistics. *Australian Health Survey*. Australia; 2013.
- 22. Health Physics Society. Radiation Risk in Perspective Position Statement of the Health Physics Society.; 2010.
- 23. American Association of Physicists in Medicine. *Position Statement on Radiation Risks from Medical Imaging Procedures. PP 25-C.*; 2018.
- 24. Deak PD, Smal Y, Kalender WA. Multisection CT protocols: sex- and age-specific conversion factors used to determine effective dose from dose-length product. *Radiology*. 2010;257(1):158-166. doi:10.1148/radiol.10100047.
- 25. Hart D, Wall BF. NRPB-W4 Radiation Exposure of the UK Population from Medical and Dental X-ray Examinations.
- 26. Internation Atomic Energy Agency. Radiation protection and safety of radiation sources: international basic safety standards general safety: No GSR part 3. *IAEA Saf Stand*. 2014;3:436. doi:STI/PUB/1578.
- Howlader N, Noone A, Krapcho M, Miller D, Bishop K, Altekruse S, Kosary C, Yu M, Ruhl J, Tatalovich Z, Mariotto A, Lewis D, Chen H, Feuer E, Cronin K. Cancer Statistics Review, 1975-2013 - SEER Statistics. SEER Cancer Stat Rev 1975-2013, Natl Cancer Institute Bethesda, MD. 2015.
- 28. Ahmad AS, Ormiston-Smith N, Sasieni PD. Trends in the lifetime risk of developing cancer in Great Britain: Comparison of risk for those born from 1930 to 1960. *Br J Cancer*. 2015;112(5):943-947. doi:10.1038/bjc.2014.606.
- 29. Calabrese EJ, O'Connor MK. Estimating Risk of Low Radiation Doses A Critical Review of the BEIR VII Report and its Use of the Linear No-Threshold (LNT) Hypothesis. *Radiat Res.* 2014;182(5):463-474. doi:10.1667/rr13829.1.
- Siegel JA, Greenspan BS, Maurer AH, Taylor AT, Phillips WT, Van Nostrand D, Sacks B, Silberstein EB. The BEIR VII Estimates of Low-Dose Radiation Health Risks Are Based on Faulty Assumptions and Data Analyses: A Call for Reassessment. J Nucl Med. February 2018:jnumed.117.206219. doi:10.2967/jnumed.117.206219.
- Lutterman AC, Moreno CC, Mittal PK, Kang J, Applegate KE. Cumulative Radiation Exposure Estimates of Hospitalized Patients from Radiological Imaging. J Am Coll Radiol. 2014;11(2):169-175. doi:10.1016/j.jacr.2013.08.028.
- 32. Fazel R, Krumholz HM, Wang Y, Ross JS, Chen J, Ting HH, Shah ND, Nasir K, Einstein AJ, Nallamothu BK. Exposure to low-dose ionizing radiation from medical imaging procedures. *N Engl J Med*. 2009;361(9):849-857. doi:10.1056/NEJMoa0901249.
- 33. Mettler F, Wiest P, Locken J, Kelsey C. CT scanning: patterns of use and dose. *J Radiol Prot*. 2000;20:353-359. doi:10.1088/0952-4746/20/4/301.
- Cruz J, Ferra M, Kasarabada A, Gasperino J, Zigmund B. Evaluation of the Clinical Utility of Routine Daily Chest Radiography in Intensive Care Unit Patients With Tracheostomy Tubes. J Intensive Care Med. 2016;31(5):333-337. doi:10.1177/0885066614538393.
- 35. Oba Y, Zaza T. Abandoning Daily Routine Chest Radiography in the Intensive Care Unit: Meta-Analysis. *Radiology*. 2010;255(2):386-395. doi:10.1148/radiol.10090946.
- 36. Ganapathy A, Adhikari NK, Spiegelman J, Scales DC. Routine chest x-rays in intensive care units: a systematic review and meta-analysis. *Crit Care*. 2012;16(2):R68. doi:10.1186/cc11321.

CHAPTER 3

CUMULATIVE RADIATION IN HOSPITALISED PREGNANT PATIENTS: A RETROSPECTIVE AUDIT OF IONIZING RADIATION EXPOSURE

SUMMARY

Ionizing radiation is a valuable tool in modern medicine, however, in the case of pregnancy, the potential effects from ionizing radiation are not only to the mother but also to the foetus. As a first step to understanding the risks and benefits of radiation exposure to pregnant patients, we aimed to assess the cumulative levels of ionizing radiation received by pregnant patients during a single admission to a tertiary hospital. From January 2013 till December 2017 inclusive, 28,275 pregnant patients were admitted to Flinders Medical Centre, Adelaide, South Australia. There were 3742 patients who received a radiological procedure, including ultrasound and MRI, but only 547 patients received ionizing radiation. These 547 patients underwent 841 procedures totalling 601.2 mSv of ionizing radiation. The median cumulative effective dose (CED) was 0.02 mSv and only 5 patients received more than 10 mSv, with 19.07 mSv the highest dose received. Stays longer than 10 days had significantly higher CED as did those with cardiovascular or cerebral related admission. The median foetal CED was 0.01 mSv with only 3 foetuses receiving more than 10 mSv. Overall, only 1.9% of patients admitted over the 5 years received ionizing radiation, with

more than 50% of these patients receiving less than a chest X-ray (0.02 mSv). These results suggest that pregnant patients are exposed to relatively low doses of ionizing radiation, in both the individual procedure and cumulative doses, and thus the relative risk to the foetus is negligible. If clinicians are avoiding foetal radiation exposure due to the radiation risk, they should recognise that this risk is negligible from diagnostic procedures.

3.1 INTRODUCTION

We are continuously exposed to ionizing radiation from both natural and artificial sources. The exposure varies around the world, with the average natural background level for Australian's being 1.5 mSv annually whilst the United States of America (USA) residents receive approximately 3.1 mSv^{1,2}. In addition to this, levels can increase with artificial exposures, another 1.7 mSv for Australia and 3.1 mSv for the USA, totalling an average yearly exposure of 3.2 mSv and 6.2 mSv respectively^{2,3}. Of these artificial exposures, most derive from medical sources with ionizing radiation playing a crucial role in medicine; aiding in both diagnosis and therapy.

Diagnostic radiation typically utilises X-irradiation, for conventional radiography, computed tomography (CT) and fluoroscopy, or gamma-emitting radioisotopes, such as Technetium-99 used in nuclear medicine. Average patient doses can range between 0.001 – 16 mSv, with a chest X-ray about 0.02 mSv and an abdominal CT 8 mSv⁴. When a female of childbearing age requires diagnostic radiation, confirmation of pregnancy is required and can alter the course of diagnosis⁵. During pregnancy, exposure levels can vary significantly between mother and foetus due to the procedure's field of view, the delivery method of nuclear medicine or the shielding used. When the foetus is not in the direct field of the Xrays, such as head and neck radiography, it will only receive a fraction of the dose, limited to scatter radiation⁶. When the radionuclide is administered via a route which is unlikely to travel near the foetus, for example, inhaled gas, then the dose will be low, compared to methods which come in close contact, such as I.V. injection, which may transfer into placental circulation^{7,8}. When pregnant patients receive ionizing radiation, they generally have lead shielding applied to areas not intended for the scan. The lead will absorb radiation and reduce to the dose to the tissues underneath. Lead shielding is applied over the uterus to reduce the dose to the foetus during diagnostic radiation.

Among hospitalised patients, diagnostic and interventional radiology is potentially used least for pregnant patients due to the additional fear of ionizing radiation exposure to the unborn foetus. This fear stems from the reported stochastic effects of cancer risk^{9,10}, although the reality of this risk, particularly from diagnostic levels of radiation, is continually challenged^{11–13}. Similarly, for deterministic effects, there is uncertainty as to the effect of low doses of radiation but the current understanding is that adverse outcomes increase with higher doses and earlier exposures^{14–16}. Although controversial within the radiation protection profession, there is no universally recognised threshold for the effects of radiation. The consensus is that there is technically no 'safe level' of ionizing radiation and that any dose carries a potential risk to the developing foetus. Regardless, the possible risk of ionizing radiation to the foetus should always be weighed against the clinical benefit from the diagnostic procedure and against the risk of not receiving that procedure.

As a first step, it is important that we accurately understand the current levels of radiation these patients, and more specifically their unborn foetuses, are receiving. This study will inform clinicians regarding their prescribing practices and the resultant exposures thereby aiding their future assessment of the risk-to-benefit analysis. The aim of this study was to conduct a retrospective audit of cumulative ionizing radiation exposure of all pregnant patients admitted to Flinders Medical Centre over a 5-year period.

3.2 METHOD

3.2.1 Study cohort

The study cohort included patients admitted to Flinders Medical Centre, Adelaide, Australia, between 1st January 2013 and 31st December 2017 with a pregnancy-related admission code. In some circumstances, patients were coded with pregnancy status, but the primary diagnosis code was unrelated to pregnancy; these patients were still included. In other circumstances, patients were coded as pregnant due to suspicion of pregnancy. Cases of suspected pregnancy which were later confirmed not to be pregnant were removed. This audit was reviewed and approved by the South Adelaide Clinical Human Research Ethics Committee (OFR # 172.18) in line with the requirements of the *National Statement on Ethical Conduct in Human Research* (2007) and *the Australian Code for the Responsible Conduct of Research* (2007) and the requirement of informed consent was waived.

3.2.2 Data collection

Hospital records from Open Architecture Clinical Information System (OACIS), and Health Information Portal (HIP) and the radiological records from Picture Archiving and Communication System (PACS) were used to collect demographic information, clinical data and radiology reports for each patient.

All ionizing forms of radiation were recorded including conventional X-rays, Computed Tomography (CT), fluoroscopy and nuclear medicine, as well as non- ionizing procedures, such as ultrasound and magnetic resonance imagining.

3.2.3 Calculation of effective dose

Effective doses for each procedure were calculated using values from patient radiology reports and reported conversion factors by Deak *et al* and Hart and Wall^{17,18}. In short, values for Dose Area Product (DAP) or Dose Length Product (DLP) produced from the radiology reports were collected and converted to effective dose (mSv) via multiplication by tissue-specific conversion factors (Table 3-1). For conventional X-rays, accepted diagnostic reference levels were used for each individual exam due to lack of information in reports, except in cases where enough information enabled calculation of effective doses.

Effective doses for foetal exposures were derived using values from Sharp et al, Russel et al and Hauer et al^{6,7,19}. In brief, average procedure doses were used for common procedures, where available (Table 3-1). Maternal doses from procedures in which the field of view encompassed the foetus were equated to equivalent foetal doses, such in abdominal and pelvic X-ray or CT.

| | Conversion factors | Foetal doses |
|------------------------|--------------------|-------------------|
| | | (mSv) |
| CONVENTIONAL X-RAY | | |
| Head/neck | 0.000028 | 0.01ª |
| Chest | 0.00012 | 0.01ª |
| Abdominal and pelvic | 0.00027 | MD |
| Extremities | 0.00001 | 0.01ª |
| COMPUTED TOMOGRAPHY | | |
| Head | 0.0019 | 0.01 ^a |
| Neck | 0.0051 | 0.01 ^a |
| Chest | 0.0145 | 0.06 ^a |
| Abdominal and pelvic | 0.0153 | MD |
| FLUOROSCOPY | | |
| Cardiovascular | 0.00012 | 0.3 ^c |
| Gastrointestinal | 0.00026 | MD |
| Urinary | 0.00018 | MD |
| Peripheral | 0.00001 | 0.01 ^c |
| NUCLEAR MEDICINE | | |
| ^{99m} Tc MAA | 0.011 | 0.75 ^b |
| ^{99m} Tc Tgas | 0.015 | 0.3 ^b |

Table 3-1: Conversion factors for each modality, divided by body region scanned or radiopharmaceutical used adopted from Deak *et al* and Hart and Wall^{17,18}, and foetal doses following common diagnostic procedures, adapted from Sharp *et al*, Russel *et al* and Dauer *et al* ^{6,7,19}.

mSv = *millisieverts*, *CT* = *computed tomography*,

99mTc = Technetium-99m; MMA = macro aggregated albumin; Tgas = Technegas

Dose from a = Sharp et al, b = Russel et al, c = Hauer

MD = *maternal dose*

As this study focuses on radiation exposures in pregnancy, only patients who had live pregnancies at the time of the procedure were included. No radiological information was collected for scans during pre-conception, non-viable pregnancies or postpartum.

3.2.4 Statistics

All outcomes measured in this audit were non-normally distributed, therefore data was expressed as median (interquartile range (IQR)) and assessed using non-parametric tests. Categorical variables, including patient indigenous status, gestational age of the foetus, admission unit, admission diagnosis and length of stay, were assessed by Kruskal-Wallis analysis with *post hoc* Mann-Whitney U test. The scale variable of age was assessed using Spearman rank-order correlations. Statistical significance was accepted as p < 0.05.

3.3.1 Overall patient cohort

There were 28,275 patients admitted to Flinders Medical Centre with a pregnancyrelated admission over the 5-year period of this study, Figure 3-1. A total of 3742 (13.24%) patients received 5310 diagnostic radiological procedures. Of these patients, 3099 patients (82.82%) received an ultrasound, 96 patients (2.57%) received an MRI and 547 patients (14.62%) received a procedure using ionizing radiation: X-ray, CT, fluoroscopy or nuclear medicine (Table 3-2). Conventional X-ray was the most frequent procedure then CT scans, 11.45% and 2.47% respectively. As ultrasound and MRI do not produce ionizing radiation, the final exposed cohort was reduced to 547 patients, Figure 3-1.

3.3.2 Exposed patient cohort

Within the exposed cohort of 547 patients, the median age was 33 years old (IQR, 28-38) with a median length of stay of 3 days (1-7), Table 3-3. These patients were admitted with either pregnancy or non-pregnancy related primary diagnosis, with delivery the main reason for pregnancy-related admissions and trauma the main reason for non-pregnancy related admissions. The median cumulative effective dose (CED) of the cohort was 0.02 mSv (IQR, 0.02-0.75) and did not significantly change based on indigenous status (p = 0.291).



Figure 3-1: Flow chart of patient eligibility criteria for inclusion in retrospective audit. FMC = Flinders Medical Centre

Table 3-2: Distribution of patient and procedure frequency of pregnant patients who received any diagnostic radiation during the length of stay.

| | Contribution to frequency | Contribution to frequency |
|----------------------------|---------------------------|---------------------------|
| | of patients | of procedures |
| | (n = 3742 patients) | (n = 5310 procedures) |
| NON-IONIZING RADIATION | 84.18% | 84.15% |
| Magnetic resonance imaging | 2.57% | 1.85% |
| Ultrasound | 82.82% | 82.3% |
| IONIZING RADIATION | 14.62% | 15.85% |
| Conventional X-ray | 12.08% | 11.45% |
| Computed tomography | 3.13% | 2.47% |
| Fluoroscopy | 1.39% | 1.13% |
| Nuclear medicine | 1.12% | 0.79% |
| Table 3-3: Summary | / of | patient | demog | raphic | and | clinical | characteristics. |
|--------------------|------|---------|-------|--------|-----|----------|------------------|
| | | | | | | | |

| Demographic and clinical characteristics | Values (%) | | |
|--|-------------|--|--|
| TOTAL NUMBER OF PATIENTS | 547 | | |
| AGE (years), median (IQR) | 33 (28-38) | | |
| INDIGENOUS STATUS | | | |
| Aboriginal | 54 (9.9%) | | |
| Non-indigenous | 474 (86.7%) | | |
| Data not available | 19 (3.5%) | | |
| LENGTH OF STAY (days), median (IQR) | 3 (1-7) | | |
| GESTATIONAL AGE | | | |
| <5 – 13 wks | 69 (12.3%) | | |
| 14 – 19 wks | 61 (10.8%) | | |
| 20 – 25 wks | 80 (14.2%) | | |
| 26 – 34 wks | 194 (34.5%) | | |
| 35 - >37 wks | 159 (28.2%) | | |
| ADMISSION DIAGNOSIS | | | |
| Non-pregnancy related | 195 (35.6%) | | |
| Cardiovascular | 30 (5.5%) | | |
| Cerebral | 17 (3.1%) | | |
| Gastrointestinal | 27 (4.9%) | | |
| Respiratory | 46 (8.4%) | | |
| Trauma | 48 (8.8%) | | |
| Other | 27 (4.9%) | | |
| Pregnancy related | 352 (64.4%) | | |
| Cardiovascular | 43 (7.9%) | | |
| Cerebral | 18 (3.3%) | | |
| Delivery | 155 (28.3%) | | |
| Gastrointestinal | 18 (3.3%) | | |
| Respiratory | 75 (13.7%) | | |
| Other | 43 (7.9%) | | |

IQR = *interquartile range; wks* = *weeks*

Non-pregnancy related Other (n) = Haematological (1), Cancer (1), Genitourinary (3), Medication related (3), Mental disorder (2), Musculoskeletal (2), Dermatological (4), Infection (5), Observational (7).

Pregnancy-related Delivery (n) = False Labour (7), Pre-eclampsia (8), Delivery complications (24) Abortion (26).

Pregnancy-related Other (n) = Cancer (1), gestational diabetes (2), Pelvic abnormality (2), Foetal abnormality (2), Dermatological (2), Haematological (4), Musculoskeletal (5), Severe morning sickness (7), Infection (8), Genitourinary (10).

3.3.3 Distribution of procedure frequency and dose

As seen in Table 3-4, 452 patients (82.63%) received conventional X-rays and 117 patients (21.39%) received CT scans. Fluoroscopy was received by 52 patients (9.51%) while 42 patients (7.68%) received nuclear medicine. Conventional X-ray was the most frequent procedure, accounting for 72.29% of 841 procedures but only contributed 4.28% to the 601.2 mSv received by the cohort. The most frequent individual procedure was a chest Xray accounting for 57.36% of all procedures (Table 3-4). CT scans contributed the most to total dose (74.25%), with chest CT accounting for 46.21% of total dose, but only accounted for 15.58% of all procedures. Although chest CT contributed the most to the total dose, the abdominal and pelvic CT produced the highest median and individual dose of any procedure, 12.00 mSv and 16.31 mSv respectively (Table 3-5). There were 4 abdominal and pelvic CT scans and they produced 4/5 highest dose-producing procedures (Table 3-6). Fluoroscopy was performed more frequently than nuclear medicine but contributed slightly less to total cohort dose, 7.13% vs 4.99% of procedures compared to 10.52 vs 10.95% in dose contribution. Doses for each individual procedure could range significantly, sometimes up to 500-fold (e.g. with gastrointestinal fluoroscopy).

3.3.4 Pregnant patient cumulative radiation exposure

In this cohort, the median CED was 0.02 mSv. The large majority (68.7%) of patients received less than 1 mSv CED during their stay and only 22 patients (4%) received more than 5 mSv (Figure 3-2). Of these, only 5 patients (1%) received more than 10 mSv, with the highest CED recorded as 19.07 mSv.

| | Contribution to | Contributi | on to | Contribu total | ition to |
|----------------------|--------------------|--------------|----------|-------------------|----------|
| | (n = 547 patients) | n = 841 proc | cedures) | (n = 601 | .2 mSv) |
| CONVENTIONAL X-RAY | 82.63% | 72.29% | | 4.28% | |
| Head/neck | 7.13% | | 5.34% | | 0.04% |
| Chest | 70.02% | | 57.36% | | 1.61% |
| Abdominal and pelvic | 7.13% | | 7.01% | | 0.02% |
| Extremities | 3.66% | | 2.49% | | 2.62% |
| COMPUTED TOMOGRAPHY | 21.39% | 15.58% | | 74.25% | |
| Head | 7.50% | | 5.70% | | 17.69% |
| Neck | 1.28% | | 0.83% | | 2.00% |
| Chest | 12.80% | | 8.55% | | 46.21% |
| Abdominal and pelvic | 0.73% | | 0.48% | | 8.35% |
| FLUOROSCOPY | 9.51% | 7.13% | | 10.52% | |
| Cardiovascular | 1.65% | | 1.31% | | 0.02% |
| Gastrointestinal | 6.95% | | 4.75% | | 10.40% |
| Urinary | 0.91% | | 0.59% | | 0.10% |
| Peripheral | 0.73% | | 0.48% | | 0.00% |
| NUCLEAR MEDICINE | 7.68% | 4.99% | | 10.95% | |
| Pulmonary | 7.68% | | 4.99% | | 10.95% |

Table 3-4: Distribution of procedures which use ionizing radiation and the frequency at which they contribute to the number of patients, number of procedures and total dose.

mSv = millisieverts

| | Median effective dose (mSv) | Range of effective dose |
|----------------------|-----------------------------|-------------------------|
| | Median (IQR) | (mSv) |
| CONVENTIONAL X-RAY | | |
| Head/neck | 0.006 (0.006-0.006) | 0.006-0.006 |
| Chest | 0.020 (0.020-0.020) | 0.020-0.020 |
| Abdominal and pelvic | 0.750 (0.750-0.750) | 0.750-0.750 |
| Extremities | 0.002 (0.002-0.002) | 0.002-0.002 |
| COMPUTED TOMOGRAPHY | | |
| Head | 1.83 (1.52-3.06) | 0.47-3.60 |
| Neck | 1.63 (1.02-2.53) | 0.75-2.97 |
| Chest | 3.64 (2.90-4.79) | 0.82-12.46 |
| Abdominal and pelvic | 12.00 (10.42-14.66) | 9.87-16.31 |
| FLUOROSCOPY | | |
| Cardiovascular | 0.002 (0.000-0.010) | 0.000-0.081 |
| Gastrointestinal | 1.102 (0.435-2.613) | 0.014-6.278 |
| Urinary | 0.132 (0.123-0.144) | 0.039-0.164 |
| Peripheral | 0.003 (0.000-0.003) | 0.000-0.004 |
| NUCLEAR MEDICINE | | |
| Pulmonary | 1.53 (1.44-1.64) | 1.30-2.30 |

Table 3-5: Exposures from each modality divided by body region scanned.

mSv = millisieverts, IQR = Interquartile range

Ph.D. Thesis – J. McEvoy McMaster University – Biology Flinders University – CoMPH.

Table 3-6: Top 5 highest dose procedures with patient characteristics and diagnoses.

| Patient | ient Age gestation Procedure | | Procedure | Dose | Admission | Reason for scan | Primary diagnosis | Confounding factors | |
|---------|------------------------------|-----------|--|-------|-----------|------------------------------------|---------------------------------|--|--|
| | | | (mSv) | unit | | | | | |
| 1 | 28 | 26-34wks | Abdominal and pelvic CT plain | 16.31 | ICCU | HELLP syndrome | eclampsia | Kidney disease, cerebral haemorrhage, caesarean delivery of stillborn, asthma, smoker, mental disorder | |
| 2 | 22 | 05-13wks | Abdominal and pelvic CT w contrast | 13.01 | ICCU | motor vehicle accident | Trauma | Fracture of femur, thoracic spine, intracranial injuries, pneumonia | |
| 3 | 29 | 34->37wks | CT pulmonary angiography | 12.46 | OBS | Suspected plural embolism | Caesarean delivery | Gestational diabetes, excess foetal growth, polyhydramnios, tachycardia, smoker | |
| 4 | 33 | 05-13wks | Abdominal and pelvic CT plain | 10.99 | RENAL | Reflux nephropathy post-transplant | Medical abortion w complication | Severe kidney disease, corpus luteum cyst, anaemia | |
| 5 | 40 | 14-19wks | Abdominal and pelvic CT w contrast | 9.87 | ICCU | Intraabdominal sepsis | Medical abortion | Sepsis, ARDS, intestinal perforation, intestinal endometriosis, pelvic infection, anaemia | |

mSv = *millisieverts*, *CT* = *computed tomography*, *w* = *with*

ICCU = *intensive* and *critical* care unit, *OBS* = *obstetrics*, *RENAL* = *renal* ward

ARDS = acute respiratory distress syndrome

Ph.D. Thesis – J. McEvoy McMaster University – Biology Flinders University – CoMPH.



Figure 3-2: Distribution of patient cumulative exposure grouped by cumulative effective dose. Most patients received less than 1 mSv exposure to ionizing radiation. Bars represent absolute count.

There was no significant difference in CED among gestational age groups or admission unit (Figure 3-3, p = 0.085 and p = 0.409, respectively) or significant correlation between CED and age (p = -0.04, p = 0.349). However, as in Figure 3-3, those patients who stayed more than 10 days had significantly higher CED than any other group (p < 0.01). Additionally, admission categories appeared to separate into three distinct groups, regardless of whether the patient was primarily admitted due to pregnancy-associated reasons (p < 0.001). Patients with cardiovascular or cerebral admission received a higher CED than the rest and those with gastrointestinal or trauma-related admissions received a lower CED. Finally, there was a significant difference in CED among year of admission (p< 0.001), with a slight increase in median CED in 2015 and a peak in 2016 before decreasing again in 2017. A breakdown of modality and dose by year is available in Table 3-7.

3.3.5 Foetal cumulative radiation exposure

The median foetal CED was 0.01 mSv, with only 26 foetuses (4.5%) receiving >1 mSv and 3 (0.4%) receiving >10 mSv. The highest exposure to a foetus was 16.33 mSv. Similar to maternal exposures, there was no significant difference among admission unit (p = 0.589) or significant correlation to maternal age (p = 0.021, p = 0.621), but there were slight variations in other factors (Figure 3-4). Mothers who stayed in hospital >10 days had higher foetal exposures than any other group (p < 0.001) and foetuses exposed in 2016, but not 2015, had a higher CED compared to other years (p < 0.001). Maternal admission diagnosis also resulted in differences in foetal exposures (p < 0.001). Foetuses from mothers with cardiovascular admissions, but not cerebral, again received the highest median CED (0.07 mSv) which was significantly higher than the rest of the groups. Although all remaining groups had a median CED of 0.01 mSv, foetuses from the mothers admitted for delivery had significantly different CED compared to all other groups, due to the increased range of exposures. Of these mothers, 24/26 of foetal exposures were above 1 mSv. Unlike maternal exposures, gestational age was significantly associated with a higher CED (p = 0.013).



Figure 3-3: Factors that affect maternal cumulative doses A) Gestational age at exposure, B) Year of admission, C) Admission unit, D) Admission diagnosis and E) Length of stay. A) There was no difference in CED among groups of gestational ages (p = 0.085, n = 69-194). B) There was a significant difference in CED among admission years (p < 0.001, n = 85-121). Median CEDs increased in 2015 and peaked at 2016 before reducing back down in 2017, 0.026 0.487 and 0.020 respectively. C) There was no difference in CED depending on admission unit (p = 0.409, n = 17-354). **D)** There was a significant difference in the CED based on patient admission diagnoses (p < 0.001, n = 35-155), separating into three statistically different levels of CED. Cardiovascular and cerebral related admission received the highest CEDs, 1.49 and 2.91 mSv respectively, whereas patients with gastrointestinal or trauma-related admission received the lowest CEDs, 0.02 and 0.016 mSv. Finally, E), There was a significant difference among length of stay groups (p = 0.04), with patients who stayed more than 10 days having a significantly higher CED, median = 0.16mSv, compared to all other groups, median = 0.02 mSv for all individual groups, n = 77-204. Box plots represent the median and interquartile range (IQR), and whisker plots represent $1.5 \times IQR$ below and above the lower and upper quartiles respectively. Single data points represent outliers. Non-visible medians are due to equal and aligned lower quartile and median lines. Factors were statistically assessed by Kruskal Wallis with post hoc Mann-Whitney U analysis. Superscripts denote a statically significant difference of p < 0.05.

Ph.D. Thesis – J. McEvoy McMaster University – Biology Flinders University – CoMPH.

| | | CON | IVENTIONAL X-RAY | | СТ | | | |
|------------------------------|--|---|---|---|---|---|---|---|
| | n (%) | percent per modality | Sum of Dose mSv | Median dose (IQR) mSv | n (%) | percent per modality | Sum of Dose mSv | Median dose (IQR) mSv |
| 2013 | 112 (18.4%) | 79.4% | 4.9 | 0.02 (0.02-0.02) | 13 (9.9%) | 9.2% | 33.0 | 2.70 (1.67-2.98) |
| 2014 | 158 (26%) | 79.4% | 4.0 | 0.02 (0.02-0.02) | 29 (22.1%) | 14.6%/ | 101.2 | 2.49 (1.73-4.16) |
| 2015 | 116 (19.1%) | 69.5% | 6.4 | 0.02 (0.02-0.02) | 32 (24.4%) | 19.2% | 111.3 | 3.33 (2.50-4.27) |
| 2016 | 118 (19.4%) | 65.6% | 6.5 | 0.02 (0.02-0.02) | 31 (23.7%) | 17.2% | 105.0 | 3.05 (1.57-3.45) |
| 2017 | 104 (17.1%) | 67.5% | 4.1 | 0.02 (0.02-0.02) | 26 (19.8% | 16.9% | 95.9 | 3.39 (1.78-4.57) |
| | | | | | NUCLEAR MEDICINE | | | |
| | | | FLUOROSCOPY | | | NU | CLEAR MEDICINE | |
| _ | n (%) | percent per modality | FLUOROSCOPY Sum of Dose mSv | Median dose (IQR) mSv | n (%) | NU percent per modality | CLEAR MEDICINE Sum of Dose mSv | Median dose (IQR) mSv |
| 2013 | n (%) 9 (15%) | percent per modality 6.4% | FLUOROSCOPY Sum of Dose mSv 3.2 | Median dose (IQR) mSv 0.14 (0.01-0.37) | n (%) 7 (16.7%) | NU percent per modality 5.0% | CLEAR MEDICINE Sum of Dose mSv 11.4 | Median dose (IQR) mSv 1.5 (1.43-1.7) |
| 2013 2014 | n (%) 9 (15%) 11 (18.3%) | percent per modality 6.4% 5.5% | FLUOROSCOPY Sum of Dose mSv 3.2 9.4 | Median dose (IQR) mSv 0.14 (0.01-0.37) 0.08 (0.00-1.13) | n (%) 7 (16.7%) 1 (2.4%) | NU percent per modality 5.0% 0.5% | CLEAR MEDICINE Sum of Dose mSv 11.4 1.4 | Median dose (IQR) mSv 1.5 (1.43-1.7) 1.45 (1.45-1.45) |
| 2013 2014 2015 | n (%) 9 (15%) 11 (18.3%) 11 (18.3%) | <i>percent per modality</i> 6.4% 5.5% 6.6% | FLUOROSCOPY Sum of Dose mSv 3.2 9.4 18.3 | Median dose (IQR) mSv 0.14 (0.01-0.37) 0.08 (0.00-1.13) 0.80 (0.12-3.69) | n (%) 7 (16.7%) 1 (2.4%) 8 (19%) | NU percent per modality 5.0% 0.5% 4.8% | CLEAR MEDICINE Sum of Dose mSv 11.4 1.4 13.2 | Median dose (IQR) mSv 1.5 (1.43-1.7) 1.45 (1.45-1.45) 1.63 (1.62-1.68) |
| 2013 2014 2015 2016 | n (%) 9 (15%) 11 (18.3%) 11 (18.3%) 17 (28.3%) | percent per modality 6.4% 5.5% 6.6% 9.4% | FLUOROSCOPY Sum of Dose mSv 3.2 9.4 18.3 18.7 | Median dose (IQR) mSv 0.14 (0.01-0.37) 0.08 (0.00-1.13) 0.80 (0.12-3.69) 0.65 (0.03-1.65) | n (%) 7 (16.7%) 1 (2.4%) 8 (19%) 14 (33.3%) | NU percent per modality 5.0% 0.5% 4.8% 7.8% | CLEAR MEDICINE Sum of Dose mSv 11.4 1.4 13.2 21.6 | Median dose (IQR) mSv 1.5 (1.43-1.7) 1.45 (1.45-1.45) 1.63 (1.62-1.68) 1.45 (1.36-1.64) |

Table 3-7: Breakdown of frequency and dose information for each modality across each admission year.

IQR = Interquartile range, mSv = millisieverts

Ph.D. Thesis – J. McEvoy McMaster University – Biology Flinders University – CoMPH.



Figure 3-4: Factors that affect foetal cumulative effective doses A) Gestational age at exposure, B) Year of admission, C) Admission unit, D) Admission diagnosis and E) Length of stay. A) There was a significant difference in CED among groups of gestational ages (p = 0.013, n = 69-194). There was a trend that increasing gestational age-related to increased CED however, the median CED was constant among groups, 0.01 mSv. B) There was a significant difference in CED among admission years (p < 0.001, n = 85-121). Admissions in 2016 had significantly higher CED, median = 0.02 mSv, compared to the other groups, median = 0.01 mSv C) There was no difference in CED depending on admission unit (p = 0.409, n = 17-354). D) There was a significant difference in foetal CED based on maternal admission diagnoses (p < 0.001, n = 35-155). Cardiovascular admissions had the highest CED compared to the other admission groups, 0.07 vs 0.01 mSv respectively. Although delivery admission also had a median CED of 0.01 mSv, they were significantly different from the rest of the groups due to the larger and higher distribution of exposures. Finally E), There was a significant difference among length of stay groups (p = 0.04), with mothers who stayed more than 10 days having a significantly higher foetal CED, median = 0.036mSv, compared to all other groups, median = 0.01 mSv for all individual groups, n = 77-204. Box plots represent the median and interquartile range (IQR), and whisker plots represent $1.5 \times IQR$ below and above the lower and upper quartiles respectively. Single data points represent outliers. Non-visible medians are due to equal and aligned lower quartile and median lines. Factors were statistically assessed by Kruskal Wallis with post hoc Mann-Whitney U analysis. Superscripts denote a statically significant difference of p < 0.05.

3.4 DISCUSSION

In this 5-year, retrospective audit, only 1.9% (546/28275) of patients received ionizing radiation, which was in the form of conventional X-ray, CT, fluoroscopy or nuclear medicine. Of these exposures, most patients accumulated very small CED, median = 0.02 mSv, similar to 5 days of background radiation in Australia¹.

To the author's knowledge, this is the first audit of pregnant patients regardless of procedure-specific inclusion criteria, for example: must have received a CT scan, to include all forms of ionizing radiation in Australia. In 2006, a 10-year retrospective audit at a tertiary care centre in the USA had much higher median doses for each modality with calculated foetal doses for conventional radiography = 0.43 mGy, CT = 4.3 mGy, fluoroscopy = 2.91 mGy, nuclear medicine = and 0.41 mGy²⁰. However, during this study, there were several advancements in technology to reduce dose, especially with regards to exposure to venerable patients like foetuses and children, which have continued to now²¹. Additionally, in the scans received by our cohort, the dose reports consistently reported dose reductions of up to 40%, which would significantly reduce the median procedure, as well as cumulative effective, doses.

Other studies focused on foetal doses with inclusion criteria based on the prescription of specific procedures. Some studies focused on fluoroscopic exams and showed that foetal doses were negligible except those from barium enemas which can vary in dose, 1.14-16.27 mGy^{22,23}. The majority of studies focused on CT scans during pregnancy and therefore reported much higher calculated mean foetal doses $4.3 - 24.8 \text{ mGy}^{20,24-29}$. Whilst some were based on patient cohorts^{20,24-26}, others were on anthropomorphic

models^{27–29}, and although there are large variations in mean dose, based on scan location, scan parameters, type of scanner used and method of calculation, the conclusion that abdominal and pelvic CT scans result in the largest dose, remained consistent.

In this cohort, abdominal and pelvic CT scans similarly produced the highest median and individual procedure dose and although there were only 4 abdominal CT scans received, they were all in the top 5 highest dose procedures. During an abdominal/pelvic CT, the foetus is directly exposed and therefore receives a high dose as well. In following the clinical guideline of ALARA, as low as reasonably achievable, clinicians are likely to avoid using this procedure to reduce the dose, and thereby perceived risk, to the foetus. This may explain why the frequency of abdominal and pelvic CT scans are low compared to other patient populations within the hospital^{4,30–33}.

In the 4 patients that received abdominal and pelvic CT scans, the risk of not performing this procedure had to be weighed against the risk of ionizing radiation to the foetus. These scans were performed on patients who had serious primary diagnoses and confounding diagnoses/factors that specifically required ionizing radiation to confirm and manage diagnosis; these were not healthy patients (Table 3-6). Thus, the risk to mother, and thereby foetus, by not performing the procedure was deemed greater than the radiation risk to the foetus.

Choice of procedure during diagnosis may also change due to confirmation of pregnancy status, as in the diagnosis of pulmonary embolism which can be with CT or nuclear medicine³⁴. Chest CT results in higher exposures than respiratory nuclear medicine, with 97.2% (70/72) of chest CT being CT pulmonary angiography (CTPA) and 100% (42/42) of

respiratory nuclear medicine being ventilation and perfusion scans (Table 3-5), which is consistent with the literature^{4,35}. It is important to note, that CTPA will result in a higher dose to the breast tissue, which, depending on stage of gestation, could be undergoing proliferation in preparation for lactation. This breast tissue is therefore at increased risk of mutation and, if repeatedly exposed, possible cancer induction³⁶. As reviewed by Pahade *et al*, it is estimated that CTPA scans result in breast absorbed doses, not effective doses, of 10 - 70 mGy, compared to 0.22 - 0.28 mGy for ventilation-perfusion scans, however, these numbers would be reduced in pregnancy by a reduction in kVp and mAs³⁵. Unfortunately, ventilation and perfusion scans produce a higher dose to the foetus compared to CTPA, as the injected radiopharmaceuticals come in closer proximity to the foetus, 0.8 vs 0.06 mSv respectively^{35,37,38}. All these factors need to be considered when prescribing these procedures.

Several factors affected maternal and foetal doses. Pregnant patients who were admitted under cardiovascular and cerebral diagnosis received significantly higher CED than other admissions (Figure 3-3), presumably due to these scans encompassing organs outside the field of view of the foetus and would, therefore, pose little risk to the foetus. Clinicians would, therefore, have an easier risk-to-benefit analysis with prescribing these scans, and thus a higher volume of these higher-dose procedures was performed, increasing maternal dose. However, foetal doses were only higher in mothers with cardiovascular and not cerebral admission. Both diagnoses had CT scans and general radiology, but only cardiovascular patients had ventilation-perfusion scans. The technetium labelled albumin aggregate (^{99m}Tc-MAA) perfused intravenously in these scans can come into close proximity to the foetus via circulation leading to a higher foetal doses^{7,8}. A length of stay more than 10 days significantly increased maternal and foetal CED, however, this finding is not unexpected, as patients don't usually stay in the hospital unless clinically required or for patient welfare. Due to disease severity or observational requirements, these patients require continual monitoring for disease progression, cumulatively increasing their CED. As with length of stay, the median CED also changed based on year of admission, but not in a discernible trend. Although 2015 was significantly higher than 2013, 2014 and 2017, the median only increased by 0.006 mSv for maternal exposure, less than $1/3^{rd}$ the dose of a chest X-ray⁴, and this small change was not seen in foetal doses. Admissions in 2016 saw a larger maternal median increase to 0.487 mSv and foetal median to 0.02 mSv. It also had a higher amount of CT, fluoroscopy, and nuclear medicine procedures producing higher total exposures per year (Table 3-7). As admissions per year were comparable among years, a higher total CED would result in a higher median CED.

Gestational age of the foetus did not significantly alter the maternal CED in this study, however, it did slightly increase foetal CED, which has been observed in other studies^{27,37}. Increases in gestational age typically align with increases in size. Thus for indirect exposures, such as in chest CT, the foetus becomes closer to the field of view, receiving more scatter radiation and therefore a higher dose³⁷. For direct exposures, a larger mass means that the foetus is not as deep in the uterus, and therefore not as shielded by the mother's abdominal fascia. This decrease in foetal depth is also associated with an increased foetal dose³⁹. Although these factors increased the foetal exposures, the vast majority still received very low exposures, which are associated with negligible risks.

The use of the term 'risk' or 'safe' needs to be interpreted in the context of the benefits versus the risks. This cohort included three foetal exposures above 10 mSv. When modelling high dose data and extrapolating down to low doses of radiation, this 10 mGy exposure has been linked to a small increase in cancer risk, approximately $0.06\%^{10,40,41}$. However, these models were intended for radiation protection standards which may not accurately calculate cancer risk⁴². Additionally, the authors suggest caution in interpretation of this value due to uncertainties in the risk estimate¹⁰. Furthermore, there have been several reviews on the deterministic effect of ionizing radiation on the developing foetus and they all conform to the ideology that these doses are too low to pose a comprehendible risk^{14,16,43,44}. In fact, the American College of Radiologists, the Royal College of Radiologists, the Royal Australian and New Zealand College of Radiologists, and the American College of Obstetrics and Gynaecology state that risk of carcinogenesis, miscarriage, or malformations from foetal doses of diagnostic radiation <50-100 mGy is negligible, and should not be considered causation to terminate the pregnancy^{45–49}. However, understanding of dose and risk by some clinicians and radiologists does not align with this statement 50-54.

To further minimize the risk, radiation protection methods, including lead shielding, can be applied to the patient in areas not intending to be scanned. The intention is that the denser material absorbs the radiation and therefore reduce the dose to the tissues underneath. However, being outside the direct exposure field, these tissues would only receive small amounts of internal scatter radiation, which lead shielding cannot protect against. As reviewed by Marsh and Silosky⁵⁵, there is little evidence to suggest a benefit in lead aprons to reduce foetal dose, with the dose reduction to the uterus during a chest X-ray (0.02mSv) only 4 – 19%^{56,57}. Inadvertently, the use of lead apron may potentially increase individual dose or increase repeat scan rates thereby increasing cumulative dose⁵⁵. Modern imaging can have built-in automatic exposure controls to optimize the dose during the scan which, when recognising the attenuation from lead shielding, compensates by increasing dose. This is generally only seen during fluoroscopy-based procedures, as X-ray and CT scan regions are predefined before scan commences. Misplacement of lead shielding can obscure relevant anatomy and therefore require additional scanning⁵⁸. For these reasons, the American Association of Physicists in Medicine recommend the discontinuation of routine lead shielding, inferring that it is only psychologically beneficial for fearful or anxious patient⁵⁹ Other methods, such as refining scan area or improving collimation, provide just as much dose reduction as lead aprons without the possibility of interfering with automatic exposure controls or interfering with anatomical definition⁶⁰.

This study and its results stimulate two grand questions which require further exploration. Firstly, are clinicians choosing to avoid using ionizing radiation to avoid the perceived radiation risks: this being that if a patient was not pregnant, would they receive the same procedures. The level of in-depth radiation training clinicians receive is minimal, which leads clinicians to believe that radiation is dangerous and should be used sparingly. Radiation itself can be dangerous, but it all comes down to dose. The average LD_{50} for humans is 4.5 Gy, so in this sense it is dangerous, but the doses received during diagnostic radiation, up to 100,000 × less, are $not^{4,61,62}$. Likewise, it is possible to overdose on paracetamol/acetaminophen but the appropriate dosage can be taken safely, even during pregnancy⁶³. But to answer this first grand question would require a clinical study assessing procedures used in patients who are pregnant vs non-pregnant. Disregarding the ethical concerns, a large obstacle for this study is that these two patient populations are inherently different. Most pregnant patients who receive radiation do so because of pregnancy-related problems and therefore would not be present in non-pregnant patients. This makes comparability of these populations very difficult. Another option, such as removing the pregnancy status from a patient cohort and assessing how the clinicians proceed to obtain diagnosis could answer this question but has large ethical or moral impracticalities.

In situations where a disease may present in both pregnant and non-pregnant patients, such as kidney stones, is it warranted to use non-ionizing techniques first, which, if negative, are then followed by ionizing radiation procedures anyway. In pregnant patients, McCollough et al. discuss the diagnosis of suspected kidney stones via ultrasound and CT scan⁴⁴. Specifically, they denote that to reduce the exposure to ionizing radiation, ultrasound should be used first and repeated 24 hours later then if still negative a CT scan should then be used. This method not only increases the amount of time the patient is in pain but also increases the patient length of stay and hospital cost. The radiation dose received by patients in the CT assigned group was larger on day of admission, but the specificity of CT scans means that patients receive less follow up radiological procedures, 5% compared to 40% for ultrasonography, which actually trended lower in total cumulative dose to the patient over a 1 month and 6 month period⁶⁴. So, the intention to reduce dose is not necessarily true. Lastly, although there is no difference in misdiagnosis, serious adverse effects or ED readmission, in the case where a patient is negative for kidney stones, CT scans are advantageous in determining other causes of symptoms than ultrasonography, making it more practical⁶⁴. Thus, regardless if a patient is pregnant or not, should clinicians remove sometimes unnecessary steps to gain the answer quicker at the risk of increasing dose to the patient, which is not necessarily true?

A limitation of this study is the calculation of foetal dose. The most accurate estimation of foetal dose would be dependent on procedure parameters and clinical characteristics including scanner type, kVp, mAs, pitch, x-ray beam collimation, procedure length, procedure time, procedure position, radiopharmaceutical used, amount of radiopharmaceutical, patient size, foetal depth and shielding. This study did not calculate foetal doses based on procedure or clinical factors but instead were based on average foetal doses in the literature applied to each modality and scan position. This limits the individual variability between similar procedures and therefore both over and underestimates individual foetal exposures. The intention was to gain a preliminary overview of foetal exposures and to highlight the procedures which result in large doses due to the proximity to scan field of view or delivery method of radiopharmaceutical. Only by incorporating details of the listed factors above, can realistic and precise foetal doses be calculated, which would improve the accuracy of these results, however, based on this audit's results, the improved foetal doses would still be presumably low.

Another limitation is that this is a single-centred study. No conclusions can be made regarding ionizing radiation levels to admitted pregnant patients at other tertiary care facilities without a multicentred approach. It is thereby difficult to extrapolate these results as common exposures or Australian standards. The finding that foetal doses were

low in this hospital is not surprising given that there is a consistent mindset around the world to keep exposures as low as reasonably achievable^{14,46}.

3.5 CONCLUSION

Diagnosis of disease can be difficult for clinicians without the use of diagnostic radiation. There is a stigma surrounding the use of ionizing radiation during pregnancy which is observable in the very low level of radiation received by the cohort in this audit. Was this because clinicians chose not to give ionizing radiation, or because it was not needed. Regardless, attempts to reduce dose to the foetus will decrease the overall perceived risk to the foetus, but if clinically indicated, the procedure should still be prescribed because the perceived risk is negligible. Further investigation into more accurate measurements of foetal dose should occur as well as an expansion of this analysis of pregnant and foetal exposures to other hospitals around Australia and internationally.

Ph.D. Thesis – J. McEvoy McMaster University – Biology Flinders University – CoMPH.

3.6 REFERENCES

- 1. Webb D V, Solomon SB, Thomson DEM. Background radiation levels and medical exposure levels in Australia. *Radiat Prot Aust*. 1999;16(2):25-32.
- 2. National Council on Radiation Protection and Measurements. *Ionizing Radiation Exposure of the Population of the United States. NCRP Report No. 93.* National Council on Radiation Protection and Measurements; 2009.
- 3. Hayton A, Wallace A, Marks P, Edmonds K, Tingey D, Johnston P. Australian per caput dose from diagnostic imaging and nuclear medicine. *Radiat Prot Dosimetry*. 2013;156(4):445-450. doi:10.1093/rpd/nct101.
- 4. Mettler FA, Huda W, Yoshizumi TT, Mahesh M. Effective doses in radiology and diagnostic nuclear medicine: a catalog. *Radiology*. 2008;248(1):254-263. doi:10.1148/radiol.2481071451.
- 5. Australian Radiation Protection and Nuclear Safety Agency (ARPANSA). *Code of Practice for Radiation Protection in the Medical Applications of Ionizing Radiation (2008).*; 2008.
- 6. Sharp C, Shrimpton JA, Bury RF. *Diagnostic Medical Exposures: Advice on Exposure to Ionising Radiation during Pregnancy.*; 1998.
- 7. Russell JR, Stabin MG, Sparks RB, Watson E. Radiation absorbed dose to the embryo/fetus from radiopharmaceuticals. *Health Phys.* 1997;73(5):756-769. doi:10.1097/00004032-199711000-00003.
- 8. Russell JR, Stabin MG, Sparks RB. Placental transfer of radiopharmaceuticals and dosimetry in pregnancy. *Health Phys.* 1997;73(5):747-755.
- 9. Stewart A, Webb J, Giles D, Hewitt D. Malignant disease in childhood and diagnostic irradiation in utero. *Lancet.* 1956;268(6940):447. doi:10.1093/jac/23.2.247.
- 10. Wakeford R, Little MP. Risk coefficients for childhood cancer after intrauterine irradiation: A review. *Int J Radiat Biol.* 2003;79(5):293-309. doi:10.1080/0955300031000114729.
- 11. Preston DL, Cullings H, Suyama A, Funamoto S, Nishi N, Soda M, Mabuchi K, Kodama K, Kasagi F, Shore RE. Solid cancer incidence in atomic bomb survivors exposed in utero or as young children. *J Natl Cancer Inst.* 2008;100(6):428-436. doi:10.1093/jnci/djn045.
- 12. Rajaraman P, Simpson J, Neta G, De Gonzalez AB, Ansell P, Linet MS, Ron E, Roman E. Early life exposure to diagnostic radiation and ultrasound scans and risk of childhood cancer: Case-control study. *BMJ*. 2011;342(7794):424. doi:10.1136/bmj.d472.
- 13. Boice JD, Miller RW. Childhood and adult cancer after intrauterine exposure to ionizing radiation. In: *Teratology*. Vol 59. ; 1999:227-233. doi:10.1002/(SICI)1096-9926(199904)59:4<227::AID-TERA7>3.0.CO;2-E.
- 14. The International Commission on Radiological Protection. *Biological Effects after Prenatal Irradiation (Embryo and Fetus). ICRP Publication 90.* Vol 90.; 2003.
- 15. Uma Devi P, Baskar R. Influence of gestational age at exposure on the prenatal effects of gamma-radiation. *Int J Radiat Biol.* 1996;70(1):45-52.
- Sreetharan S, Thome C, Tharmalingam S, Jones DE, Kulesza A V., Khaper N, Lees SJ, Wilson JY, Boreham DR, Tai TC. Ionizing Radiation Exposure During Pregnancy: Effects on Postnatal Development and Life. *Radiat Res.* 2017;187(6):647-658. doi:10.1667/rr14657.1.
- 17. Deak PD, Smal Y, Kalender WA. Multisection CT protocols: sex- and age-specific conversion factors used to determine effective dose from dose-length product. *Radiology*. 2010;257(1):158-166. doi:10.1148/radiol.10100047.
- 18. Hart D, Wall BF. NRPB-W4 Radiation Exposure of the UK Population from Medical and Dental X-ray

Examinations.

- 19. Dauer LT, Thornton RH, Miller DL, Damilakis J, Dixon RG, Marx MV, Schueler BA, Vañó E, Venkatesan A, Bartal G, Tsetis D, Cardella JF. Radiation management for interventions using fluoroscopic or computed tomographic guidance during pregnancy: A joint guideline of the Society of Interventional Radiology and the Cardiovascular and Interventional Radiological Society of Europe with endorse. J Vasc Interv Radiol. 2012;23(1):19-32. doi:10.1016/j.jvir.2011.09.007.
- 20. Lazarus E, DeBenedectis C, North D, Spencer PK, Mayo-Smith WW. Utilization of Imaging in Pregnant Patients: 10-year Review of 5270 Examinations in 3285 Patients—1997–2006. *Radiology*. 2009;251(2):517-524. doi:10.1148/radiol.2512080736.
- 21. Lee C, Pearce MS, Salotti JA, Harbron RW, Little MP, McHugh K, Chapple CL, De Gonzalez AB. Reduction in radiation doses from paediatric CT scans in Great Britain. *Br J Radiol.* 2016;89(1060):20150305. doi:10.1259/bjr.20150305.
- 22. Smith I, Gaidhane M, Goode A, Kahaleh M. Safety of endoscopic retrograde cholangiopancreatography in pregnancy: Fluoroscopy time and fetal exposure, does it matter? *World J Gastrointest Endosc*. 2013;5(4):148-153. doi:10.4253/wjge.v5.i4.148.
- 23. Osei EK, Darko J. Foetal Radiation Dose and Risk from Diagnostic Radiology Procedures: A Multinational Study. *ISRN Radiol.* 2012;2013:1-7. doi:10.5402/2013/318425.
- 24. Woussen S, Lopez-Rendon X, Vanbeckevoort D, Bosmans H, Oyen R, Zanca F. Clinical indications and radiation doses to the conceptus associated with CT imaging in pregnancy: a retrospective study. *Eur Radiol.* 2016;26(4):979-985. doi:10.1007/s00330-015-3924-8.
- 25. Goldberg-Stein S, Liu B, Hahn PF, Lee SI. Body CT during pregnancy: Utilization trends, examination indications, and fetal radiation doses. *Am J Roentgenol*. 2011;196(1):146-151. doi:10.2214/AJR.10.4271.
- Ozbayrak M, Cavdar I, Seven M, Uslu L, Yeyin N, Tanyildizi H, Abuqbeitah M, Acikgoz AS, Tuten A, Demir M. Determining and Managing Fetal Radiation Dose from Diagnostic Radiology Procedures in Turkey. *Korean J Radiol.* 2015;16(6):1276-1282. doi:10.3348/kjr.2015.16.6.1276.
- 27. Gilet AG, Dunkin JM, Fernandez TJ, Button TM, Budorick NE. Fetal radiation dose during gestation estimated on an anthropomorphic phantom for three generations of CT scanners. *Am J Roentgenol*. 2011;196(5):1133-1137. doi:10.2214/AJR.10.4497.
- 28. Doshi SK, Negus IS, Oduko JM. Fetal radiation dose from CT pulmonary angiography in late pregnancy: A phantom study. *Br J Radiol.* 2008;81(968):653-658. doi:10.1259/bjr/22775594.
- 29. Kelaranta A, Kaasalainen T, Seuri R, Toroi P, Kortesniemi M. Fetal radiation dose in computed tomography. *Radiat Prot Dosimetry*. 2015;165(1-4):226-230. doi:10.1093/rpd/ncv097.
- 30. Lutterman AC, Moreno CC, Mittal PK, Kang J, Applegate KE. Cumulative Radiation Exposure Estimates of Hospitalized Patients from Radiological Imaging. *J Am Coll Radiol*. 2014;11(2):169-175. doi:10.1016/j.jacr.2013.08.028.
- Krishnan S, Moghekar A, Duggal A, Yella J, Narechania S, Ramachandran V, Mehta A, Adhi F, Vijayan AKC, Han X, Wang X, Dong F, Martin C, Guzman J. Radiation Exposure in the Medical ICU: Predictors and Characteristics. *Chest.* 2018;153(5):1160-1168. doi:10.1016/j.chest.2018.01.019.
- 32. Fabritius G, Brix G, Nekolla E, Klein S, Popp HD, Meyer M, Glatting G, Hagelstein C, Hofmann WK, Schoenberg SO, Henzler T. Cumulative radiation exposure from imaging procedures and associated lifetime cancer risk for patients with lymphoma. *Sci Rep.* 2016;6:35181. doi:10.1038/srep35181.
- 33. Griffey RT, Sodickson A. Cumulative radiation exposure and cancer risk estimates in emergency department patients undergoing repeat or multiple CT. *Am J Roentgenol*. 2009;192(4):887-892. doi:10.2214/AJR.08.1351.
- Anderson DR, Kahn SR, Rodger MA, Kovacs MJ, Morris T, Hirsch A, Lang E, Stiell I, Kovacs G, Dreyer J, Dennie C, Cartier Y, Barnes D, Burton E, Pleasance S, Skedgel C, O'Rouke K, Wells PS. Computed tomographic pulmonary angiography vs ventilation-perfusion lung scanning in patients with suspected pulmonary embolism: A randomized controlled trial. J Am Med Assoc. 2007;298(23):2743-2753. doi:10.1001/jama.298.23.2743.

- 35. Pahade JK, Litmanovich D, Pedrosa I, Romero J, Bankier AA, Boiselle PM. Imaging Pregnant Patients with Suspected Pulmonary Embolism: What the Radiologist Needs to Know. *RadioGraphics*. 2009;29(3):639-654. doi:10.1148/rg.293085226.
- 36. Tsai KKC, Chuang EYY, Little JB, Yuan ZM. Cellular mechanisms for low-dose ionizing radiation-induced perturbation of the breast tissue microenvironment. *Cancer Res.* 2005;65(15):6734-6744. doi:10.1158/0008-5472.CAN-05-0703.
- 37. Winer-Muram HT, Boone JM, Brown HL, Jennings SG, Mabie WC, Lombardo GT, Mabie WC, Jennings SG, Boone JM. Pulmonary Embolism in Pregnant Patients: Fetal Radiation Dose with Helical CT. *Radiology*. 2002;224(2):487-492. doi:10.1148/radiol.2242011581.
- Niemann T, Nicolas G, Roser HW, Müller-Brand J, Bongartz G. Imaging for suspected pulmonary embolism in pregnancy—what about the fetal dose? A comprehensive review of the literature. *Insights Imaging*. 2010;1(5-6):361-372. doi:10.1007/s13244-010-0043-6.
- Angel E, Wellnitz C V, Goodsitt MM, Yaghmai N, DeMarco JJ, Cagnon CH, Sayre JW, Cody DD, Stevens DM, Primak AN, McCollough CH, McNitt-Gray MF. Radiation Dose to the Fetus for Pregnant Patients Undergoing Multidetector CT Imaging: Monte Carlo Simulations Estimating Fetal Dose for a Range of Gestational Age and Patient Size. *Radiology*. 2008;249(1):220-227. doi:10.1148/radiol.2491071665.
- 40. The International Commission on Radiological Protection. The 2007 Recommendations of the International Commission on Radiological Protection. ICRP Publication 103. Ann ICRP. 2007;37(2-4). doi:10.1177/ANIB_37_2-4.
- 41. Doll R, Wakeford R. Risk of childhood cancer from fetal irradiation. *Br J Radiol*. 1997;70(FEB.):130-139. doi:10.1259/bjr.70.830.9135438.
- 42. Calabrese EJ, O'Connor MK. Estimating Risk of Low Radiation Doses A Critical Review of the BEIR VII Report and its Use of the Linear No-Threshold (LNT) Hypothesis. *Radiat Res.* 2014;182(5):463-474. doi:10.1667/rr13829.1.
- 43. Nguyen CP, Goodman LH. Fetal Risk in Diagnostic Radiology. *Semin Ultrasound, CT MRI*. 2012;33(1):4-10. doi:10.1053/j.sult.2011.09.003.
- 44. McCollough CH, Schueler BA, Atwell TD, Braun NN, Regner DM, Brown DL, LeRoy AJ. Radiation Exposure and Pregnancy: When Should We Be Concerned? *RadioGraphics*. 2007;27(4):909-917. doi:10.1148/rg.274065149.
- 45. The International Commission on Radiological Protection. *Pregnancy and Medical Radiation. ICRP Publication* 84. Vol 30. Published for the International Commission on Radiological Protection by Pergamon; 2000.
- 46. American College of Radiologists. *Practice Parameter for Imaging Pregnant or Potentially Pregnant Adolescents and Women With Ionizing Radiation*. Vol 1076.; 2018.
- 47. American College of Obstetricians and Gynecologists. *Guidelines for Diagnostic Imaging during Pregnancy and Lactation*. Vol 130.; 2017. doi:10.1097/AOG.0000000002355.
- 48. Royal Australian and New Zealand College of Radiologists. *Diagnostic Radiology and Pregnancy. RANZCR Position Statement.*; 2017.
- 49. Royal College of Radiologists and the College of Radiographers. *Protection of Pregnant Patients during Diagnostic Medical Exposures to Ionising Radiation.*; 2009.
- 50. Ratnapalan S, Bona N, Chandra K, Koren G. Physicians' Perceptions of Teratogenic Risk Associated with Radiography and CT during Early Pregnancy. *Am J Roentgenol.* 2004;182(5):1107-1109. doi:10.2214/ajr.182.5.1821107.
- 51. Gervaise A, Esperabe-Vignau F, Pernin M, Naulet P, Portron Y, Lapierre-Combes M. Évaluation des connaissances des prescripteurs de scanner en matière de radioprotection des patients. *J Radiol*. 2011;92(7-8):681-687. doi:10.1016/j.jradio.2011.03.023.
- 52. Groves AM, Yates SJ, Win T, Kayani I, Gallagher FA, Syed R, Bomanji J, Ell PJ. CT Pulmonary Angiography versus Ventilation-Perfusion Scintigraphy in Pregnancy: Implications from a UK Survey of Doctors' Knowledge of Radiation Exposure. *Radiology*. 2007;240(3):765-770. doi:10.1148/radiol.2403050910.

- 53. Salerno S, Marchese P, Magistrelli A, Tomà P, Matranga D, Midiri M, Ugazio AG, Corsello G. Radiation risks knowledge in resident and fellow in paediatrics: A questionnaire survey. *Ital J Pediatr.* 2015;41(1):21. doi:10.1186/s13052-015-0130-x.
- 54. Ramanathan S, Ryan J. Radiation awareness among radiology residents, technologists, fellows and staff: where do we stand? *Insights Imaging*. 2015;6(1):133-139. doi:10.1007/s13244-014-0365-x.
- 55. Marsh RM, Silosky M. Patient shielding in diagnostic imaging: Discontinuing a legacy practice. *Am J Roentgenol.* 2019;212(4):755-757. doi:10.2214/AJR.18.20508.
- 56. Matyagin Y V, Collins PJ. Effectiveness of abdominal shields in chest radiography: A Monte Carlo evaluation. *Br J Radiol*. 2016;89(1066):20160465. doi:10.1259/bjr.20160465.
- 57. Jackson G, Brennan PC. Radio-protective aprons during radiological examinations of the thorax: An optimum strategy. *Radiat Prot Dosimetry*. 2006;121(4):391-394. doi:10.1093/rpd/ncl076.
- 58. Frantzen MJ, Robben S, Postma AA, Zoetelief J, Wildberger JE, Kemerink GJ. Gonad shielding in paediatric pelvic radiography: Disadvantages prevail over benefit. *Insights Imaging*. 2012;3(1):23-32. doi:10.1007/s13244-011-0130-3.
- 59. American Association of Physicists in Medicine. AAPM Position Statements on the Use of Patient Gonadal and Fetal Shielding. PP 32-A.; 2019.
- 60. Ryckx N, Sans-Merce M, Schmidt S, Poletti PA, Verdun FR. The use of out-of-plane high Z patient shielding for fetal dose reduction in computed tomography: Literature review and comparison with Monte-Carlo calculations of an alternative optimisation technique. *Phys Medica*. 2018;48:156-161. doi:10.1016/j.ejmp.2018.03.017.
- 61. Mole RH. The LD50 for uniform low LET irradiation of man. *Br J Radiol*. 1984;57(677):355-369. doi:10.1259/0007-1285-57-677-355.
- 62. Lushbaugh CC, Fry SA, Ricks RC. Nuclear reactor accidents: preparedness and medical consequences. *Br J Radiol*. 1987;60(720):1159-1188. doi:10.1259/0007-1285-60-720-1159.
- 63. Cheelo M, Lodge CJ, Dharmage SC, Simpson JA, Matheson M, Heinrich J, Lowe AJ. Paracetamol exposure in pregnancy and early childhood and development of childhood asthma: a systematic review and metaanalysis. *Arch Dis Child*. 2015;100(1):81-89. doi:10.1136/archdischild-2012-303043.
- Smith-Bindman R, Aubin C, Bailitz J, Bengiamin RN, Camargo CA, Corbo J, Dean AJ, Goldstein RB, Griffey RT, Jay GD, Kang TL, Kriesel DR, Ma OJ, Mallin M, Manson W, Melnikow J, Miglioretti DL, Miller SK, Mills LD, Miner JR, Moghadassi M, Noble VE, Press GM, Stoller ML, Valencia VE, Wang J, Wang RC, Cummings SR. Ultrasonography versus Computed Tomography for Suspected Nephrolithiasis. N Engl J Med. 2014;371(12):1100-1110. doi:10.1056/NEJMoa1404446.

CHAPTER 4

IN UTERO EXPOSURE TO LOW DOSE IONIZING RADIATION: LONG TERM CARDIOVASCULAR AND RESPIRATORY OUTCOMES IN C57BL/6J MICE.

SUMMARY

Diagnostic radiation is essential to modern medicine. Advancements in technologies have led to improved diagnosis and patient outcomes, however, the effect of ionizing radiation on the patient is still debated. In the case of pregnancy, the potential effects are not only to the mother but also to the foetus. The aim of this study was to observe if exposure from ionizing radiation during pregnancy alters the development of the cardiovascular and respiratory system of the offspring. Pregnant C57Bl/6 mice were irradiated at gestational day 15 with a ¹³⁷Cs gamma radiation emitting source at 0 mGy (sham), 50 mGy, 300 mGy, or 1000 mGy. After weaning, male and female pup weights and blood pressure measurements were taken weekly until euthanasia at 16-17 weeks postnatal age. Immediately following, the trachea was cannulated, and the lungs and heart excised. The lung was then examined to assess respiratory physiological outcomes. *In utero* exposures to only 1000 mGy caused significant growth reduction, which remained persistent for both male and female pups. There was no significant change in cardiovascular or respiratory outcomes. Overall, intrauterine exposures to ionizing radiation do not appear to significantly alter the development of the cardiovascular and respiratory system in C57Bl/6 pups up to 17 weeks postnatal age.

4.1 INTRODUCTION

Diagnostic radiation is essential to modern medicine, with its use having increased rapidly over the past few decades. Advancements in technologies have led to improved diagnosis and patient outcomes, however, the effect of this radiation on the patient is still debated. In the case of pregnancy, the potential effect is not only to the mother but also to the foetus. Perinatal care is an important and influential period for the development of the foetus, where sudden changes to the intrauterine environment can have long-lasting effects on health trajectory of the foetus¹. This paradigm, known as foetal programming, can affect many organ systems of the foetus including cardiovascular and respiratory systems^{2–4}. Long term changes and dysfunction can lead to disease adding to the already high health care burden cardiovascular and respiratory diseases have in Australia^{5,6}. Further to this, ionizing radiation during pregnancy may also penetrate and directly affect the foetus itself.

Most work that encapsulates irradiation *in utero* has focused on mortality, intrauterine growth restriction, malformations or neurobehavioral outcomes⁷. The larger

the exposure, the higher the risk or severity of these outcomes, but typically, irradiations earlier in gestation lead to mortality whereas later leads more towards malformations or behavioural changes^{7–9}. Overall, the dose required to elicit these responses is above 300 – 500 mGy, but can change significantly due to gestational timing of exposure^{10,11}.

At gestational day 15 in the mouse, the foetus has progressed through implantation and embryogenesis and is now in the foetal development stage¹². The main cardiac structures have been defined, with the refinement of atrioventricular and semilunar valve occurring, and the respiratory system is at the pseudoglandular phase, where the general gland-like structure of the lung is prominent, with a differentiated conducting region^{13,14}. However, most work that investigates *in utero* irradiation at gestational day 15 does not examine the cardiovascular or respiratory systems but instead focuses on intrauterine growth restriction or neurobiological defects^{15–17}. These studies found a dose threshold of 1000 mGy for persistent growth restriction and 500 – 1000 mGy for neurobiological defects.

Previous attempts to relate human *in* utero irradiation exposure to cardiovascular function later in life used chronic or single high dose exposures and typically do not focus on the early foetal development period, and for respiratory function, there have been no studies found thus far^{18–23}. One C57BL/6 study assessed growth and cardiovascular outcomes and saw no significant effect of irradiation on either outcome, except 1000 mGy that caused persistent and reduced weight of the animals²⁴. Despite a significant transport-

induced, stress-related response to their irradiation acclimation procedure, which may have interfered with the cardiovascular data, the growth restriction from 1000 mGy was still apparent. This leaves a paucity of data on the relationship between ionizing radiation exposure during the early foetal developmental stage, particularly of acute medical exposures which are low dose, and the development of the cardiovascular and respiratory system. This study aims to address this relationship to identify if ionizing radiation during late gestation/the early foetal development period, influences cardiovascular function or respiratory physiology in adolescence.

4.2 METHODS

4.2.1 Experimental animals and irradiations

Male and female C57BI/6J mice (Jackson Laboratories, USA) were housed in a 12:12 hour light:dark cycle and allowed food and water *ab libitum*. The mothers were maintained on standard lab mouse chow (9% fat, 44.9% carbohydrate, 19% protein, Teklad Diets Envigo, USA) for the duration of pregnancy and pups received a similar chow post-weaning (6.2% fat, 44.2% carbohydrate, 18% protein, Teklad Diets Envigo, USA) for the duration of the study.

Female mice were paired 2:1 with male mice over a single night to obtain an accurate gestational day 0. The following morning, the females were checked for vaginal plugs, indicative of sexual activity, and then housed singly. Confirmation of pregnancy was

obtained by abdominal palpitations later in gestation. At gestational day 15, pregnant C57Bl/6J female mice were transported across campus to McMaster University's Taylor Radiobiology Source, acclimated *in situ* for 20 minutes and irradiated (sham (irradiation control), 50 mGy, 300 mGy, 1000 mGy) using a ¹³⁷Cs Gamma radiation source (620 Ci, 662 keV energy, 10 mGy/min). To reduce the transport effect seen in Sreetharan *et al*²⁴, animals were only transported once to the source for irradiation and then returned back to the animal facility. All animals were restricted from food and water during irradiations.

Pups were weaned at 3-4 weeks of age, and up to 3 male and 4 female pups were housed together, respectively. To maximise group sizes and control maternal effect, up to 2 male and 2 female pups from any one mother were used to create n = 8 of each sex in each radiation group. Postweaning weight and cardiovascular measurements were taken weekly until euthanasia at 16-17 weeks old.

All described animal procedures were reviewed and approved by the Animal Research Ethics Board at McMaster University (AUP #15-11-26) in line with the requirements of the *Canadian Council on Animal Care Guidelines*.

4.2.2 Cardiovascular measurements

Blood pressure was non-invasively measured in non-anesthetised pups via tail-cuff plethysmography using the CODA8 high throughput non-invasive blood pressure system (Kent Scientific Corporation, USA). This method has been previously described and results in minimal discomfort or stress for the animal²⁵. Systolic blood pressure (SBP), diastolic

blood pressure (DBP), mean arterial blood pressure (MAP), and heart rate (HR) were collected 3 times weekly/animal and averaged, between 7-16 weeks of age.

4.2.3 Respiratory measurements

At 16-18 weeks of age, animals were anaesthetised with 5% Isoflurane, and a tracheotomy then thoracotomy was performed. Maximal blood volume was collected via a cardiac puncture, following which the heart and lungs were excised. The left lung lobe was resected and freeze-dried for lung lobe wet:dry weight analysis. The remaining lobes were degassed (0.5 atm for 60 sec) and lavaged with 3 × 16 mL/Kg body weight aliquots of cold 0.9% saline. Lavage supernatant was assessed for total protein concentration using the Pierce BCA protein assay kit (Thermo Fisher Scientific, USA). Cells were stained using trypan blue and counted on a haemocytometer to calculate cellular infiltrate. Remaining cells were fixed in 4% paraformaldehyde, smeared then stained with Hematoxylin and Eosin and photographed under a BX50 brightfield microscope (Olympus, Japan) to assess differences in cell populations.

4.2.4 Statistical Analysis

Longitudinal data (weight and cardiovascular outcomes) are presented as mean weekly measurements and were analysed using a generalised additive model (GAM) fitted to a generalised linear mixed model (GLMM) accounting for nesting of pups within mothers. Cross-sectional data (respiratory outcomes) are presented as mean and 95% confidence interval and were analysed using generalised linear mixed models with a normal

distribution, accounting for nesting of pups within mothers. A *p*-value < 0.05 was considered statistically significant. Statistical analysis was completed using IBM SPSS v.24 statistical software (IBM Corporation, USA).

4.3 RESULTS

4.3.1 Growth rate

Litter size and sex ratio did not differ between doses of radiation (data not shown). There was no difference in weight at first measurement (age wk4) between any radiation group and sham control. However, there was a significant difference in growth over the longitudinal 12 weeks (Figure 4-1). An *in utero* exposure of 1000 mGy significantly altered the growth trajectory resulting in a 1.42 g reduction in weight overall (p = 0.004). Other doses of radiation were not different from sham over the 12 weeks. Sexes also differed in growth curve (p < 0.001), but growth restriction due to 1000 mGy was similar, 8-10%.

4.3.2 Cardiovascular system

In utero exposure to radiation did not significantly alter the cardiovascular outcomes of SBP, DBP, MAP or HR, p = 0.091, p = 0.889, p = 0.227 and p = 0.060, respectively (Figure 4-2). Therefore, *in utero* irradiation at these doses did not cause hypertension or hypotension over 16 weeks post-weaning. Sexes differed on SBP, DBP and MAP (p < 0.05) but not HR.



Figure 4-1: longitudinal weight measurements. *In utero* exposure to 1000 mGy significantly alters the trajectory of growth compared to sham control as tested using GAM fitted to GLMM (p = 0.004, n = 7-9) resulting in a 1.42g reduction in weight overall. Mice exposed to other doses of radiation were not different to sham. Sexes differed in growth curves (p<0.001). Graphs represent mean weekly measurements. * indicates statistical significance of p < 0.05



Figure 4-2: Longitudinal cardiovascular outcomes of Systolic blood pressure (SBP), diastolic blood pressure (DBP), mean arterial blood pressure (MAP), and heart rate (HR). *In utero* exposure to any dose of radiation did not significantly alter the curve compared to sham for either sex, p = 0.091, p = 0.889, p = 0.227 and p = 0.060, respectively, tested with GAM fitted to a GLMM (n = 7-9). Sexes differ for SBP, DBP, MAP but not HR (p < 0.05). Graphs represent mean weekly measurements, n = 7-9.
4.3.3 Respiratory system

There was no oedema or change in alveolar-capillary barrier integrity, as indicated by lung lobe wet:dry weight ratio (p = 0.618) and total BAL protein concentration (p = 0.450), by *in utero* exposure to ionizing radiation (Figure 4-3). Similarly, cellular infiltrate did not increase due to *in utero* exposure to ionizing radiation, p = 0.753, nor did the populations of cells within the lung differ, p = 0.413 (Figure 4-4). There was no statistical difference between sex for any outcome except BAL cell count. Female animals had 33% more cells at baseline compared to males, p = 0.022.



Figure 4-3: Respiratory physiological outcomes of **A**) wet:dry lung weight ratio and **B**) total protein concentration in BAL. *In utero* exposures to any dose of ionizing radiation did not significantly change the wet:dry ratio or BAL total protein concentrations compared to sham control for either sex, p = 0.450 and p = 0.618, respectively. There was also no difference between sexes, p = 0.152 and p = 0.696, respectively. Data represented as mean and 95% confidence intervals. Statistical significance was tested by GLMM, n = 6-7.



Figure 4-4: Respiratory immunological outcomes of **A**) BAL cell count and **B**) Cell populations within BAL. *In utero* exposures to any dose of ionizing radiation did not significantly change the total cell count or the cell populations within the BAL compared to sham control for either sex, p = 0.753 and p = 0.413, respectively. There was also no difference between sexes for percentages of populations, p = 0.839, but there was for overall cell count, p = 0.022. Data represented as mean and 95% confidence intervals. Statistical significance was tested by GLMM, n = 6-7.

4.4 **DISCUSSION**

The current guidelines worldwide have no limit for ionizing radiation exposure during pregnancy but instead suggest that the exposure be clinically justified on an individual basis and kept as low as reasonably achievable^{10,26–28}. Therefore, exposure of pregnant patients can vary significantly. In this study, the low dose of 50 mGy is representative of a high-end diagnostic procedure, such as a multi-scan abdominopelvic $CT^{29,30}$. The middle range of 300 mGy represents the amount of radiation someone might accumulate for a severe and extended stay in hospital accruing multiple diagnostic procedures over admission. The highest dose of 1000 mGy is well above diagnostic radiation levels but is still below therapeutic levels. It is important to recognise that the radiosensitivity between humans and mice vary significantly. The human LD50 in 30 days for whole-body exposures is between 2000 – 8000 mGy, 4500 mGy for a healthy adult, whereas for C57Bl/6 mice is 8500 - 9000 mGy³¹⁻³³. Thus, this strain of mice is approximately twice as resistant to ionizing radiation than humans, so it could be postulated that *in utero* exposure to 1000 mGy in mice might equate to 500 mGy in humans in terms of its biological effects.

In this study, only the highest dose of *in utero* radiation caused intrauterine growth restriction, which was about 8-10% reduction in body weight consistently over the lifetime of the mouse. The finding that rodent irradiations during pregnancy, and specifically around gestational day 15, induces intrauterine growth restriction has been reported

before for doses less than 1000 mGy, but matching this study's results, persistent growth restriction is only seen from doses \geq 1000 mGy^{17,34–37}. Unfortunately, no conclusion can be made regarding the immediate postnatal growth of these animals, as birthweight and postnatal-preweaning weight were not collected in order to reduce maternal stress and cannibalism. It is therefore uncertain if growth restriction occurred from exposures <1000 mGy and the 'catch up' effect resulted in these animals matching the sham control weight post weaning³⁸. Additionally, no comment can be made about the long-term growth of these animals to the natural endpoint of their life.

Intrauterine growth restriction, as a hallmark of foetal programming, has been linked individually to cardiovascular and respiratory dysfunction^{3,4,39,40}. Low birthweight reduces childhood lung function and increases the risk of hospitalisation in adolescence because of respiratory distress^{40,41}. Similarly, the risk of hypertension appears to be set in foetal development from intrauterine growth restriction³⁹. This study showed no development of hypertension or hypotension due to intrauterine radiation exposure. However, the risk of cardiovascular disease increases with age and often doesn't arise symptomatically until late adulthood so possibly this study's 3-month follow up was not long enough to observe the full effects of the intrauterine radiation⁴². Additionally, it is important to note that there was large intra-group variation for all cardiovascular outcomes which may contribute to the non-statistically significant inter-group variation.

The longer-term effect of radiation on cardiovascular disease incidence or mortality may have implications to public health given the increase in ionizing radiation use in the medical field over the past few decades^{43–45}. Several studies on non-pregnant subjects have recorded that similar-to-medical levels of low dose radiation, but from environmental and occupational exposures, have an association with cardiovascular disease incidence and mortality^{18,19,46}. These studies observe chronic exposures instead of acute and have multiple limitations, including cohort age, adjustments for confounding factors, length of exposure, and accurate dosimetry. Studies of medical exposures found a relationship but only at cumulative doses above 150 mGy or from fractionated radiotherapy doses, whereas others, including a large multi-country investigation, found no relationship at all^{20–23,47–50}. Studies with pregnant patients are scarce. One study from the Hiroshima and Nagasaki atomic bomb survivors found an association of radiation dose to cardiovascular disease later in life but only at exposure >500 mGy, much higher than diagnostic exposures⁵¹. Therefore, radiation may have some effect on the foetal cardiovascular system trajectory, but apparently not at doses received from diagnostic radiation sources.

There was also no change in the respiratory outcomes measured in this study. As reviewed by lles *et al*⁵², non-cardiogenic pulmonary oedema can be caused by damage to the alveolar epithelial layer, directly via reactive oxygen species (ROS) and indirectly via infiltrating leukocytes, or by dysregulation of the fluid clearing process. Less than a second after radiation exposure, ROS are formed, primarily through water radiolysis which can induce perpetual cellular oxidative stress to continue cell damage⁵³. However, this study

found no measurable oedema, alveolar epithelial layer damage, or cellular infiltrate at 16-17 weeks postpartum suggesting the ROS production from water radiolysis was quenched and damage, if created, was repaired before these measurements were taken. Observations of these outcomes at a much earlier time postnatally would be required to assess the immediate effects of *in utero* exposures however, this study suggests that irradiations \leq 1000 mGy do not permanently alter the development of the respiratory system or cause long-term physiological changes indicative of lung injury.

One limitation of this study is that it looked at exposures at a single time point in gestation, as it is well known that similar exposures at different gestational stages result in different outcomes. Although no detrimental cardiovascular or respiratory outcomes were observed in this study, additional work is required at various stages of gestation to fully assess the effect of diagnostic levels of radiation during pregnancy. Equally, additional timings of outcome measurements would strengthen this study, especially to observe the cardiovascular system much later in life or the respiratory system at a younger age.

Another limitation to highlight is that these mice received whole-body irradiations, and thus the foetal dose received is very similar to the maternal dose, whereas clinical application of radiation typically limits exposures to a field of view of interest. For example, a chest CT will result in a much smaller foetal exposure compared to a pelvic CT because the foetus is not in the field of view³⁰. This, therefore, will remove or reduce the direct effect of irradiation on the foetus, although the indirect effects from foetal programming may still exist. Thus, as this experimental model used whole-body exposure, it is not possible to delineate the direct and indirect effects of radiation.

4.5 CONCLUSION

This study aimed to assess the relationship between *in utero* ionizing radiation exposure at GD15 and the development of the cardiovascular and respiratory system. It appears that the radiation exposure did not significantly induce hypertension or hypotension in these mice, nor did it physiologically change the respiratory system, but exposures of 1000 mGy caused intrauterine growth restriction that remained until adolescence. Overall, from this healthy animal model, exposures to diagnostic levels of ionizing radiation during late gestation caused no significant effects on the growth or cardiovascular and respiratory systems, which support the statement that the effects from low dose radiation at this gestational stage are negligible.

- 1. Barker D. The fetal and infant origins of adult disease. *BMJ Br Med J.* 1990;301(156):1111. doi:10.1136/bmj.301.6761.1111
- 2. Barker DJP, Osmond C, Winter PD, Margetts B, Simmonds SJ. Weight in infancy and death from ischaemic heart disease. *Lancet*. 1989;334(8663):577-580. doi:10.1016/S0140-6736(89)90710-1
- Barker DJ, Godfrey KM, Fall C, Osmond C, Winter PD, Shaheen SO. Relation of birth weight and childhood respiratory infection to adult lung function and death from chronic obstructive airways disease. *BMJ*. 1991;303(6804):671-675. doi:10.1136/bmj.303.6804.671
- 4. Barker DJP, Godfrey KM, Gluckman PD, Harding JE, Owens JA, Robinson JS. Fetal nutrition and cardiovascular disease in adult life. *Lancet*. 1993;341(8850):938-941. doi:10.1016/0140-6736(93)91224-A
- 5. Australian Institute of Health and Welfare. *The Burden of Chronic Respiratory Conditions in Australia. Cat. No. BOD 15*, 2017.
- 6. Australian Institute of Health and Welfare. *Deaths in Australia. Cat. No. PHE 229.*; 2019.
- 7. The International Commission on Radiological Protection. *Biological Effects after Prenatal Irradiation (Embryo and Fetus). ICRP Publication 90.* Vol 90.; 2003.
- 8. Rugh R, Wohlfromm M. Prenatal X-Irradiation and Postnatal Mortality. *Radiat Res.* 1965;26(4):493-506. doi:10.2307/3571860
- 9. Russell L, Russel W. The effect of radiation on the preimplantation stages of the mouse embryo. *Anat Rec.* 1950;108.
- 10. The International Commission on Radiological Protection. *Pregnancy and Medical Radiation. ICRP Publication* 84. Vol 30. Published for the International Commission on Radiological Protection by Pergamon; 2000.
- 11. Sreetharan S, Thome C, Tharmalingam S, et al. Ionizing Radiation Exposure During Pregnancy: Effects on Postnatal Development and Life. *Radiat Res.* 2017;187(6):647-658. doi:10.1667/rr14657.1
- 12. Theiler K. The House Mouse: Atlas of Embryonic Development.; 1989. doi:10.1007/978-3-642-88418-4
- 13. Pinkerton KE, Joad JP. The mammalian respiratory system and critical windows of exposure for children's health. *Environ Health Perspect*. 2000;108(SUPPL. 3):457-462. doi:10.1289/ehp.00108s3457
- 14. Krishnan A, Samtani R, Dhanantwari P, et al. A detailed comparison of mouse and human cardiac development. *Pediatr Res*. 2014;76(6):500-507. doi:10.1038/pr.2014.128
- 15. Hossain M, Uma Devi P. Effect of irradiation at the early fetal stage on adult brain function in the mouse: locomotor activity. *Int J Radiat Biol*. 2000;76(10):1397-1402. doi:10.1080/09553000050151673
- 16. Sienkiewicz ZJ, Haylock RGE, Saunders RD. Differential learning impairments produced by prenatal exposure to ionizing radiation in mice. *Int J Radiat Biol*. 1999;75(1):121-127. doi:10.1080/095530099140889
- 17. Nash DJ, Gowen JW. Effects of X-irradiation upon postnatal growth in the mouse. *Biol Bull*. 1962;122(1):115-136. doi:10.2307/1539326
- Muirhead CR, O'Hagan JA, Haylock RGE, et al. Mortality and cancer incidence following occupational radiation exposure: Third analysis of the National Registry for Radiation Workers. Br J Cancer. 2009;100(1):206-212. doi:10.1038/sj.bjc.6604825
- 19. Mcgeoghegan D, Binks K, Gillies M, Jones S, Whaley S. The non-cancer mortality experience of male workers at British Nuclear Fuels plc, 1946-2005. *Int J Epidemiol*. 2008;37(3):506-518. doi:10.1093/ije/dyn018
- 20. Vrijheid M, Cardis E, Ashmore P, et al. Mortality from diseases other than cancer following low doses of ionizing radiation: Results from the 15-Country Study of nuclear industry workers. *Int J Epidemiol*. 2007;36(5):1126-1135.

doi:10.1093/ije/dym138

- 21. Berrington A, Darby SC, Weiss HA, Doll R. 100 years of observation on British radiologists: Mortality from cancer and other causes 1897-1997. *Br J Radiol*. 2001;74(882):507-519. doi:10.1259/bjr.74.882.740507
- 22. Bolotnikova MG, Koshurnikova NA, Komleva NS, Budushchev EB, Okatenko P V. Mortality from cardiovascular diseases among male workers at the radiochemical plant of the "Mayak" complex. *Sci Total Environ*. 1994;142(1-2):29-31. doi:10.1016/0048-9697(94)90070-1
- 23. Kreuzer M, Kreisheimer M, Kandel M, Schnelzer M, Tschense A, Grosche B. Mortality from cardiovascular diseases in the German uranium miners cohort study, 1946-1998. *Radiat Environ Biophys*. 2006;45(3):159-166. doi:10.1007/s00411-006-0056-1
- 24. Sreetharan S, Stoa L, Cybulski ME, et al. Cardiovascular and growth outcomes of C57Bl/6J mice offspring exposed to maternal stress and ionizing radiation during pregnancy. *Int J Radiat Biol*. March 2019:1-9. doi:10.1080/09553002.2019.1589025
- 25. Feng M, Whitesall S, Zhang Y, Beibel M, D'Alecy L, DiPetrillo K. Validation of volume-pressure recording tail-cuff blood pressure measurements. *Am J Hypertens*. 2008;21(12):1288-1291. doi:10.1038/ajh.2008.301
- 26. Royal Australian and New Zealand College of Radiologists. *Diagnostic Radiology and Pregnancy. RANZCR Position Statement.*; 2017.
- 27. American College of Radiologists. *Practice Parameter for Imaging Pregnant or Potentially Pregnant Adolescents and Women With Ionizing Radiation*. Vol 1076.; 2018.
- 28. Royal College of Radiologists and the College of Radiographers. *Protection of Pregnant Patients during Diagnostic Medical Exposures to Ionising Radiation.*; 2009.
- 29. Mettler FA, Huda W, Yoshizumi TT, Mahesh M. Effective doses in radiology and diagnostic nuclear medicine: a catalog. *Radiology*. 2008;248(1):254-263. doi:10.1148/radiol.2481071451
- 30. Sharp C, Shrimpton JA, Bury RF. *Diagnostic Medical Exposures: Advice on Exposure to Ionising Radiation during Pregnancy.*; 1998.
- 31. Nunamaker EA, Artwohl JE, Anderson RJ, Fortman JD. Endpoint refinement for total body irradiation of C57BL/6 mice. *Comp Med*. 2013;63(1):22-28.
- 32. Mole RH. The LD50 for uniform low LET irradiation of man. *Br J Radiol*. 1984;57(677):355-369. doi:10.1259/0007-1285-57-677-355
- 33. Lushbaugh CC, Fry SA, Ricks RC. Nuclear reactor accidents: preparedness and medical consequences. *Br J Radiol*. 1987;60(720):1159-1188. doi:10.1259/0007-1285-60-720-1159
- 34. Kimler BF, Norton S. Behavioral changes and structural defects in rats irradiated in utero. *Int J Radiat Oncol Biol Phys.* 1988;15(5):1171-1177. doi:10.1016/0360-3016(88)90200-3
- 35. Murphree RL, Pace HB. The Effects of Prenatal Radiation on Postnatal Development in Rats. *Radiat Res.* 1960;12(5):495. doi:10.2307/3570971
- Hossain M, Uma Devi P, Bisht KS. Effect of prenatal gamma irradiation during the late fetal period on the postnatal development of the mouse. *Teratology*. 1999;59(3):133-138. doi:10.1002/(SICI)1096-9926(199903)59:3<133::AID-TERA4>3.0.CO;2-T
- 37. Jensh RP, Eisenman LM, Brent RL. Postnatal neurophysiologic effects of prenatal x-irradiation. *Int J Radiat Biol*. 1995;67(2):217-227. doi:10.1080/09553009514550271
- 38. Karlberg J, Albertsson-Wikland K, Karlberg J. *Growth in Full-Term Small-for-Gestational-Age Infants: From Birth to Final Height.*; 1995.
- 39. Law CM, Shiell AW, Newsome CA, et al. Fetal, infant, and childhood growth and adult blood pressure: A longitudinal study from birth to 22 years of age. *Circulation*. 2002;105(9):1088-1092. doi:10.1161/hc0902.104677

- 40. Dezateux C. Low birth weight for gestation and airway function in infancy: exploring the fetal origins hypothesis. *Thorax*. 2004;59(1):60-66.
- 41. Walter EC, Ehlenbach WJ, Hotchkin DL, Chien JW, Koepsell TD. Low birth weight and respiratory disease in adulthood: A population-based case-control study. *Am J Respir Crit Care Med.* 2009;180(2):176-180. doi:10.1164/rccm.200901-0046OC
- 42. Berry JD, Lloyd-Jones DM, Van Horn L, et al. Lifetime Risks of Cardiovascular Disease. *N Engl J Med*. 2012;366(4):321-329. doi:10.1056/nejmoa1012848
- 43. Royal College of Radiologist and National Radiation Protection Board. *Patient Dose Reduction in Diagnostic Radiology*. Vol 1.; 1990.
- 44. Lutterman AC, Moreno CC, Mittal PK, Kang J, Applegate KE. Cumulative Radiation Exposure Estimates of Hospitalized Patients from Radiological Imaging. *J Am Coll Radiol*. 2014;11(2):169-175. doi:10.1016/j.jacr.2013.08.028
- 45. Shrimpton PC, Jones DG, Hillier MC, Wall BF, Le Heron JC, Faulkner K. NRPB-R249 Survey of CT practice in the UK. Part 2: dosimetric aspects. *NRPB-R249*. 1991.
- 46. Howe GR, Zablotska LB, Fix JJ, Egel J, Buchanan J. Analysis of the mortality experience amongst U.S. nuclear power industry workers after chronic low-dose exposure to ionizing radiation. *Radiat Res.* 2004;162(5):517-526.
- 47. Ivanov VK. Late cancer and noncancer risks among chernobyl emergency workers of Russia. *Health Phys.* 2007;93(5):470-479. doi:10.1097/01.HP.0000282195.34508.b0
- 48. Abe O, Abe R, Enomoto K, et al. Effects of radiotherapy and of differences in the extent of surgery for early breast cancer on local recurrence and 15-year survival: An overview of the randomised trials. *Lancet*. 2005;366(9503):2087-2106. doi:10.1016/S0140-6736(05)67887-7
- 49. Swerdlow AJ, Higgins CD, Smith P, et al. Myocardial infarction mortality risk after treatment for hodgkin disease: A collaborative British cohort study. *J Natl Cancer Inst*. 2007;99(3):206-214. doi:10.1093/jnci/djk029
- 50. Darby SC, McGale P, Taylor CW, Peto R. Long-term mortality from heart disease and lung cancer after radiotherapy for early breast cancer: Prospective cohort study of about 300 000 women in US SEER cancer registries. *Lancet Oncol*. 2005;6(8):557-565. doi:10.1016/S1470-2045(05)70251-5
- 51. Shimizu Y, Kodama K, Nishi N, et al. Radiation exposure and circulatory disease risk: Hiroshima and Nagasaki atomic bomb survivor data, 1950-2003. *BMJ*. 2010;340(7739):193. doi:10.1136/bmj.b5349
- 52. Iles KE, Song W, Miller DW, Dickinson DA, Matalon S. Reactive species and pulmonary edema. *Expert Rev Respir Med.* 2009;3(5):487-496. doi:10.1586/ers.09.41
- 53. Azzam El, Jay-Gerin JP, Pain D. Ionizing radiation-induced metabolic oxidative stress and prolonged cell injury. *Cancer Lett*. 2012;327(1-2):48-60. doi:10.1016/j.canlet.2011.12.012

CHAPTER 5

IN UTERO EXPOSURE TO LOW DOSE IONIZING RADIATION AND THE RESPIRATORY RESPONSE TO AN ACUTE LUNG INJURY STIMULUS AT ADOLESCENCE IN BALB/C MICE.

PREFACE

The animal model used in this chapter, BALB/c, is a different strain of mouse compared to chapter 4, C57BI/6J. The decision to change mouse strain during this project was due to the high cannibalism rate observed from C57bl/6 mice, which increased the number of pregnancies required to obtain the group sizes; increasing cost, experimental time, and possibly impacting animal welfare. The high possibility of cannibalism caused researchers to be very particular in the way they handled the pregnant C57BI/6J mice. In comparison, BALB/c mice have a much lower cannibalism rate and were, therefore, easier to handle during pregnancy¹. This change in mouse strain was adopted by everyone in our group for all future experiments in this program of work. A comparison of the effect this species change had on the results, will be undertaken in the overall thesis discussion, chapter 6.

SUMMARY

Diagnostic radiation during pregnancy is a controversial topic with the potential effects not only to the mother but also to the developing foetus. Interruptions to the intrauterine environment, possibly from maternal exposures to radiation, could alter the development of the foetus and long-term outcomes. Not only this, but the immunomodulatory effects from ionizing radiation exposures could alter the response to inflammation later in life. The aim of this study was to observe if in utero exposure to ionizing radiation altered the respiratory immune response during adolescence. Pregnant BALB/c mice were irradiated at gestational day 15 with a ¹³⁷Cs gamma radiation (662 keV energy) emitting source at 0 mGy (sham), 10 mGy, 100 mGy, or 1000 mGy. At 16-17 weeks postnatal age, 3 mg/Kg of lipopolysaccharide (LPS) was instilled intratracheally. Mice were then euthanized 24h later, the trachea was cannulated and the lungs excised. The left lung lobe was resected for wet:dry weight analysis. A bronchoalveolar lavage (BAL) was performed for total cell count, cell differential and protein analysis. Remaining lung tissue was examined for antioxidant levels. Exposures to 100 and 1000 mGy significantly reduced the growth of the pups for both males and females compared to sham animals (p<0.05). Administration of LPS significantly increased lung lobe wet:dry weight ratios, BAL total protein, BAL cell count and transforming growth factor (TGF)- β levels and changed antioxidant levels compared to saline control (p<0.001) indicating establishment of acute lung injury, however, there was no overall effect from ionizing radiation. Overall, in utero exposures to 100 and 1000 mGy late in gestation significantly reduced the growth of the animals but had no effect on the response to a respiratory bacterial stimulus.

5.1 INTRODUCTION

Medical uses of ionizing radiation, in which single procedures typically fall below 100 mGy, play an important role in the diagnosis and management of disease. However, there is a bias surrounding its use during pregnancy due to the fear of its effect on the developing foetus. Currently, there has been little investigation into the effect of foetal radiation exposure on the respiratory system with most work instead focusing on mortality, intrauterine growth restriction, malformation and neurobehavioral outcomes². These effects are dependent on dose, with larger doses (>500 mGy) giving higher risk or severity of effect, and timing during gestation, where early exposures (1st trimester) lead to increased mortality and later exposures lead to growth restriction, malformation or neurobehavioral effects^{2–4}. For both animals and humans, the current overall threshold for effects to the foetus are >100 mGy but individual thresholds can be higher dependent on outcome and foetal development period².

At gestational day 15, the lung is in the pseudoglandular phase, which is a critical time point for the development of the bronchus and bronchioles and the connection of the pulmonary blood supply via peripheral capiliaries^{5,6}. Previous work in our laboratory⁷, and chapter 4, have shown that irradiation at gestational day 15 in a mouse model causes a reduced growth rate at the highest dose (1000 mGy) but had little effect on the respiratory system. These results came from a healthy animal model; however, radiation also has immunomodulatory properties, which may affect lung responses during inflammatory injury, resolution and repair.

Acute pathogenic lung injury occurs when a foreign stimulus, such as lipopolysaccharide (LPS), activates inflammation in the lung. Although there are several methods of lung injury, none accurately model human acute lung injury, however, LPS is an effective stimulus to observe the acute inflammation phase of acute lung injury. LPS, found on gram-negative bacteria, bind to toll-like receptor 4 (TLR4) found on alveolar macrophages and epithelial cells^{8–11}. This starts a cascade of events to trigger increased pro-inflammatory mediator release, increased leukocyte infiltration into alveolar spaces, increased epithelial cell damage, increased alveolar-capillary permeability and pulmonary oedema, resulting in decreased lung function^{11–13}. Most of these outcomes occur to enable neutralisation and clearance of the pathogen, eventually stimulating resolution and repair of the injury.

Many *in vitro* studies investigating the immunomodulatory response of ionizing radiation focus on monocyte or macrophage responses using both primary and immortalised cell lines in an unstimulated setting. Exposures to >1000 mGy, result in increased release of pro-inflammatory cytokines, for example, IL-1 β , which can also be seen *in vivo*^{14–17}. Additionally, high doses can cause increase leukocyte adhesion to endothelial cells, increasing leukocyte infiltration and promoting inflammation^{18,19}. In contrast, doses below 1000 mGy stimulate an anti-inflammatory environment through decreased proinflammatory and increased anti-inflammatory mediator release^{14,20–23}. These doses

135

reduce leukocyte migration through downregulation of adhesion molecules, reducing leukocyte infiltration and slowing the inflammation process^{24–28}.

Most studies that explore the immunomodulatory response of radiation during inflammation/stimulation do so using *in vitro and ex vivo* models. Exposure of LPS stimulated *in vitro* and *ex vivo* mouse macrophages to 500 mGy showed anti-inflammatory responses, including reduced TNF- α and IL-1 β levels and increased TGF- β levels^{20,21,23}. Contrarily, IL-1 β levels were increased from IFN- γ stimulation in similar conditions with a similar cell type²⁹. However, whether monocultured or co-cultured with endothelial cells, 100 – 500 mGy exposure with LPS, IL-1 β or TNF- α stimulated mononuclear cells show decreased adhesion and migration, which can be replicated *in vivo*^{23–25,30}. Thus, although there can be stimulus-dependent outcomes for cytokine profiles, the overall immunological effect of low dose ionizing radiation on migration, infiltration and inflammation appears similar, irrespective of stimuli.

While these models can loosely relate to respiratory infection, as macrophages and infiltrating leukocytes have major roles in stimulating and propagating inflammation within the lung, there is limited data specifically looking at the effects of ionizing radiation on an acute lung injury model and furthermore, the long-term effects from *in utero* exposures in a whole animal system. This study aims to address this missing research by exploring if *in utero* exposure to ionizing radiation influences the response to an acute respiratory immune stimulus in adolescent BALB/C mice.

136

5.2 METHODS

5.2.1 Experimental animals and irradiations

Male and female BALB/cJ mice (Jackson Laboratories, USA) were housed in a 12:12 hour light:dark cycle and allowed food and water *ab libitum*. The mothers were maintained on standard mouse chow (9% fat, 44.9% carbohydrate, 19% protein, Teklad Diets Envigo, USA) for the duration of pregnancy and pups received a similar chow post-weaning (6.2% fat, 44.2% carbohydrate, 18% protein, Teklad Diets Envigo, USA) for the duration of the study.

Female mice were paired 2:1 with male mice once overnight to obtain an accurate gestational day 0. The following morning, the females were checked for vaginal plugs, an indication for sexual activity, and then housed singly. Confirmation of pregnancy was obtained by abdominal palpitations later in gestation. At gestational day 15, pregnant BALB/c mice were transported across campus to McMaster University's Taylor Radiobiology Source, acclimated for *in situ* for 20 mins and irradiated (sham (irradiation control), 10 mGy, 100 mGy, 1000 mGy) using a ¹³⁷Cs Gamma radiation source (620 Ci, 662 keV energy, 10 mGy/min). All animals were restricted from food and water during irradiations. A transport control group (naïve) was added to determine the possible transport-related stress effect.

Pups were weaned at 3-4 weeks of age and up to 3 male and 4 female pups were housed together. To maximise group sizes and control maternal effect, up to 2 male and 2 female pups from any one mother were used in each challenge group (saline or LPS) for each

radiation group. Postweaning weight measurements were taken weekly until euthanasia at 16-17 weeks old.

All described animal procedures were reviewed and approved by the Animal Research Ethics Board at McMaster University (AUP #15-11-26) in line with the requirements of the *Canadian Council on Animal Care Guidelines*.

5.2.2 LPS Model

At 16-17 weeks postnatal age, mice were anaesthetised with 5% isoflurane and 3mg/Kg of lipopolysaccharide (LPS, *Escherichia coli* O55:B5, Sigma-Aldrich, USA) in 50 μL of 0.9% saline was instilled intratracheally followed by an air bolus (0.15 mL). Mice were allowed to recover and monitored for 24 hours until euthanasia.

5.2.3 Surgery and tissue collection

Animals were anaesthetised by 5% isoflurane and a tracheotomy then thoracotomy was performed. Maximal blood volume was collected via a cardiac puncture into lithium heparin blood collection tubes, following which the heart and lungs were excised. The blood was spun at 500g and the plasma was collected and stored at -80°C. The left lung lobe was resected and freeze-dried for lung lobe wet:dry weight analysis. The remaining lobes were degassed (0.5 atm for 60s) and lavaged with 3x 16mLs/Kg volume aliquots of cold 0.9% saline. Lobes were then resected and snap-frozen in liquid nitrogen. Bronchoalveolar lavage was spun at 500g to pellet cells and the supernatant was collected for protein analysis. Cells in the lavage were counted and stained to assess cellular infiltrate and different cell populations with BAL. Cells were stained using trypan blue and counted on a haemocytometer to calculate cellular infiltrate. Remaining cells were fixed in 4% paraformaldehyde, smeared then stained with Hematoxylin and Eosin and photographed under a BX brightfield microscope (Olympus, Japan) to assess differences in cell populations.

5.2.4 Immunoassays

Bronchoalveolar lavage supernatant was assessed for total protein concentration using the Micro BCA protein assay kit (Thermo Fisher Scientific, USA). Transforming growth factorbeta (TGF- β) was assessed in lavage supernatant and plasma by enzyme-linked immunosorbent assay (Rat TGF- β Duoset ELISA, RnD system, USA) as per manufactures instructions. The ELISA had a lower detection limit of 32pg/mL.

5.2.5 Tissue homogenisation and western blot

The snap-frozen, lower-right, lung lobe was homogenised in Pathscan buffer (25mM Tris, 150mM NaCl, 1mM EDTA and 1% Triton-X) containing sodium fluoride (20mM), sodium orthovanadate (2mM), and protease inhibitor cocktail (Sigma Aldrich, USA) using a Bead Ruptor 4 (Omni International, USA). Samples were then centrifuged at 90,000g for 25 minutes (Optima-Max TL Ultracentrifuge with TLA-55 rotor, Beckman Coulter, USA) and supernatant was collected and stored at -80°C. Protein content was quantified by EZQ protein assay as per the manufacturer's instructions (Thermo Fisher Scientific, USA). Next, 50µg of total protein from each sample was boiled and then 4X loading buffer

(+Bromophenol Blue) with 4% fresh dithiothreitol (DTT, Sigma-Aldridge, USA) was added. The samples were separated on 4-20% Criterion TGX, stain-free, precast, polyacrylamide gel (Bio-Rad, USA) in a Criterion cell (Bio-Rad, USA) using an electrophoresis power supply (EPS 1001, Amersham Pharmacia Biotech, UK) set to 300 volts for 20 minutes. The gel was transferred to Immuno-Blot low fluorescence PVDF (Bio-Rad, USA) using a Trans-Blot Turbo transfers system (Bio-Rad, USA) and imaged on a Gel Doc EZ imager (Bio-Rad, USA).

The membranes were blocked with 5% skim milk in PBS-T (Phosphate buffered saline + 0.1% tween 20) for 1 hour at room temperature, and then incubated with appropriate primary antibody overnight at 4°C. Following washing in PBS-T, the membranes were incubated with the appropriate horseradish peroxidase (HRP)-conjugated secondary antibody for 1 hour at room temperature. After washing again, the protein levels were visualised with enhanced chemiluminescence reagent and imaged on a ChemiDoc Touch imaging system (Bio-Rad, USA). Protein bands were analysed using Bio-Rad Image Lab software (Bio-Rad, USA) and were normalised to total protein concentration. Primary antibodies included: rabbit anti-glutathione peroxidase 1 (GPx 1/2000 dilution, [ab22604]), rabbit anti-superoxide dismutase 2 (SOD2 1/2000 dilution, [ab68155]), and rabbit anticatalase (1/2000 dilution, [ab209211]) all purchased from Abcam, UK. The secondary antibody was a Donkey anti-rabbit IgG conjugated to HRP ([715-035-152], Jackson Laboratories, USA). All antibody dilutions were made in 2.5% skim milk PBS-T.

5.2.6 Statistical Analysis

Sample size for the primary outcome of bronchoalveolar lavage total protein concentration was calculated to be n = 7 per group, using previous data ($\alpha = 0.05$, 1- $\beta = 0.8$, effect size = 0.641).

Longitudinal data (weight) is presented as a mean weekly measurement and was analysed using a generalised additive model fitted (GAM) to a generalised linear mixed model (GLMM) accounting for nesting of pups within mothers. Cross-sectional data (respiratory outcomes) are presented as mean and 95% confidence interval (95% C.I.), and were analysed using GLMM, accounting for nesting of pups within mothers. A *p*-value < 0.05 was considered statistically significant.

5.3 RESULTS

5.3.1 Transport effect

There was no difference between naïve transport controls and sham irradiation controls for the number of living males (p = 0.711), living females (p = 0.357) or total living pups (p = 0.685) indicating no transport effect on litter size (Figure 5-1). Similarly, there was no difference in weight at 4 wks or growth trajectory over the 12 weeks measured between naïve mice and sham irradiated mice indicating no transport effect on growth rate (Figure 5-2). There was also no difference in respiratory physiological outcomes of lung lobe wet:dry weight ratio, p = 0.773, or BAL total protein concentration, p = 0.709 (Figure 5-3); and respiratory immunological outcomes of BAL cell count, p = 0.828, or cell populations within BAL (Figure 5-4). From these findings, it appears there was no significant transport effect and so naïve animals were not included in the subsequent statistical assessments between groups.

5.3.2 Litter size and growth rate

There was no difference in number of living males, living females or total living pups among sham and irradiated groups, p = 0.484, p = 0.128, and p = 0.078, respectively. Mice exposed to 1000 mGy *in utero* weighed significantly less than sham control at 4 wks of age, males = 2.3 g difference (95% C.I. = 0.5 - 3.9 g.) and females = 1.7 g difference (95% C.I. = 0.1 - 3.4 g). This lighter weight continued throughout the 12 weeks of measurements with the overall growth trajectory of mice exposed to both 100 mGy and 1000 mGy significantly lower compared to the sham control, p = 0.002 and p < 0.001, respectively (Figure 5-2). Specifically, for the 1000 mGy exposed mice, there was a 15% reduction in weight for the males by 16 weeks of age and 25% reduction for the females.



Figure 5-1: Litter size of irradiated mothers. There was no significant difference in number of living male pups, living female pups, or total living pups between Naïve transport control and Sham irradiation control groups (p = 0.711, p = 0.357, p = 0.685 respectively), as tested by independent *t* tests, n = 9-10. There was also no difference among control and irradiated groups (p = 0.484, p = 0.128, and p = 0.078 respectively), as tested by one-way ANOVA n = 7-11. Bars and error bars represent mean and 95% confidence interval of the mean.



Figure 5-2: longitudinal weight measurements. *In utero* exposure to 100 mGy and 1000 mGy significantly alters the trajectory of growth compared to sham control (p = 0.0024 and p < 0.001). Exposure to 10 mGy of radiation was not different to sham. Sexes differed in growth curves (p<0.001). Graphs represent mean weekly measurements, n = 7-9. * indicates statistical significance of p < 0.05.



Figure 5-3: Physiological outcomes of the respiratory system, **A**) Lung lobe wet:dry weight ratio and **B**) Total BAL protein concentration. Naïve transport control animals were not significantly different to sham irradiation control animals indicating no effect of transport on these physiological outcomes **A**) LPS significantly increased the wet:dry weight ratio of the lung lobe, from 2.29 - 2.78 for females and 2.49 - 3.00 for males (p < 0.001, n = 7 - 12). In utero exposure to ionizing radiation did not significantly change the wet:dry ratio for saline or LPS stimulated animals back to the sham control for either sex, however, there was a significant difference for Female saline animals between 100 mGy and 1000 mGy, 2.60 - 1.97 (p = 0.038). **B**) LPS significantly increased the total BAL protein concentration, from $258.2 - 341.4 \mu g/mL$ for females and $234.3 - 394.1 \mu g/mL$ for males (p < 0.001, n = 5 - 7). In utero exposure to ionizing radiation did not significantly change the concentration of total BAL protein for saline or LPS stimulated animals back to sham control for either sex. Data represented as mean and 95% confidence intervals. Statistical significance was tested by GLMM with *post hoc* least significant difference (LSD). * denotes statistical significance between groups p < 0.05.



Figure 5-4: Immunological cell outcomes of the respiratory system, A) BAL cell count and B) Cell populations within BAL. Naïve transport control animals were not significantly different to sham irradiation control animals indicating no effect of transport on these immunological outcomes A) LPS significantly increased the BAL cell count, from $6.2x10^4 - 6.2x10^5$ cells/mL for females and $5.7x10^4 -$ 7.4×10^5 cells/mL for males (p < 0.001, n = 6 - 11). In utero exposure to ionizing radiation did not significantly change the cell counts for LPS stimulated female animals back to the sham control, but for saline animals 100 mGy was significantly higher compared to sham and 10 mGy groups, p = 0.007 and 0.002 respectively. Animals exposed to 10mGy were also significantly higher to 1000 mGy animals (p =0.017). In males, in utero exposure to ionizing radiation did not significantly alter any group back to sham control, but, 10 mGy was significantly less than 100 mGy for both saline (p = 0.025) and LPS stimulated (p = 0.027) mice. B) LPS significantly increased the percent of neutrophils (orange) seen in BAL cells, from 0.82% - 83.0% for females and 0.37% - 87.2% for Males (p < 0.001, n = 3 - 7). In utero exposure to ionizing radiation did not significantly change the any percentage of cells between any radiation group in saline or LPS stimulated animals of either sex. Data represented as mean and 95% confidence intervals. Statistical significance was tested by GLMM with post hoc least significant difference (LSD). * denotes statistical significance between groups p < 0.05.

5.3.3 Respiratory physiology

Instillation of LPS caused oedema within the lung and disrupted the alveolarcapillary barrier, as indicated by significantly increased lung lobe wet:dry weight ratio and BAL total protein compared to saline control, p < 0.001 (Figure 5-3). For the lung lobe wet:dry weight ratio, both females and males increased 21% and for BAL protein concentration, females increased 36% and males 68%. For both saline and LPS stimulated mice, *in utero* exposure to ionizing radiation did not significantly alter the lung lobe wet:dry weight ratio or the BAL total protein concentration compared to sham irradiated animals for either sex. There was only one inter dose difference noted, which was between salinetreated females exposed to 100 mGy and 1000 mGy, p = 0.038 (Figure 5-3A.1)

5.3.4 Respiratory immunology

Instillation of LPS significantly increased BAL cell count (p < 0.001, Figure 5-4) and altered the cell populations collected in BAL. Total cell count after LPS instillation increased 1000% for females and 1300% for males. For females, *in utero* exposure to ionizing radiation did not alter cell count in LPS stimulated mice but did for saline mice, where there was a significant difference between sham and 100 mGy animals, but not for 10 or 1000 mGy. There were also some significant differences noted among doses of radiation, as seen in Figure 5-4A.1, but not in a discernible trend. Irradiated male mice did not differ in cell counts compared to sham irradiated control, for either saline or LPS stimulated mice, but 10 mGy was significantly different to 100 mGy for both instillation groups, p = 0.025 and p= 0.027 respectively (Figure 5-4A.2). For both females and males, LPS instillation caused a significant increase in the percent of neutrophils within BAL cell count, 10121% and 23567%, respectively. *In utero* exposed female and male mice did not differ in the population of macrophages, lymphocytes or neutrophils within the BAL for either instillation groups.

Instillation of LPS significantly increased the amount of active TGF- β in BAL by 30%, p = 0.03 (Figure 5-5A). Irrespective of sex or stimulus, animals exposed to 100 mGy *in utero* had 45%-65% higher TGF- β concentrations than the other groups, p = 0.018. When considering sex and stimulus, the difference is not retained for females and only occurs in LPS males, but the difference is now 141%-202%, p < 0.05. Instillation of LPS significantly decreased the amount of Latent TGF- β levels in plasma by 45%, p = 0.002. There were no differences in latent plasma TGF- β between LPS males, saline males or LPS females, but there was for saline females (Figure 5-5). Both 10 mGy and 1000 mGy exposed animals were statistically higher than sham and 100 mGy irradiated animals, p < 0.05.



Figure 5-5: Soluble TGF- β concentrations by ELISA in BAL and Plasma as **A**) Active TGF- β in BAL supernatant and **B**) latent TGF- β in plasma samples. Active TGF- β concentrations was assessed in the BAL supernatant and latent TGF- β concentrations was assessed in plasma samples by acid activation of sample. **A**) LPS significantly increased TGF- β concentrations in BAL supernatants of both sexes, p = 0.03, n = 4-5. Irrespective of sex or stimulus, *in utero* exposure to 100 mGy resulted in a higher TGF- β concentrations than the other groups, p = 0.018. However, with *post hoc* analysis, female irradiated groups, for both saline and LPS stimulated animals, were not statistically different: only LPS males exposed to 100 mGy were different to the rest, p < 0.05. **B**) Both females and males exposed to LPS had significantly less TGF- β concentrations than saline animals, p = 0.002, n = 4-5. There was a significant interaction between dose and sex leading to opposite trends with increasing dose, p = 0.035. In *post hoc* analysis, 10 mGy and 1000 mGy saline animals had statistically higher TGF- β concentrations in the plasma compared to sham and 100 mGy irradiated animals, p < 0.05. Data represented as mean and 95% confidence intervals. Statistical significance was tested by GLMM with *post hoc* least significant difference (LSD). BAL samples were analysed using untransformed data and plasma samples were analysed using log transformed data and plasma samples were analysed using log transformed data and plasma samples were analysed using log transformed groups p < 0.05.

5.3.5 Respiratory antioxidant levels

Levels of three antioxidant enzymes, catalase, SOD2, and GPx, were assessed using western blot techniques, (Figure 5-6 and Figure 5-7). A protein curve confirmed that 50µg/well of total protein was sufficient to obtain protein bands without antibody saturation (Figure 5-8). For both sexes, intratracheal instillation of LPS significantly decreased catalase levels by 28% but increased SOD2 levels by 64% and GPx levels by 27%, p = 0.038, p < 0.001 and p = 0.047, respectively. SOD2 levels were 75% higher in female than males, p < 0.001, and there was a significant interaction between dose and sex, with increasing foetal radiation dose leading to opposite trends in protein levels, p = 0.01. There were no statistical differences between sexes or among irradiation groups for catalase and GPx



Figure 5-6: Western blot membrane and gel bands for antioxidant proteins catalase, SOD2, and GPx. Band intensities on membrane were normalised back to respective intensities from total protein bands on gel. 1 = Male Saline sham, 2 = Male Saline 10 mGy, 3 = Male Saline 100 mGy, 4 = Male Saline 1000 mGy, 5 = Male LPS sham, 6 = Male LPS 10 mGy, 7 = Male LPS 100 mGy, 8 = Male LPS 1000 mGy9 = Female Saline sham, 10 = Female Saline 10 mGy, 11 = Female Saline 100 mGy, 12 = Female Saline 1000 mGy, 13 = Female LPS sham, 14 = Female LPS 10 mGy, 15 = Female LPS 100 mGy, 16 = Female LPS 1000 mGy.



Figure 5-7: Western blot intensities of antioxidant proteins **A**) Catalase, **B**) SOD2, and **C**) GPx. **A**) LPS significantly lowered Catalase protein levels but *in utero* exposure to ionizing radiation had no effect for either sex (p = 0.038, n = 3-5). **B**) LPS significantly increased SOD2 protein levels but *in utero* exposure to ionizing radiation had no effect for either sex (p < 0.001, n = 4-5). There was a statistical difference between sex (p < 0.001) and an interaction between dose and sex leading to opposite trends when increasing dose (p = 0.002). **C**) LPS significantly lowered GPx protein levels but *in utero* exposure to ionizing radiation had no effect for either sex (p = 0.047, n = 3-4). Data represented as mean and 95% confidence intervals. Data was log transformed for statistical analysis by GLMM and antilog transformed for graphical representation. SOD2 = Superoxide dismutase 2, GPx = Glutathione peroxidase 1.











Figure 5-8: Saturation curves of antioxidant proteins A) Catalase, B) SOD2, and C) GPx proteins. All concentrations of total protein appear to be increasing and not hitting the asymptote; thus, saturation of antibody has not occurred. There were 4 increasing concentrations of total homogenised protein (n = 1): 10, 25, 50, and 75 ug/well. SOD2 = Superoxide dismutase 2, GPx = Glutathione peroxidase 1.
5.4 DISCUSSION

This study is the first to look at the long-term immunomodulatory effects of radiation on these respiratory outcomes in an *in utero* exposed setting. Therefore, there is a paucity of data with which to compare and relate the separate and combined immunomodulatory effects of radiation and LPS. The intention behind this study was to understand if radiation during pregnancy, at a time that had minimal possibility of foetal resorption or severe physical malformations, had any long-term effects on physiology or immunology during a respiratory bacterial infection. To test this aim, three doses were used in this study relating to different plausible clinical exposures. The lowest exposure, 10 mGy, represents the dose of a typical abdominal and pelvic CT scan³¹. Although not routinely received by pregnant patients, as evident in Chapter 3, clinical justification can result in its use e.g. in trauma, such as motor vehicle accidents, or delivery complication, such as placenta accreta. Although almost all diagnostic procedures fall below the middle exposure of 100 mGy, there are limited instances in which the foetus is exposed to a single procedure above this level³², nonetheless, this dose represents an unlikely, but possible, cumulative exposure to the foetus. The last exposure of 1000 mGy is well above diagnostic levels, and although it falls below therapeutic level of medical radiation, it was selected as the high dose control for this study.

In terms of translation of this animal model to a clinical setting, the gestational development at exposure and radiosensitivities between BALB/c mice and humans are very

similar. The gestational day 15 in mice (GD15) falls around the time of the pseudoglandular phase⁵. This phase in development relates to 5 - 17 weeks gestation in humans, with transcriptomics narrowing GD15 to approximately 12 – 15 weeks in human gestation³³. GD15, and 15 weeks in humans, are both on the late end of the pseudoglandular, early canicular phase of development, where the stereotypic lung structure is finalising development complete with intertwining airways and vessel branching. In terms of the overall radiosensitivity, BALB/c mice are also very comparable to humans. The human LD50 in 30 days from full body exposure is between 2000 – 8000 mGy, 4500 mGy for a healthy adult, and for BALB/c mice, is about 4500 mGy^{34–36}. Therefore, the radiobiological effects seen from irradiation of these mice at GD15 with these doses should be, at least partially, translatable to human exposures during early 2nd trimester of pregnancy.

In this study, there were significant growth restrictions from weeks 4 - 16 in mice exposed to 100 and 1000 mGy *in utero*. Although the result that a high dose of 1000 mGy can cause a persistently reduced growth rate among mice was expected, as previously seen in chapter 4, the degree of growth restriction in this study was much greater^{7,37–41}. Exposures of 1000 - 1500 mGy, at similar gestational ages, previously resulted in 11-15% reduction in body weight in mice and rats but this mouse study showed 15% reduction for males and 25% for females^{7,37,38}. This larger difference in weight may be due to species and strain variation, with varying radiosensitivities, or sex differences, where due to combining sexes rather than examining individual sexes despite inherently different weights and growth curves^{7,35–} ³⁸. As reviewed by both Alur and Clifton^{42,43}, the sex-based differences of the placenta and foetus allow for differential coping mechanisms to adverse environmental stimuli via protein, micro-RNA and gene expression variations. Their findings indicate that females conserved more energy, thereby reducing growth, as a coping mechanism to possibly prepare against further insult.

A surprising result from the growth outcomes was the reduction in body weight from 100 mGy exposed animals. To the author's knowledge, this is the first report of persistent growth restriction at this dose; an 8% reduction for males and 10% for females. Although it is statistically significant, the biological difference may be less significant. Even though this study was not able to assess weight preweaning, and therefore cannot state this phenomenon existed from birth, when extrapolated to human growth, a 10% reduction in body weight may not result in significant physiological effects, even from birth weight. For example, the average birth weight in Australia is 3.33kg and the average adult weight is 82 kg for males and 72 kg for females. A 10% reduction would not meet the criteria for low birth weight status, <2.5 kg or 25% reduction from the average birth weight, and the adults would actually be healthier in terms of average BMI^{44,45}. In comparison, the 25% reduction from the 1000 mGy exposure would result in a low birth weight status, increasing risk of complications and delayed postnatal growth, as well as push adult BMI closer to being underweight⁴⁶. Therefore, although statistically significant, the growth restriction by 100 mGy may not be biologically significant as in the 1000 mGy exposed animals.

LPS is a lipoglycan and endotoxin found on the outer membrane of gram-negative bacteria^{10,11}. It is a common, non-infectious, non-replicating, pro-inflammatory stimulus used in acute lung injury models to understand host inflammatory responses during bacterial infection¹³. Intratracheal instillation of LPS in this study increased both respiratory physiological outcomes of total BAL protein concentration and lung lobe wet:dry weight ratio, indicative of a successful acute lung injury model. Total BAL protein is an indicator for decreased alveolar-capillary barrier integrity, and therefore vascular leak, and is an expected finding following LPS instillation⁴⁷. Lung wet:dry weight ratio, an indicator of pulmonary oedema, is also very common with LPS instillation in the lung^{48,49}. Even measured by other methods, the concept that LPS induces pulmonary oedema is well known⁵⁰. In both a clinical and animal setting, increased vascular permeability can cause non-cardiogenic pulmonary oedema characterised by the influx of protein-rich fluid into the interstitium and alveolar space leading to reduced pulmonary function and reduced gas exchange^{51–54}. Intratracheal instillation of similar and lower doses of LPS (0.05-4 mg/Kg) reduced pulmonary function, including lung compliance, elastance and airway resistance, compared to saline control animals^{55,56}. Even via a different delivery method (inhalation), LPS still altered lung function 24 hours post instillation⁴⁷. These outcomes of dysfunction of alveolar-capillary barrier, pulmonary oedema, and changes in lung function, are all aspects present in human acute lung injury, indicating that LPS instillation produces a similar physiological profile observed in clinical patients^{12,57}.

The mechanisms through which LPS causes physiological outcomes are heavily influenced by its ability to stimulate a pro-inflammatory response. Intratracheal instillation of LPS increased total BAL leukocyte cell count, which was dominated by neutrophilic infiltration. This is again well known, even for LPS concentrations much less than 3mg/Kg^{47,48,55,58}. Intratracheal instillation of LPS stimulates alveolar epithelial cells and macrophages to promote a pro-inflammatory milieu that mediates chemotactic, predominately neutrophilic, leukocyte infiltration. The activated and infiltrated neutrophils release additional pro-inflammatory mediators while attempting to neutralise the stimuli, which perpetuate inflammation and inevitably damage healthy host tissues.

TGF-β is produced by macrophages and fibroblasts in a latent protein form and circulates in the blood or is secreted to the extracellular matrix. Activation of TGF-β can occur via integrin-dependent cell-mediated cleavage, or integrin independent means such as acidification, ROS and proteases^{59–61}. In response to tissue damage, local TGF-β is activated in an integrin-mediated process, as was indicated by LPS instillation significantly increasing active TGF-β in the lung⁶². Once activated, TGF-β binds to TGF-β receptors on epithelial, endothelial, stromal cells and leukocytes to mediate anti-inflammatory/pro-fibrotic responses^{59,63}. Instillation of LPS also significantly reduced the amount of latent TGF-β in the plasma in this study, which may be due to transfer into the lung. Other studies of intratracheal instillation of LPS (0.1–0.5 mg/kg) demonstrated increased pro-inflammatory mediators in BAL, including IL-6 and TNF-α, 24h post instillation. However, these levels were lower than earlier time points, and in conjunction with an increase in TGF-β, this suggests that the inflammatory process could have been shifting from the recruitment phase to the resolution/repair phase^{55,56,58}.

As reviewed by Matès *et al*⁶⁴, reactive oxygen species, for example, hydroxyl radicals, can be created via several pathways, including directly by aerobic respiration or radiolysis of water and indirectly by LPS stimulation^{65,66}. SOD2 converts hydroxyl radicals created in the mitochondria to hydrogen peroxide (H₂O₂), which is also considered a reactive oxygen species. When entered into the cytosol, this H_2O_2 is further catalysed to oxygen and water by both GPx and catalase, thereby defusing the reactiveness. In this study, intratracheal instillation of 3 mg/Kg LPS decreased levels of catalase and increased levels of SOD2 and GPx, with a pronounced increase of SOD2 indicating a mechanistic preference for this antioxidant. This could be due to oxidative stress activating intracellular redox-sensitive transcription factors, such as Nrf2, which translocate to the nucleus to upregulate antioxidant enzyme transcription, strongly for SOD2, and weakly for catalase and GPx1^{67,68}. In a similar study, mice exposed to a lower LPS concentration (0.5 mg/kg), had comparable outcomes for SOD2 and catalase⁵⁵. SOD2 continually increased up to 24hrs post instillation but catalase was increased at 6hrs then decreased below control levels at 24hrs. The levels of catalase had an inverse relationship with lipid and protein oxidative damage levels, indicating that an increase in H₂O₂ by SOD, but insufficient removal by catalase or GPx, resulted in H_2O_2 mediated lipid and protein oxidation and damage. Therefore, along with the LPS mediated reactive oxygen species damage, SOD2-created H₂O₂ may additively be contributing to damage within the lung.

Similar to chapter 4, this study also found no significant effect of *in utero* exposure to ionizing radiation compared to sham irradiated animals for physiological outcomes in control animals, nor in the LPS stimulated animals. Previously, in healthy, adolescent mice exposed to 2000 mGy and above, lung lobe wet:body weight ratio increased 6 weeks postirradiation, and returned to baseline before 16 weeks post irradiation⁶⁹. This increase in damage associated oedema was likely due to leaky vasculature or damage to the alveolarcapillary barrier. Therefore, it is plausible that at 6 weeks there would have also been increased BAL total protein concentration in these animals, that would have been reabsorbed before 16 weeks post-irradiation. Thus, it is likely that the timeframe between irradiation and outcome for chapters 4 and 5 may have been too great to observe any overall radiation-related effects on LPS response in these outcomes. In this current study (Figure 5-3), there was one statistical difference in control female animals; a slight decrease in wet:dry weight from 100 to 1000 mGy exposed animals that were also trending in the same direction for LPS simulated animals. However, this difference may not be biologically different as such small changes in ratio have not been previously associated with altered lung function.

Radiation did not have a pronounced effect on the respiratory immunological outcomes examined. *In utero* exposure to ionizing radiation did not significantly alter the amount of cellular infiltrate during inflammation. In saline control females, 100 mGy exposed animals had significantly more BAL cells than all other irradiated groups including the sham irradiated animals. Although statistically higher, the variation in groups resulted in a count of only 45 cells between the lower confidence interval of the 100 mGy females and the upper interval of the sham irradiation control, a technically unmeasurable amount. From Figure 5-4B, it appears that the increased cell count could consequentially be from neutrophil influx. However, this influx is not comparable to the influx of total cells and neutrophils from LPS instillation, and so the effect of these infiltrated cells in the saline 100 mGy mice would presumably be subtle. This statement should be confirmed by assessment of the level of activation of these cells extra infiltrated cells. Males exposed to 100 mGy were also statistically different to 10 mGy for both saline and LPS stimulated animals. The variation between these two doses is unusual, especially for LPS stimulated animals, as both 10 and 100 mGy falls under the anti-inflammatory range of ionizing radiation's effect^{24,25,30}. Overall, the variations in LPS stimulated animals, both statistical and trend-like, should be replicated to confirm a true finding and begin exploration as to why this phenomenon occurs.

There were no obvious trends in the type of cells infiltrating into the lung due to ionizing radiation. As previously reported, it was predominately macrophages for control animals and neutrophils following LPS stimulation, but radiation exposure did not alter this. Changes in cell subset populations might not have been visible due to macrophage or neutrophil cells dominating the population percentages. Additionally, it is also possible that small changes in subset populations were unable to be seen with the limited methodology used in this study. Cell differentials by hematoxylin and eosin staining lacks the ability to correctly identify subpopulations with physiologically similar cell types. Flow cytometry might have been able to distinguish between subpopulations of cells that ionizing radiation commonly affect, such as T lymphocytes and macrophages. It is possible that the immunomodulatory effects of radiation were not strong enough to cause significant changes in overall populations 16 weeks post-irradiation. Most work that focuses on the immunomodulatory effects of ionizing radiation use macrophages, peripheral mononuclear cells or T lymphocytes, generally investigating short term outcomes^{22–25,70,71}. One mouse study found effects the 12 weeks later but in a specialised T lymphocyte subset (T regulatory cells) stimulated by very high dose exposure (15 Gy)⁷¹. Additionally, there is a scarcity of evidence on the radio-immunomodulatory effects on neutrophil responses, however, attachment and migration of these cells from the vasculature to the site of infection use similar adhesion molecules that are known to have a down-regulated response to low dose radiation^{30,72}. Thus, although there were no significant changes in cell number or cell population, likely due to length of time post-irradiation, improving the methodological technology could confirm that there was truly no change in populations.

TGF- β concentration was affected by *in utero* exposure to ionizing radiation but not in a dose-dependent pattern. In BAL supernatant, active TGF- β was unchanged in females and saline males, but BAL from LPS stimulated, 100 mGy irradiated males was significantly greater than the rest. In short term studies, the current understanding is that low doses (<1000 mGy) of ionizing radiation activate TGF- β , through both integrin meditated and non-mediated pathways, and that increased activation of TGF- β results in an anti-inflammatory response, at least in part, though TGF- β - mediated decreased leukocyte

migration^{27,28,30,73–75}. However, this is not the case in our results, where there is a trend of increasing cellular infiltrate for 100 mGy LPS stimulated animals. In longer-term experiments, up to 16 weeks post-irradiation, doses much higher than ours are associated with increased active TGF- β , and resulting fibrosis, but there is a lack of evidence associating low doses with similar effects at this timepoint^{76,77}. Additionally, if this were to be true and did extrapolate down to lower doses, it would be expected that both 100 and 1000 mGy exposed animals, regardless of stimuli, would be higher, rather than just 100 mGy LPS stimulated animals. Therefore, this result is a surprising anomaly in which further investigation is required to understand this outcome and its relation to LPS induced pathophysiological changes.

In latent TGF- β plasma concentrations, it was saline females that showed variations due to *in utero* irradiation. Animals exposed to 10 mGy and 1000 mGy had significantly higher latent TGF- β levels compared to both sham and 100 mGy irradiated animals. This biphasic shape itself is not unusual for TGF- β , as similar shapes can be seen for active TGF- β at similar exposures, but in a much shorter timeframe⁷⁸. Additionally, as the literature focuses on active TGF- β , there is limited data on the effects of irradiation on latent TGF- β . These results would need to be repeated with a small pilot group of saline instilled females to confirm these results. The difference between irradiated animals in the saline females was not visible in the LPS animals, suggesting that LPS stimulation activated all of the extra latent TGF- β , caused by *in utero* irradiation, to a similar level to sham and 100 mGy animals. However, when incorporating Figure 5-5A.1, the activated LPS did not accumulate in the lungs. Soluble latent TGF-β has a longer half-life in the plasma (> 100 minutes) than active TGF-β (2 – 3 minutes), which is readily taken up by tissues⁶³. This shorter half-life of active TGF-β, the non-significant finding in active TGF-β in the lungs of females, and the large difference in units between latent and active TGF-β levels, could suggest that the TGF-β response had occurred within the lung before the 24 hours measurement. This hypothesis would need to be confirmed by exploring a time-response curve.

Finally, antioxidant levels in the lung appeared unchanged in 16-week-old mice following in utero ionizing radiation exposure. There is large variation between groups for all three antioxidants examined in the lung tissue. Increasing sample size may reduce this variation and allow more accurate determination of potential differences. Overall, radiation did not significantly alter the response to LPS for any antioxidant measured. However, when removing grouping by stimulus factor (saline or LPS), the concentration of SOD2 was influenced by radiation based on sex. This sex:radiation-dose interaction led to directionally different outcomes, with SOD2 concentrations increasing for females and decreasing for males with increasing dose. The sham irradiated levels between sexes are very similar, so it appears that there is a sex-associated difference that is triggered by radiation that is causing this variation in direction. There is a lack of evidence supporting sex differences in SOD2 expression following ionizing radiation, so it is difficult to propose any causation. In in vitro models, doses < 1000 mGy have shown increased SOD2 expression, via a Nrf1/2 pathway, but only for short periods of up to 24 hours⁷⁹. Doses > 1000 mGy have shown increased concentrations of SOD2 and GPx protein, due to radiation-mediated chronic

mitochondrial stress, but only up to 1-month post irradiation⁸⁰. Although these studies are conducted post-embryonic development and even though this dose:sex interaction is present, the timing in our study post-irradiation may be too long to observe specific effects due to irradiation.

The design of this study presents several limitations. The use of LPS as the model stimulus comes with several advantages and disadvantages. Due to its non-replicating ability, the use of LPS allows examination of specific phases of the inflammation process, for example, recruitment, resolution or repair, based on time periods post instillation. However, in a clinical setting, infection occurs with live pathogens which perpetuates the inflammation processes continually replicating and stimulating the immune system until cleared¹³. Additionally, live bacteria also have other immune triggering stimuli including exotoxins and effector protein infiltrations through type III secretion systems^{81,82}. Finally, although LPS triggers an immune response that is similar to intratracheal exposure of LPS in humans, the model does not include all of the complex pathophysiological changes associated with clinical acute lung injury, or its more severe form, acute respiratory distress syndrome (ARDS)^{12,13}. Instillation of LPS differs in its ability to cause severe alveolar epithelial, and thus, it is limited in its ability to accurately replicate clinical situations^{13,83}.

Although longitudinal measurements were taken for the weight of these animals, the respiratory outcomes were only collected at a single time point. This limits the examination of the effects of *in utero* exposure to ionizing radiation on the developing respiratory

system. This time point was chosen to allow for the successful assessment of growth restriction, as a validated outcome to show irradiation affected the pups. In future studies, shortening the time frame between ionizing radiation and respiratory outcomes will prove useful in understanding the temporal immunomodulatory effects of low dose ionizing radiation.

Lastly, expanding the current outcomes to increase specificity will provide more information on the potential effect of radiation on the respiratory system. Incorporating flow cytometry would provide extensive additional information regarding the circulating and tissue infiltrating cell phenotypes. Expanding the cytokine selections to include proinflammatory and additional anti-inflammatory mediators will provide greater knowledge into the long-term cytokine profile of irradiated and stimulated respiratory systems. Additionally, examining deeper the relationships between outcomes and understanding the pathways would better help the interpretation of results, such as defining if there is a radiation-induced association between infiltrating cell type and mediators present in BAL. Finally, addition of lung function testing, via blood oxygenation or respiratory mechanics, will confirm if radiation has any effect on the respiratory function directly, or if it attenuates function during acute lung injury.

5.5 CONCLUSION

Overall, this study has demonstrated that clinically relevant exposures to ionizing radiation during late gestation have little to no effect on the long-term physiology and immunology of the mouse lung during inflammation from a bacterial stimulus. Addition and refinement of some techniques would allow for a greater understanding, but the current results provide a foundation of knowledge. There was some growth restriction occurring at both 100 and 1000 mGy exposures, but no effect was seen at the most clinically relevant dose of 10 mGy. These results support the hypothesis that low dose irradiation has little effect on the foetus during development, however, significant work is required to solidify this ideal.

5.6 REFERENCES

- 1. Weber EM, Algers B, Würbel H, Hultgren J, Olsson IAS. Influence of strain and parity on the risk of litter loss in laboratory mice. *Reprod Domest Anim*. 2013;48(2):292-296. doi:10.1111/j.1439-0531.2012.02147.x.
- 2. The International Commission on Radiological Protection. *Biological Effects after Prenatal Irradiation (Embryo and Fetus). ICRP Publication 90.* Vol 90.; 2003.
- 3. Rugh R, Wohlfromm M. Prenatal X-Irradiation and Postnatal Mortality. *Radiat Res.* 1965;26(4):493-506. doi:10.2307/3571860.
- 4. Russell L, Russel W. The effect of radiation on the preimplantation stages of the mouse embryo. *Anat Rec.* 1950;108.
- 5. Pinkerton KE, Joad JP. The mammalian respiratory system and critical windows of exposure for children's health. *Environ Health Perspect*. 2000;108(SUPPL. 3):457-462. doi:10.1289/ehp.00108s3457.
- 6. DeMello DE, Sawyer D, Galvin N, Reid LM. Early Fetal Development of Lung Vasculature. *Am J Respir Cell Mol Biol*. 1997;16(5):568-581. doi:10.1165/ajrcmb.16.5.9160839.
- Sreetharan S, Stoa L, Cybulski ME, Jones DE, Lee AH, Kulesza A V., Tharmalingam S, Boreham DR, Tai TC, Wilson JY. Cardiovascular and growth outcomes of C57Bl/6J mice offspring exposed to maternal stress and ionizing radiation during pregnancy. *Int J Radiat Biol.* March 2019:1-9. doi:10.1080/09553002.2019.1589025.
- Hoshino K, Takeuchi O, Kawai T, Sanjo H, Ogawa T, Takeda Y, Takeda K, Akira S. Cutting edge: Toll-like receptor 4 (TLR4)-deficient mice are hyporesponsive to lipopolysaccharide: evidence for TLR4 as the Lps gene product. J Immunol. 1999;162(7):3749-3752.
- 9. Armstrong L, Medford ARL, Uppington KM, Robertson J, Witherden IR, Tetley TD, Millar AB. Expression of functional toll-like receptor-2 and -4 on alveolar epithelial cells. *Am J Respir Cell Mol Biol*. 2004;31(2 I):241-245. doi:10.1165/rcmb.2004-0078OC.
- 10. Raetz CR, Ulevitch RJ, Wright SD, Sibley CH, Ding A, Nathan CF. Gram-negative endotoxin: an extraordinary lipid with profound effects on eukaryotic signal transduction. *FASEB J*. 1991;5(12):2652-2660. doi:10.1096/fasebj.5.12.1916089.
- 11. Beutler B, Rietschel ET. Innate immune sensing and its roots: The story of endotoxin. *Nat Rev Immunol*. 2003;3(2):169-176. doi:10.1038/nri1004.
- 12. O'Grady NP, Preas HL, Pugin J, Fiuza C, Tropea M, Reda D, Banks SM, Suffredini AF. Local inflammatory responses following bronchial endotoxin instillation in humans. *Am J Respir Crit Care Med*. 2001;163(7):1591-1598. doi:10.1164/ajrccm.163.7.2009111.
- 13. Matute-Bello G, Frevert CW, Martin TR. Animal models of acute lung injury. *Am J Physiol Cell Mol Physiol*. 2008;295(3):L379-L399. doi:10.1152/ajplung.00010.2008.
- 14. Lim S, Ko E-J, Kang Y-J, Baek K-W, Ock MS, Song KS, Kang H-J, Keum Y-S, Hyun JW, Kwon TK, Nam SY, Cha H-J, Choi YH. Effect of irradiation on cytokine secretion and nitric oxide production by inflammatory macrophages. *Genes Genomics*. 2016;38(8):717-722. doi:10.1007/s13258-016-0416-4.
- 15. O'Brien-Ladner A, Nelson ME, Kimler BF, Wesselius LJ. Release of Interleukin-1 by Human Alveolar Macrophages after In Vitro Irradiation. *Radiat Res.* 2006;136(1):37. doi:10.2307/3578637.
- 16. Johnston CJ, Hernady E, Reed C, Thurston SW, Finkelstein JN, Williams JP. Early Alterations in Cytokine Expression in Adult Compared to Developing Lung in Mice after Radiation Exposure. *Radiat Res.* 2010;173(4):522-535. doi:10.1667/rr1882.1.

- 17. Pinto AT, Pinto ML, Cardoso AP, Monteiro C, Pinto MT, Filipe Maia A, Castro P, Figueira R, Monteiro A, Marques M, Mareel M, Dos Santos SG, Seruca R, Barbosa MA, Rocha S, JoséOliveira M. Ionizing radiation modulates human macrophages towards a pro-inflammatory phenotype preserving their pro-invasive and pro-angiogenic capacities. *Sci Rep.* 2016;6(1):18765. doi:10.1038/srep18765.
- 18. Hallahan DE, Virudachalam S. Intercellular adhesion molecule 1 knockout abrogates radiation induced pulmonary inflammation. *Med Sci.* 1997;94:6432-6437.
- 19. Gaugler MH, Squiban C, van der Meeren A, Bertho JM, Vandamme M, Mouthon MA. Late and persistent upregulation of intercellular adhesion molecule-1 (ICAM-1) expression by ionizing radiation in human endothelial cells in vitro. *Int J Radiat Biol*. 1997;72(2):201-209.
- Frischholz B, Wunderlich R, Rühle P-F, Schorn C, Rödel F, Keilholz L, Fietkau R, Gaipl US, Frey B. Reduced secretion of the inflammatory cytokine IL-1β by stimulated peritoneal macrophages of radiosensitive Balb/c mice after exposure to 0.5 or 0.7Gy of ionizing radiation. *Autoimmunity*. 2013;46(5):323-328. doi:10.3109/08916934.2012.747522.
- Tsukimoto M, Homma T, Mutou Y, Kojima S. 0.5 Gy Gamma Radiation Suppresses Production of TNF-α through Up-regulation of MKP-1 in Mouse Macrophage RAW264.7 Cells. *Radiat Res.* 2009;171(2):219-224. doi:10.1667/RR1351.1.
- 22. Schröder S, Kriesen S, Paape D, Hildebrandt G, Manda K. Modulation of Inflammatory Reactions by Low-Dose Ionizing Radiation: Cytokine Release of Murine Endothelial Cells Is Dependent on Culture Conditions. *J Immunol Res.* 2018;2018:1-13. doi:10.1155/2018/2856518.
- 23. Wunderlich R, Ernst A, Rödel F, Fietkau R, Ott O, Lauber K, Frey B, Gaipl US. Low and moderate doses of ionizing radiation up to 2 Gy modulate transmigration and chemotaxis of activated macrophages, provoke an antiinflammatory cytokine milieu, but do not impact upon viability and phagocytic function. *Clin Exp Immunol*. 2015;179(1):50-61. doi:10.1111/cei.12344.
- 24. Kern PM, Keilholz L, Forster C, Hallmann R, Herrmann M, Seegenschmiedt M-H. Low-dose radiotherapy selectively reduces adhesion of peripheral blood mononuclear cells to endothelium in vitro. *Radiother Oncol.* 2000;54(3):273-282. doi:10.1016/S0167-8140(00)00141-9.
- 25. Hildebrandt G, Maggiorella L, Rödel F, Rödel V, Willis D, Trott K-R. Mononuclear cell adhesion and cell adhesion molecule liberation after X-irradiation of activated endothelial cells in vitro. *Int J Radiat Biol*. 2002;78(4):315-325. doi:10.1080/09553000110106027.
- 26. Roedel F, Kley N, Beuscher HU, Hildebrandt G, Keilholz L, Kern P, Voll R, Herrmann M, Sauer R. Anti-inflammatory effect of low-dose X-irradiation and the involvement of a TGF-β 1 -induced down-regulation of leukocyte/endothelial cell adhesion. *Int J Radiat Biol.* 2002;78(8):711-719. doi:10.1080/09553000210137671.
- 27. Rödel F, Schaller U, Schultze-Mosgau S, Beuscher HU, Keilholz L, Herrmann M, Voll R, Sauer R, Hildebrandt G. The Induction of TGF-β1and NF-κB Parallels a Biphasic Time Course of Leukocyte/Endothelial Cell Adhesion Following Low-Dose X-Irradiation. *Strahlentherapie und Onkol*. 2004;180(4):194-200. doi:10.1007/s00066-004-1237-y.
- Arenas M, Gil F, Gironella M, Hernández V, Biete A, Piqué JM, Panés J. Time course of anti-inflammatory effect of low-dose radiotherapy: Correlation with TGF-β1 expression. *Radiother Oncol.* 2008;86(3):399-406. doi:10.1016/j.radonc.2007.10.032.
- 29. Ibuki Y, Goto R. Contribution of Inflammatory Cytokine Release to Activation of Resident Peritoneal Macrophages after in vivo Low-dose γ-irradiation. *J Radiat Res.* 1999;40(3):253-262. doi:10.1269/jrr.40.253.
- Arenas M, Gil F, Gironella M, Hernández V, Jorcano S, Biete A, Piqué JM, Panés J. Anti-inflammatory effects of low-dose radiotherapy in an experimental model of systemic inflammation in mice. *Int J Radiat Oncol Biol Phys.* 2006;66(2):560-567. doi:10.1016/j.ijrobp.2006.06.004.
- 31. Mettler FA, Huda W, Yoshizumi TT, Mahesh M. Effective doses in radiology and diagnostic nuclear medicine: a

catalog. Radiology. 2008;248(1):254-263. doi:10.1148/radiol.2481071451.

- 32. Osei EK, Faulkner K. Fetal doses from radiological examinations. *Br J Radiol*. 1999;72(AUG.):773-780. doi:10.1259/bjr.72.860.10624343.
- 33. Maeda Y, Davé V, Whitsett JA. Transcriptional control of lung morphogenesis. *Physiol Rev.* 2007;87(1):219-244. doi:10.1152/physrev.00028.2006.
- 34. Mole RH. The LD50 for uniform low LET irradiation of man. *Br J Radiol*. 1984;57(677):355-369. doi:10.1259/0007-1285-57-677-355.
- 35. Grahn D, Hamilton KF. Genetic Variation in the Acute Lethal Response of Four Inbred Mouse Strains to Whole Body X-Irradiation. *Genetics*. 1957;42(3):189-198.
- 36. Kohn HI, Kallman RF. The Influence of Strain on Acute X-Ray Lethality in the Mouse: I. LD 50 and Death Rate Studies. *Radiat Res.* 1956;5(4):309. doi:10.2307/3570420.
- 37. Nash DJ, Gowen JW. Effects of X-irradiation upon postnatal growth in the mouse. *Biol Bull*. 1962;122(1):115-136. doi:10.2307/1539326.
- 38. Kimler BF, Norton S. Behavioral changes and structural defects in rats irradiated in utero. *Int J Radiat Oncol Biol Phys.* 1988;15(5):1171-1177. doi:10.1016/0360-3016(88)90200-3.
- 39. Murphree RL, Pace HB. The Effects of Prenatal Radiation on Postnatal Development in Rats. *Radiat Res.* 1960;12(5):495. doi:10.2307/3570971.
- 40. Jensh RP, Eisenman LM, Brent RL. Postnatal neurophysiologic effects of prenatal x-irradiation. *Int J Radiat Biol*. 1995;67(2):217-227. doi:10.1080/09553009514550271.
- 41. Hossain M, Uma Devi P, Bisht KS. Effect of prenatal gamma irradiation during the late fetal period on the postnatal development of the mouse. *Teratology*. 1999;59(3):133-138. doi:10.1002/(SICI)1096-9926(199903)59:3<133::AID-TERA4>3.0.CO;2-T.
- 42. Alur P. Sex Differences in Nutrition, Growth, and Metabolism in Preterm Infants. *Front Pediatr.* 2019;7:22. doi:10.3389/fped.2019.00022.
- 43. Clifton VL. Review: Sex and the Human Placenta: Mediating Differential Strategies of Fetal Growth and Survival. *Placenta*. 2010;31(SUPPL.):S33-S39. doi:10.1016/j.placenta.2009.11.010.
- 44. Australian Institute of Health and Welfare. *Australia's Health 2018, Report No 16*. Canberra; 2018.
- 45. Australian Bureau of Statistics. National Health Survey: first results, Australia 2014-2015 4364.0.55.001. 2018;(December).
- 46. Belbasis L, Savvidou MD, Kanu C, Evangelou E, Tzoulaki I. Birth weight in relation to health and disease in later life: An umbrella review of systematic reviews and meta-analyses. *BMC Med.* 2016;14(1):147. doi:10.1186/s12916-016-0692-5.
- 47. Faffe DS, Seidl VR, Chagas PSC, Gonc Ëalves De Moraes VL, Capelozzi VL, Rocco PRM, Zin WA. Respiratory effects of lipopolysaccharide-induced inflammatory lung injury in mice. *Eur Respir J*. 2000;15(1):85-91. doi:10.1034/j.1399-3003.2000.15a16.x.
- Takashima K, Matsushima M, Hashimoto K, Nose H, Sato M, Hashimoto N, Hasegawa Y, Kawabe T. Protective effects of intratracheally administered quercetin on lipopolysaccharide-induced acute lung injury. *Respir Res.* 2014;15(1):150. doi:10.1186/s12931-014-0150-x.
- 49. Pei X, Zhang XJ, Chen HM. Bardoxolone treatment alleviates lipopolysaccharide (LPS)-induced acute lung injury through suppressing inflammation and oxidative stress regulated by Nrf2 signaling. *Biochem Biophys Res Commun*. 2019;516(1):270-277. doi:10.1016/j.bbrc.2019.06.006.

- 50. Conti G, Tambalo S, Villetti G, Catinella S, Carnini C, Bassani F, Sonato N, Sbarbati A, Marzola P. Evaluation of lung inflammation induced by intratracheal administration of LPS in mice: Comparison between MRI and histology. *Magn Reson Mater Physics, Biol Med.* 2010;23(2):93-101. doi:10.1007/s10334-010-0203-1.
- 51. Ware LB, Matthay MA. Alveolar fluid clearance is impaired in the majority of patients with acute lung injury and the acute respiratory distress syndrome. *Am J Respir Crit Care Med.* 2001;163(6):1376-1383. doi:10.1164/ajrccm.163.6.2004035.
- 52. Ware LB, Matthay MA. The Acute Respiratory Distress Syndrome. *N Engl J Med*. 2000;342(18):1334-1349. doi:10.1056/NEJM200005043421806.
- 53. Matthay MA, Ware LB, Zimmerman GA. The acute respiratory distress syndrome. *J Clin Invest*. 2012;122(8):2731-2740. doi:10.1172/JCI60331.
- 54. Chen H, Wu S, Lu R, Zhang Y, Zheng Y, Sun J. Pulmonary permeability assessed by fluorescent-labeled dextran instilled intranasally into mice with LPS-induced acute lung injury. *PLoS One*. 2014;9(7):e101925. doi:10.1371/journal.pone.0101925.
- 55. Lima Trajano ET, Sternberg C, Caetano M, Santos Silva MA, Porto LC, Santos JC, Ribeiro ML, Magalhães CB, Zin WA, Benjamim CF, Valença SS. Endotoxin-induced acute lung injury is dependent upon oxidative response. *Inhal Toxicol.* 2011;23(14):918-926. doi:10.3109/08958378.2011.625994.
- 56. Verjans E, Kanzler S, Ohl K, Rieg AD, Ruske N, Schippers A, Wagner N, Tenbrock K, Uhlig S, Martin C. Initiation of LPS-induced pulmonary dysfunction and its recovery occur independent of T cells. *BMC Pulm Med.* 2018;18(1). doi:10.1186/s12890-018-0741-2.
- 57. Proudfoot AG, McAuley DF, Griffiths MJD, Hind M. Human models of acute lung injury. *Dis Model Mech*. 2011;4(2):145-153. doi:10.1242/dmm.006213.
- 58. Vernooy JHJ, Dentener MA, Van Suylen RJ, Buurman WA, Wouters EFM. Intratracheal instillation of lipopolysaccharide in mice induces apoptosis in bronchial epithelial cells: No role for tumor necrosis factor-α and infiltrating neutrophils. *Am J Respir Cell Mol Biol*. 2001;24(5):569-576. doi:10.1165/ajrcmb.24.5.4156.
- 59. Shi M, Zhu J, Wang R, Chen X, Mi L, Walz T, Springer TA. Latent TGF-β structure and activation. *Nature*. 2011;474(7351):343-351. doi:10.1038/nature10152.
- 60. Barcellos-Hoff MH, Dix TA. Redox-mediated activation of latent transforming growth factor-beta 1. *Mol Endocrinol*. 1996;10(9):1077-1083. doi:10.1210/mend.10.9.8885242.
- 61. Lyons RM, Keski-Oja J, Moses HL. Proteolytic activation of latent transforming growth factor-β from fibroblastconditioned medium. *J Cell Biol*. 1988;106(5):1659-1665. doi:10.1083/jcb.106.5.1659.
- 62. Pittet JF, Griffiths MJ, Geiser T, Kaminski N, Dalton SL, Huang X, Brown LA, Gotwals PJ, Koteliansky VE, Matthay MA, Sheppard D. TGF-beta is a critical mediator of acute lung injury. *J Clin Invest*. 2001;107(12):1537-1544. doi:10.1172/JCl11963.
- 63. Wakefield LM, Winokur TS, Hollands RS, Christopherson K, Levinson AD, Sporn MB. Recombinant latent transforming growth factor β1 has a longer plasma half-life in rats than active transforming growth factor β1, and a different tissue distribution. *J Clin Invest*. 1990;86(6):1976-1984. doi:10.1172/JCl114932.
- 64. Matés JM, Pérez-Gómez C, De Castro IN. Antioxidant enzymes and human diseases. *Clin Biochem*. 1999;32(8):595-603. doi:10.1016/S0009-9120(99)00075-2.
- 65. Kallapura G, Pumford NR, Hernandez-Velasco X, Hargis BM, Tellez G. Mechanisms Involved in Lipopolysaccharide Derived ROS and RNS Oxidative Stress and Septic Shock. *J Microbiol Res Rev.* 2014;2(1):6-11.
- 66. Hsu HY, Wen MH. Lipopolysaccharide-mediated reactive oxygen species and signal transduction in the regulation of interleukin-1 gene expression. *J Biol Chem*. 2002;277(25):22131-22139. doi:10.1074/jbc.M111883200.
- 67. Zhao H, Eguchi S, Alam A, Ma D. The role of nuclear factor-erythroid 2 related factor 2 (Nrf-2) in the protection

against lung injury. Am J Physiol Cell Mol Physiol. 2016;312(2):L155-L162. doi:10.1152/ajplung.00449.2016.

- 68. Kensler TW, Wakabayashi N, Biswal S. Cell Survival Responses to Environmental Stresses Via the Keap1-Nrf2-ARE Pathway. *Annu Rev Pharmacol Toxicol*. 2007;47(1):89-116. doi:10.1146/annurev.pharmtox.46.120604.141046.
- 69. Sharplin J, Franko AJ. Irradiation of mouse lungs causes a dose-dependent increase in lung weight. *Int J Radiat* Oncol Biol Phys. 1982;8(6):1065-1069. doi:10.1016/0360-3016(82)90179-1.
- 70. Park BS, Hong GU, Ro JY. Foxp3(+) -Treg Cells Enhanced by Repeated Low-Dose Gamma-Irradiation Attenuate Ovalbumin-Induced Allergic Asthma in Mice. *Radiat Res.* 2013;179(5):570-583. doi:10.1667/rr3082.1.
- 71. Wirsdörfer F, Cappuccini F, Niazman M, de Leve S, Westendorf AM, Lüdemann L, Stuschke M, Jendrossek V. Thorax irradiation triggers a local and systemic accumulation of immunosuppressive CD4+ FoxP3+ regulatory T cells. *Radiat Oncol.* 2014;9(1):98. doi:10.1186/1748-717X-9-98.
- 72. Hallahan D, Kuchibhotla J, Wyble C. Cell Adhesion Molecules Mediate Radiation-induced Leukocyte Adhesion to the Vascular Endothelium'. *Cancer*. 1996;56:5150-5155.
- 73. Ehrhart EJ, Segarini P, Tsang ML, Carroll AG, Barcellos-Hoff MH. Latent transforming growth factor beta1 activation in situ: quantitative and functional evidence after low-dose gamma-irradiation. *FASEB J*. 1997;11(12):991-1002. doi:10.1096/fasebj.11.12.9337152.
- 74. Jobling MF, Mott JD, Finnegan MT, Jurukovski V, Erickson AC, Walian PJ, Taylor SE, Ledbetter S, Lawrence CM, Rifkin DB, Barcellos-Hoff MH. Isoform-Specific Activation of Latent Transforming Growth Factor β (LTGF-β) by Reactive Oxygen Species. *Radiat Res.* 2006;166(6):839-848. doi:10.1667/rr0695.1.
- 75. Puthawala K, Hadjiangelis N, Jacoby SC, Bayongan E, Zhao Z, Yang Z, Devitt ML, Horan GS, Weinreb PH, Lukashev ME, Violette SM, Grant KS, Colarossi C, Formenti SC, Munger JS. Inhibition of integrin αvβ6, an activator of latent transforming growth factor-β, prevents radiation-induced lung fibrosis. *Am J Respir Crit Care Med*. 2008;177(1):82-90. doi:10.1164/rccm.200706-806OC.
- 76. Franko AJ, Sharplin J, Ghahary A, Barcellos-Hoff MH. Immunohistochemical Localization of Transforming Growth Factor β and Tumor Necrosis Factor α in the Lungs of Fibrosis-Prone and "Non-Fibrosing" Mice during the Latent Period and Early Phase after IrradiationImmunohistochemical Localization of Transforming. *Radiat Res.* 1997;147(2):245. doi:10.2307/3579426.
- 77. Rubin P, Johnston CJ, Williams JP, McDonald S, Finkelstein JN. A perpetual cascade of cytokines postirradiation leads to pulmonary fibrosis. *Int J Radiat Oncol Biol Phys.* 1995;33(1):99-109. doi:10.1016/0360-3016(95)00095-G.
- 78. Maslowski AJ. Low dose radiation interaction with transformation grwoth factor (TGF)-beta pathway: a dissertation. 2007.
- 79. Lee EK, Kim JKJA, Park SJ, Kim JKJA, Heo K, Yang KM, Son TG. Low-dose radiation activates Nrf1/2 through reactive species and the Ca2+/ERK1/2 signaling pathway in human skin fibroblast cells. *BMB Rep.* 2013;46(5):258-263. doi:10.5483/BMBRep.2013.46.5.199.
- 80. Lafargue A, Degorre C, Corre I, Alves-Guerra MC, Gaugler MH, Vallette F, Pecqueur C, Paris F. Ionizing radiation induces long-term senescence in endothelial cells through mitochondrial respiratory complex II dysfunction and superoxide generation. *Free Radic Biol Med*. 2017;108:750-759. doi:10.1016/j.freeradbiomed.2017.04.019.
- 81. Galle M, Carpentier I, Beyaert R. Structure and Function of the Type III Secretion System of Pseudomonas aeruginosa. *Curr Protein Pept Sci.* 2013;13(8):831-842. doi:10.2174/138920312804871210.
- 82. Kudoh I, Wiener-Kronish JP, Hashimoto S, Pittet JF, Frank D. Exoproduct secretions of Pseudomonas aeruginosa strains influence severity of alveolar epithelial injury. *Am J Physiol*. 1994;267(5 Pt 1):L551-6. doi:10.1152/ajplung.1994.267.5.L551.
- 83. Wiener-Kronish JP, Albertine KH, Matthay MA, Osorio O, Neuberger M. Differential responses of the endothelial and epithelial barriers of the lung in sheep to Escherichia coli endotoxin. *J Clin Invest*. 1991;88(3):864-875.

doi:10.1172/JCI115388.

CHAPTER 6

OVERALL DISCUSSION

This thesis investigates diagnostic levels of ionizing radiation in hospital patient populations and how these doses may affect long-term respiratory outcomes, with a primary focus on pregnancy-related exposures. Retrospective audits of two separate hospitalised cohorts, ICU patients and pregnant patients, showed that most patients cumulatively received < 1 mSv. Two animal models, assessing the long-term respiratory effects from *in utero* radiation exposure, demonstrated that compared to controls, ionizing radiation had no overall effect on the healthy respiratory system, nor did it alter the overall response to an acute lung injury stimulus for the outcomes measured. It did, however, cause persistent reduced growth from 4 - 16 weeks of age but only at doses >100 mGy.

6.1 CLINICAL AUDITS COMPARISON

Chapters 2 and 3 explored the doses of ionizing radiation received from diagnostic procedures to patients admitted to a tertiary hospital. Two polar populations were chosen based on the predicted exposure they would receive. Admitted ICU patients were predicted to have the highest levels of exposure in the hospital (Chapter 2), and admitted pregnant patients were predicted to have among the lowest exposure (Chapter 3). This predicted difference stems from the risk of prescription, which was discussed in their respective chapters. ICU patients are typically older and severely ill, thus the clinician's

risk to benefit analysis is focused more on the acute timeframe and the greater risk to the patient of not prescribing the scan. The risk of stochastic effects (chance-like relationship), such as cancer, are less than the chances of them surviving long enough to develop that cancer. In contrast, pregnant patients are generally younger, and the clinician's risk assessment extends to include that of the developing foetus. This shifts the focus to long term outcomes, as there is a lifetime of risk remaining for the mother, but more importantly for her foetus. This allows for stochastic effects of radiation to hold more weight in the risk to benefit ratio analysis. In addition, there are several studies that have highlighted that some clinicians do not accurately understand radiation dose and risk, misunderstanding procedure doses such as CT or fluoroscopy, and inferring incorrect levels of risk, such as cancer induction 1^{-6} . This inevitably ends in risk overestimation, which may make these clinicians reluctant to prescribe lifesaving diagnostic radiation. These two different points of focus, predicted to lead to different perception-based hypotheses when applied to the risk to benefit analysis, were ultimately supported with the median and highest cumulative exposures being 0.91 and 199.89 mSv for ICU patients (Chapter 2) and 0.02 and 19.07 mSv for pregnancy patients (Chapter 3).

In comparison to other departments or patient populations within a hospital, the median doses from Chapter 2 and 3 are low. ED, cancer patients, cardiovascular patients, haematological patients, and hospital-wide studies have reported median cumulative exposures of $11.7 - 34.1 \text{ mSv}^{7-11}$. These cumulative exposures are large and variable due to the low number of patients and specific disease inclusion criteria for each cohort, such as cancer patient that present as effective doses per year to include follow up visits and therefore a longer period to accumulate dose. In comparison, the dose from Chapter 2

(0.91 mSv) is low, but it is comparable to a recent ICU cohort which had a median cumulative dose of 0.72 mGy¹². The slight variation between ICU cohorts may be because this study included short-term stays (<5 days) which could have reduced the median dose. Chapter 3 had a median dose of 0.02 mSv, which is very low in comparison to other hospitalized populations but is also low in comparison to a large, USA-based pregnant cohort¹³. Lazarus *et al* reported an average cumulative exposure between 0.82 – 2.1 mGy between 1997 and 2006, which is higher compared to the mean dose from Chapter 3, 0.71 mSv¹³. That study has not clearly defined how they completed dose calculations and therefore it is possible mathematical differences could account for some variability. Additionally, when assessing individual modalities, the proportions of procedures are similar, but they reported much higher average doses for all modalities, possibly indicating that better technology and protocols to reduce dose more effectively are currently used.

However, as stated in Chapter 2 and 3, median cumulative exposures can also vary when eligibility criteria require specific procedures for inclusion, such as CT scans or fluoroscopic examination, which are relatively high dose procedures and therefore increase cumulative exposure. This is evident in an ICU study which only included patients that received a CT and reported a median cumulative exposure of 12.6 mSv¹⁴. In deciding not to limit based on this criteria, Chapter 2 and 3 were able to include a larger cohort, therefore showing a more representative and accurate understanding of dose delivered to these populations, as well as the frequency of procedures prescribed.

A deeper comparison between the two chapters highlights that the main reason for the variation in median dose is due to the use of high dose techniques, such as abdominal and

pelvic CT. In the ICU cohort (n = 526), 156 abdominal and pelvic CT scans were received totalling 3867.8 mSv, whereas in the pregnant cohort (n = 547), 4 scans totalled 50.2 mSv. Of the pregnant patients that did receive a CT scan, 3 out of the 4 were performed due to delivery-related admissions and 3 patients were admitted to the ICU, evidently describing how sick these patients were. Most scans in the ICU cohort (50.6%) came from gastrointestinal/sepsis patients, which accounted for 26.5% of total patients. In contrast, gastrointestinal/sepsis-related admissions only accounted for 10.6% of the pregnancy cohort population and none received a CT. This raises the question: are CT scans avoided due to pregnancy? From the sheer difference in the number of scans, it would appear that clinicians do avoid CT in pregnant patients, but this comparison cannot be accurately answered without matching patients that did or did not receive CT based on diagnosis and reason for prescription; for which this investigation was not designed.

This leads to a secondary question: if a diagnostic radiation procedure is prescribed, is the dose the pregnant patient receives reduced? Again, this question could be answered in the aforementioned comparison study design, but from the ICU and pregnancy cohorts in this thesis, the answer suggests yes. The average abdominal and pelvic CT scan in the ICU and pregnancy cohorts was 25.1 and 12.6 mGy, respectively. However, the literature suggests that on average, pregnant patients may receive more. Using foetal dose as a proxy for pregnant patient dose, from studies that solely look at CT scans, and from studies that are not matched by diagnosis or scan intention, the average abdominal and pelvic CT ranges between 13.3 - 21.9 mGy for ICU studies and 17.1 - 28.7 mGy for pregnancy studies^{12,13,15-19}. This higher exposure could be explained by pregnant patients only receiving CT scans in severe cases, as seen in Chapter 3, which might also explain the

low number of recruited patients, but high doses observed by these studies. Nonetheless, without further exploration, these overarching questions cannot be answered.

The intention behind these clinical audits was twofold. First, they are an educational tool to understand what dose of radiation our clinicians are prescribing, with the intent to relay this information back to the clinicians and to inform how these doses relate to risk. From these chapters, we observed that;

- Both cohorts had negligible exposures to ionizing radiation from diagnostic procedures, with medians of both cohorts below the recommended public limit of 1 mSv.
- There were factors associated with increased cumulative effective dose, however, these factors combined illustrated patients with a higher severity of illness and longer hospital stays.

Overall, these results should highlight to the clinician the relatively low cumulative exposures, and thus low level of radiation risk, these patients receive. This should lead to a better understanding of dose and risk and an easement surrounding the fear of ionizing radiation. The second intention behind these audits was to provide a translational foundation to choose clinically related exposure levels for the animal studies to investigate the long-term effects of *in utero* exposures.

6.2 ANIMAL STUDIES COMPARISON

Chapter 4 and 5 explored the long-term effect on growth and the respiratory system of *in utero* exposure to ionizing radiation. For both chapters, exposures up to 1000 mGy did

not significantly alter the outcomes measured for the respiratory system nor did the exposures in Chapter 4 affect cardiovascular parameters. A previous similar mouse model with similar doses reported no overall cardiovascular effects arising from a single, late gestation, ionizing radiation exposure²⁰. No such study has occurred in humans, but adult chronic environmental and occupational exposures have also not linked ionizing radiation exposure to negative cardiovascular outcomes^{21–23}. To the author's knowledge, there have been no long-term investigations into the effect of foetal exposure to ionizing radiation on the respiratory system. Some studies have observed respiratory fibrotic responses but these studies are on adults and required much higher doses, sometimes 20 -60 Gy^{24-26} . Instillation of LPS to cause acute lung injury in Chapter 5 worked as expected, but in utero radiation exposure had no effect on the manifestation of acute lung injury or the immune response. It is hypothesised in both studies that the length of time between irradiation and outcome was the reason for lack of immunomodulation observed, suggesting that an examination at a shorter time period post-irradiation would have found radiation-induced immunomodulation of the immune system, however, further studies are required to confirm this hypothesis.

The intentions behind the significant time delay between irradiation and main outcomes were threefold. Firstly, growth rate was observed over a long period of time to ensure that the pups didn't 'catch-up' the weight reduction back to the non-exposed normal level. As can be seen from the growth curves of both chapters, the curves trend separately and do not appear to realign as the pups age. Secondly, this outcome acted as our positive control that irradiation had occurred, as growth reduction from irradiations on gestational day 15 had already been observed²⁰. Lastly, to our knowledge, no one had looked at the

respiratory system in response to *in utero* ionizing radiation, thus, to avoid outcomes of death or malformations and focus on the possible 'foetal programming' effects of organ development, late gestation was chosen. Gestational Day 15 is in between two major stages of respiratory development responsible for structural integrity and cellular differentiation, both important in function and physiology later in life^{27,28}. Additionally at this time, the liver begins devotion to metabolic activities and so the main site of haematopoiesis moves to the thymus and spleen, before final relocation to the bone marrow^{29,30}. Thus, gestational day 15 is also situated in an important phase of the immune system development and disruption at this time may also cause 'programmable' effects.

It would be intriguing to understand if the growth restriction, a hallmark of foetal programming, was caused by maternal effects, direct exposure, or if it required the combination/interaction of both effects³¹. However, due to ionizing radiation affecting the tissue in both an indirect and direct method, it is hard to delineate the cause of the effect. Irradiating a mother and surgically swapping her offspring *in utero* with unirradiated offspring could demonstrate both maternal exposure with control foetuses (indirect) and foetal exposure within control mothers (direct), but technically this would be almost impossible and come with a plethora of maternal and pup stress that would inevitably make the findings redundant. Interestingly, irradiated pups have reduced growth postnatal and non-irradiated mothers, indicating that it is an intrauterine event that programs long-term reduced weight and not the mother's ability to nurture her pups³². Nevertheless, the delineation of cause of growth restriction has yet to be answered.

In respiratory outcomes, a baseline understanding was required (Chapter 4) before further investigation into inflammatory responses during acute lung stimulus (Chapter 5) could commence. It is important to recognise that when comparing the respiratory outcomes from these two animal studies, one used a healthy model (Chapter 4) whilst the other used an acute lung injury model (Chapter 5). Although there were non-stimulated animals in Chapter 5 (Saline control), they were anaesthetised, intubated and instilled with saline, which inevitably causes tracheal damage and slight lung inflammation³³. Thus, these cohorts are intrinsically different from each other methodologically.

Additionally, as noted in Chapter 5, the two animal studies used different strains of mice; Chapter 4 used C57BI/6 mice while Chapter 5 used BALB/c mice. The rationale for changing the strain was due to the offspring cannibalism rates of C57BL/6 dams, which was also the reason why pre-weaning weights were not collected. BALB/c mice are generally considered to be less cannibalistic, but as this was the first trial by our group with this strain of mice, caution was taken to observe their motherhood behaviour and cannibalism potential with the least interference possible, which resulted in pre-weaning weight not being collected³⁴. It is possible that the growth restriction seen in both chapters could extrapolate to preweaning and even birthweight as BALB/c mice and Wistar rats showed birth weight reductions after *in utero* exposures, even down to 200 mGy^{35,36}. Typically, this lower dose recovers, and it is only the high doses (>1000 mGy) that persist long term.

From the outcomes measured, growth was permanently affected by *in utero* ionizing radiation in both sexes and both strains. The reduction in growth following 1000 mGy in

C57BL/6 mice was found with only 100 mGy in BALB/c mice. Others have observed growth restrictions at 1000 - 1500 mGy exposures, but this is the first study in BALB/c to observe persistent growth restriction in both sexes at $100 \text{ mGy}^{20,32,36-39}$. The significant difference between strains may be due to the dissimilarity in radiosensitivity levels. C57BL/6 mice are more radioresistant than BALB/c mice, with adult LD_{50/30} of 8.5 and 4.5, respectively⁴⁰⁻⁴³. Grahn *et al* reported that the LD_{50/30} of ionizing radiation increases with age, so when inferring retrospectively, *in utero* or postnatal exposures should require less dose than adults⁴⁴. However, an earlier study identified a nonlinear age relation to dose in C57Bl/6 mice, in which postnatal day 1 mortality was similar to postnatal day 60^{45} . It is unclear if this phenomenon is consistent with other strains, but the actual LD_{50/30} of foetal exposures are still unknown and should be explored in the future.

The rationale behind the difference in adult radiosensitivity is unknown but a few interesting observations regarding genomic instability have been postulated in trying to answer this question. BALB/c mice, or crossbreds with maternal BALB/c background, had less mitochondrial DNA damage and more mitochondrial permeability after 1000 mGy irradiation⁴⁶. The increased permeability triggers mitochondrial-induced apoptosis leading to a higher mortality and therefore a higher radiosensitivity. As mitochondria are passed down via the ovum, maternal inheritance is observed. Additionally after 1000 mGy, cytogenetic aberrations were observed only in BALB/c mice, not C57BL/6 mice^{47,48}. BALB/c mice have a detrimental yet normal genetic variation in DNA protein kinases that repair DNA double-strand breaks, whereas C57BL/6 mice have normal functioning kinases⁴⁹. A genomic analysis of crossbreeds identified that numerous positive DNA repair genes are from paternal transfer and not maternal transfer, indicating the importance of

paternal inheritance⁵⁰. Thus, radiosensitivity is an element of both maternal and paternal susceptibility in terms of genomic instability, however, further research is required to fully understand how different radiosensitivities occur.

Due to the strain variation, an interesting question arises: would the respiratory results in Chapter 5 be similar if C57BI/6 mice were used? Comparison between lung injury models can be difficult due to the LPS dose and delivery technique, BAL technique, and method of outcome quantification between individual studies. Keeping the radiosensitivities aside, there are some slight differences between these strains that need to be highlighted to answer this question.

In normal healthy adulthood, these strains are fairly similar in body weight, so any calculations made during respiratory measurement should be consistant⁵¹. Respiratory anatomy is very similar in structure and lung volume but 3D visualisation shows bulging of main airways in C57BL/6 mice compared to BALB/c⁵¹. After intratracheal instillation of LPS, C57BL/6 and BALB/c mice have similar physiological responses⁵². They both have pulmonary oedema, alveolar-capillary barrier breakdown, and lung hyperinflation^{52,53}. In terms of immunological responses, they both have cellular infiltrate that is dominated by neutrophils but there are some slight differences in BAL cytokine profile^{52,54}. BALB/c mice appear hyperresponsive to LPS 4 h post instillation, with increased BAL concentrations of pro-inflammatory mediators, but at 24 h, C57BL/6 and BALB/c were comparable⁵⁴. Thus, when discounting any effects of radiation, it appears that if C57BL/6 mice were used in Chapter 5, it could be assumed that the physiological and immunological responses 24

hours after intratracheal LPS would be very similar to the BALB/c results, even though this is not necessarily the case for other lung injury models.

In a fibrosis model, intratracheal instillation of bleomycin causes severe fibrosis with reduced lung function in C57Bl/6 mice but BALB/c mice were less susceptible⁵⁵. Matrix metalloproteinases, college deposition, TGF- β , and connective tissue growth factor are all increased in C57Bl/6 compared to BALB/c, with connective tissue growth factor being a key player in strain-variable fibrotic response^{56–60}. In an ovalbumin asthma model, methacholine causes less airway constriction, measured via airway resistance and Penh, in C57Bl/6 mice compared to BALB/c⁶¹. Both had peribronchial and alveolar cellular infiltrate, but C57Bl/6 appeared to have a more eosinophilic, mixed T helper lymphocyte subset 1 ((Th₁) pro-inflammatory type) and subset 2 ((Th₂) allergic type), response whereas BALB/c had a mast cell dominant, Th₂ response, deemed responsible for the airway hyperresponsiveness. Similar airway responsiveness was observed for an LPS/methacholine model⁵³.

In terms of combining inflammation and radiation into strain differences in immunomodulation, there appear to be no studies assessing this effect. In fact, there are few that investigate immunomodulation *in vivo*, with only 2 from the same author, who used the C57BL/6 mouse strain^{62,63}. The majority instead use cell lines or *ex vivo* BALB/c primary cells, and co-cultures typically using human cells^{64–68}. As the mouse strains react similarly to LPS, as discussed above, the assumptions of *in vivo* strain differences would be based on the radiosensitivity differences between the animal strains themselves. Therefore, evidence suggests that if C57BL/6 mice were used in Chapter 5, no biologically

significant immunomodulatory effects would be seen, due to the higher radioresistance in C57Bl/6 mice.

The intention behind these animal models was to investigate the effect on the respiratory system from *in utero* exposures to diagnostic levels of ionizing radiation. From *in utero* exposure to ionizing radiation, it was observed that:

- There was no effect on the normal and challenged respiratory systems later in life, nor the cardiovascular system over the 8 weeks of measurement.
- There were significant reductions in body weight of both strains, at doses
 > 100 mGy.

These results support the idea that a single, whole-body, exposure of ionizing radiation, up to 1000 mGy during pregnancy, in two strains of mice, does not significantly affect the cardiovascular and respiratory system during adolescence, and at doses received from diagnostic radiation, does not affect animal growth. However, the growth outcomes between 10 and 100 mGy should be explored to support this statement.

6.3 TRANSLATING ANIMAL FINDINGS TO HUMANS

Extrapolating animal studies to human exposures is imperfect, and for therapeutic studies can be disheartening when effective treatments appear promising *in vivo* but fail in clinical trials. It is approximated that 30-40% of high citing animal research translates to clinical trials and only about 4% passes to phase 2 testing^{69,70}. Several factors contribute to this lack of translation including funding, ethical consideration, technical limitations, statistical power, and methodological inconsistencies, either between studies or due to complex comorbidities, external stimuli and genetic diversity apparent in human life^{71–74}. But some lack of translation also comes from the obvious limitation that animals are not humans. Although there are many similarities, there are significant differences in many mouse systems that ultimately inhibit translation of results to humans. For example, the immune system differs in normal cellular proportions and stimulated transcriptomics, and the respiratory system differs in structure and physiology^{75–77}. However, there are several obvious positives to animal research, such as ease of environmental or genetic manipulation, ability to screen new substances or interventions, exploratory insight into the pathophysiology of disease, and the ability to model the whole organism rather than at a cellular level, but care in interpretation and translation should be taken.

The results from BALB/c mice (Chapter 5) most resemble what could be predicted upon human exposures due to their radiosensitivity similarities^{44,78}. A main outcome from this chapter was that ionizing radiation during late gestation caused persistently reduced weights of offspring, but only at 100 and 1000 mGy. When recalling data from Chapter 2 or 3, and from the literature, no single diagnostic procedure reached 100 mGy, nor do pregnant patients come close to receiving this dose cumulatively⁷⁹. Therefore, pregnant patients receiving diagnostic radiation are unlikely to incur intrauterine growth restriction or long-term weight reduction of their foetus. However, testing of doses between 10 and 100 mGy would need to occur to confirm this statement, as there are some patients who receive > 10 mGy. Nevertheless, reductions in infant weight due to exposures equalling 100 mGy, a weight reduction of 8-10%, would remain well within normal birthweight variation of the Australian population⁸⁰. Therefore, even though the radiation exposure has stunted the growth potential of the offspring, the weight reduction would not be observable through the population's variation from environmental and genetic factors. Importantly, this weight reduction does not meet the requirements for low birth weight and therefore would not have increased risks associated with low birth weight. As the patient's exposure rises over 100 mGy and advances closer to 1000 mGy, it could be assumed that the weight reduction would follow the trend of the mouse models, eventually leading to low birth weight infants, however this assumption would need to be confirmed by mapping the dose-response between 100 and 1000 mGy. If this assumption was supported, the clinical impact of why the patient required so many or such detailed procedures in order to receive doses that high would need to be acknowledged⁸⁰.

Although no changes were observed in respiratory and cardiovascular outcomes measured, changes in other animal model's organ systems have been observed as dose increases up to the 1000 mGy exposure, suggesting similar changes could occur in humans. When observing studies that used single whole-body exposures at similar gestational ages to Chapters 4 and 5, *in utero* exposures up to 500 mGy can cause slight changes in mouse brain physiology and cognitive behaviour, decreasing anxiety and increasing exploration up to 1-year post irradiation^{81–84}. Female rats have growth reduction and males appear to have slight reflex delays and slowed appearance of some physiological markers^{35,85}. At exposures up to 1000 mGy, changes in brain physiology and cognitive behaviour are more apparent, with mice less nervous and more adventurous but with some reduced locomotion and learning ability up to 1.5 years post irradiation^{81–84}. The growth of the rat continues to be reduced and there is further reduction in reflex and physiological markers^{85,88}. At exposures above 1000 mGy, there is reduced pup

viability and persistent growth reduction in both males and females of mice and rats^{20,32,36–39,89}. Female rats are less fertile, and there are severe deficits in mice and rat brain physiology with less neurogenesis^{32,38,81,90–92}. Mice have severely reduced learning ability, memory and spatial awareness and have changes to their circadian rhythm^{82,83,86,93}. This shows that *in utero* exposures of rodent models can have negative effects on other organ systems in the offspring in which severity increases with dose.

At equivalent gestational exposures to ionizing radiation, humans can have similar outcomes. Much of the evidence of radiation harm to humans comes from the lifespan study (LSS) of the atomic bombing in WWII. Growth reduction, neurobiological defects, behavioural changes and mental retardation have been reported from humans exposed *in utero* to the from bomb, with severity based on gestation at exposure and dose^{94–97}. Similar to animal experiments, there appears to be a threshold dose of 300 – 500 mGy for these outcomes. When considering all of this evidence, many governing bodies agree that exposures below 100 mGy pose no significant negative outcomes for humans, nor does it merit foetal termination^{98–102}. The animal studies in this thesis support this statement as, at diagnostic levels of ionizing radiation received during late gestation, there were no biologically significant effects in the outcomes assessed.

Additional to the deterministic effects, the universal question that arises from radiation exposure is the presumed stochastic effects – e.g. the risk of cancer. As introduced in Chapter 1, this topic is highly contentious with clinical studies showing both increased risk and no association^{103–109}. Many that state a risk of cancer from *in utero* exposure to ionizing radiation use case-control study designs, which do not have the power to

delineate cause and effect, only an association, whereas those that use cohort study designs, which can delineate cause and effect, find no relationship¹¹⁰. However, if an assumption is made that there is a significant risk of cancer, this risk from *in utero* exposures is less than from childhood exposures¹⁰³. This finding does not fit with the typical understanding that older individuals are more resistant to radiation-induced cancer than younger individuals¹¹¹. The finding is supporting by foetal exposures, for both mice and humans, having less genetic instability than child or adult exposures^{112,113}. Evidence suggests that aberrant cells, induced by radiation, are eliminated during postnatal growth resulting in lower susceptibility to cancer, as inferred from exposed neonatal mice having cytogenetic aberrations persistent only up to 48 hours¹¹³.

When inferring any of these translational results to what is expected from exposures to diagnostic radiations, there are a few limitations that need consideration. Comparison between mouse gestation and human gestation is often difficult due to the difference in organ development between species. Gestational day (GD) 15 in mice can relate to 6 - 12 weeks in humans depending on the organ system under comparison, for example, the cardiovascular and respiratory system equate to approximately 10 - 12 weeks human gestation^{27,28,114,115}.

For direct human-human comparison, the subjects in the Lifespan study are inherently different from those of patients who receive diagnostic radiation as they have very different covariates. The blast from the atomic bomb resulted in poverty and famine in the surrounding region, adding to the already worn-torn state of the communities. Increased anxiety, post-traumatic stress and other mental illnesses arose not only from
their experiences of loss of family and community but also from the prejudice and discrimination in employment and marriage^{116,117}. People who receive diagnostic radiation do so based on a clinical justification highlighting an underlying issue or disease requiring attention and thus healthy people are not exposed to medical radiation. It is important to acknowledge these limitations in human-human comparisons, especially when translating results and inferring outcomes from one to another.

The idea that healthy people are not exposed to medical radiation is the underlying reasoning behind Totter and MacPherson's rebuttal to the question: does CT cause cancer¹¹⁸? As highlighted in Chapter 1, they argue that comparison between patients that received CT scans and those that don't, is itself, a limiting factor because they inherently resemble two separate subpopulations. One group has a clinical justification that requires a CT scan whereas the other does not. Thus, is it this selection factor of the clinical justification to prescribe diagnostic radiation, and not the radiation itself, that is responsible for any associated risks? Does the disease cause the CT or does the CT cause the disease? This question is impossible to answer without irradiating healthy people and following them long term. All these limitations make it difficult to interpret and translate animal and epidemiological studies and so it should be conducted with a level of caution.

6.4 OVERALL CONCLUSIONS

The main objective of this thesis was to examine the long-term effects on the respiratory system of low doses of ionizing radiation relevant to medical exposures. However, it was first necessary to understand what levels of ionizing radiation hospitalised patients receive. At the time of beginning this thesis, there was little current understanding of *in utero* radiation exposure and no understanding in terms of the respiratory system. Through the approaches discussed in chapter 2 - 5, this thesis has expanded this knowledge by

- 1. Demonstrating that most patients admitted to the ICU of a tertiary hospital cumulatively received less than 1 mSv.
- 2. Demonstrating that most pregnant patients admitted to a tertiary hospital cumulatively received less than 0.2 mSv, with a median cumulative foetal dose of 0.01 mSv.
- Demonstrating that *in utero* exposure to diagnostic levels of ionizing radiation did not cause any negative long-term outcomes on growth, cardiovascular system or respiratory system in a healthy animal model. However, at 1000 mGy exposures, persistent growth reduction was observed.
- 4. Demonstrating that *in utero* exposure to diagnostic levels of ionizing radiation does not significantly alter the pulmonary immune response within an acute lung injury model. However, at 100 and 1000 mGy exposures, persistent growth reduction was observed.

The clinical and animal work which contributed to this thesis, and the interpretation of the translation between them, strengthens the hypothesis that current doses received from diagnostic radiation during pregnancy are unlikely to have any long-term detrimental effects on the offspring.

6.5 FUTURE DIRECTIONS

Future studies carrying on from the clinical work should try to answer the following questions: Are the cumulative exposures received at FMC replicable at other hospitals around the world? Are clinicians cautious about using diagnostic radiation? Will improving understanding of dose and risk improve patient outcomes? Using similar eligibility criteria, simple comparisons could be made to gain an Australia-wide and worldwide understanding of patient exposures. Although single patient exposures depend on several factors, gaining a uniform understanding of cumulative exposures for similar populations around the world, could also highlight which countries/hospitals freely prescribe diagnostic radiation and which are more conservative. If there are differences observed, it would be helpful to know if clinicians understanding of radiation could relate to the cumulative levels received at their hospitals. As discussed in Chapter 1, the past 2 decades of assessments have highlighted a lack of professional understanding of ionizing radiation, and even though the scoring slightly improves with time, the level of misunderstanding of dose and risk may inversely relate to the clinician's use of diagnostic radiation^{1-6,112-116}. This may also help to answer the second question: are clinicians cautious about using ionizing radiation? This question could be answered simply by qualitative questionnaires, but this questionnaire could also be used to explore two related questions: If clinicians are cautious about using diagnostic radiation, what techniques or methods have they replaced this with to aid in diagnosis? And, would providing additional education change the clinician's view of ionizing radiation and their attitude towards its use?

The use of other techniques, such as ultrasound or magnetic resonance imaging, to replace ionizing radiation procedures during diagnosis has occurred with some success, but only in certain diseases^{124–126}. Although they reduce ionizing radiation exposure, some procedures are limited in their ability to give a conclusive diagnosis, without requiring additional scans or methods, when the primary suspicion was negative. It would be important to delineate if clinicians are opting for alternative methods due to the perceived risks associated with ionizing radiation or due to other personal reasons.

Finally, it would be interesting to observe if providing additional education to the clinicians about ionizing radiation would change their opinion and usage rates. If they accurately understood the dose and the risks involved with ionizing radiation, would the clinicians still choose alternative methods of diagnosis or would they use ionizing radiation? Assessing usage rates of procedures that use ionizing radiation before and after education would provide this answer, but more importantly, is there a link between improved education and improved patient outcomes? This could be possible by addressing disease positive findings, length of stay or other patient outcomes associated with improved diagnosis, in the previous study design and may also, inadvertently, answer the question does ionizing radiation avoidance detrimentally affect the patient. Consideration should be applied in that each clinician has likely received different training or experiences that all influence their clinical judgement but educating clinicians on the current understanding of radiation dose and risk could reduce the current 'scared' stigma that may inevitably save a life.

Future studies relating to the animal models should focus on filling in the gaps of this thesis in dose, timeframe and outcomes, to adequately assess at what point growth is affected and whether respiratory and cardiovascular outcomes are altered by *in utero* radiation. BALB/c mice should be continued in the investigation of further questions as they relate better to humans in radiosensitivity.

For the growth outcome, the dose curve between 10 – 1000 mGy should be further explored to assess at what point ionizing radiation is sufficient to cause statistically significant growth reduction, and to assess if this outcome is linear, quadratic, biphasic, or suits another model. Collection of preweaning weight could delineate if any catch-up effect was present but not observed in the mice exposed to 10 mGy. Additionally, extending the study would confirm that the dose reduction observed by 100 and 1000 mGy remain permanent.

The timeframe should also be altered for the respiratory and cardiovascular outcomes of this thesis. For respiratory outcomes, the animal study should be shortened to observe the effects on the postnatal and childhood timeframes. It would be interesting to report the effects of ionizing radiation on the development of respiratory physiology by cataloguing different developmental stages post-irradiation. As the lungs are still developing post-birth, changes to physiology might have been resolved before this thesis examined the lungs^{127–129}. Shortening the time between irradiation and outcome may also allow the immunomodulatory effects from ionizing radiation to be observable and therefore relate clinically to infant respiratory infections post *in utero* exposures.

Conversely, the length of the study should be extended for the cardiovascular outcomes as changes, especially hypertension, are more common in late adulthood timeframe¹³⁰.

Finally, this thesis unfortunately missed investigating a few key outcomes that may help improve the overall understanding of the effect of ionizing radiation. Assessing body composition could help determine if the reduced growth rate observed in the animal studies was a reduction of body fat, muscle or was proportional for both. Investigating metabolic activity, through tissue analysis, could help understand if the reduced weight is an effect of altered metabolism. Additionally, exploring if low birth weight animals have any other outcomes associated with low birth weight infants. Adding functional changes in the respiratory outcomes, such as incorporation of respiratory mechanics, breath rate or blood oxygenation readings, would provide a basic understanding of radiation effects on the functionality of the respiratory system. Lastly, although no functional changes to the cardiovascular system were observed from the outcomes measured, confirmation could be obtained via more sensitive echocardiographic measurements. Expanding outcomes to include hypertrophy, or fibrotic responses by Masson's trichrome may assist in assessing long-term risk of cardiovascular disease. There are many directions in how these studies can be continued to further contribute to the area of prenatal radiation exposure. This has the ability to further clinical understanding of diagnostic radiation and if used appropriately, has the potential to save a life.

- 1. Lee CI, Haims AH, Monico EP, Brink JA, Forman HP. Diagnostic CT Scans: Assessment of Patient, Physician, and Radiologist Awareness of Radiation Dose and Possible Risks. *Radiology*. 2004;231(2):393-398. doi:10.1148/radiol.2312030767.
- 2. Correia MJ, Hellies A, Andreassi MG, Ghelarducci B, Picano E. Lack of radiological awareness among physicians working in a tertiary-care cardiological centre. *Int J Cardiol*. 2005;103(3):307-311. doi:10.1016/j.ijcard.2004.08.070.
- 3. Shiralkar S, Rennie A, Snow M, Galland RB, Lewis MH, Gower-Thomas K. Doctors' knowledge of radiation exposure: questionnaire study. *BMJ*. 2003;327(7411):371-372. doi:10.1136/bmj.327.7411.371.
- 4. Keijzers GB, Britton CJ. Doctors' knowledge of patient radiation exposure from diagnostic imaging requested in the emergency department. *Med J Aust*. 2010;193(8):450-453. doi:10.5694/J.1326-5377.2010.TB03998.X.
- 5. Krille L, Hammer GP, Merzenich H, Zeeb H. Systematic review on physician's knowledge about radiation doses and radiation risks of computed tomography. *Eur J Radiol*. 2010;76(1):36-41. doi:10.1016/j.ejrad.2010.08.025.
- 6. Groves AM, Yates SJ, Win T, Kayani I, Gallagher FA, Syed R, Bomanji J, Ell PJ. CT Pulmonary Angiography versus Ventilation-Perfusion Scintigraphy in Pregnancy: Implications from a UK Survey of Doctors' Knowledge of Radiation Exposure. *Radiology*. 2007;240(3):765-770. doi:10.1148/radiol.2403050910.
- 7. Stiles BM, Mirza F, Towe CW, Ho VP, Port JL, Lee PC, Paul S, Yankelevitz DF, Altorki NK. Cumulative Radiation Dose From Medical Imaging Procedures in Patients Undergoing Resection for Lung Cancer. *Ann Thorac Surg.* 2011;92(4):1170-1179. doi:10.1016/j.athoracsur.2011.03.096.
- 8. Carpeggiani C, Landi P, Michelassi C, Marraccini P, Picano E. Trends of Increasing Medical Radiation Exposure in a Population Hospitalized for Cardiovascular Disease (1970–2009). Keyhani S, ed. *PLoS One*. 2012;7(11):e50168. doi:10.1371/journal.pone.0050168.
- 9. De Mauri A, Brambilla M, Chiarinotti D, Matheoud R, Carriero A, De Leo M. Estimated Radiation Exposure from Medical Imaging in Hemodialysis Patients. *J Am Soc Nephrol*. 2011;22(3):571-578. doi:10.1681/asn.2010070784.
- 10. Lutterman AC, Moreno CC, Mittal PK, Kang J, Applegate KE. Cumulative Radiation Exposure Estimates of Hospitalized Patients from Radiological Imaging. *J Am Coll Radiol*. 2014;11(2):169-175. doi:10.1016/j.jacr.2013.08.028.
- 11. Tien HC, Tremblay LN, Rizoli SB, Gelberg J, Spencer F, Caldwell C, Brenneman FD. Radiation exposure from diagnostic imaging in severely injured trauma patients. *J Trauma Inj Infect Crit Care*. 2007;62(1):151-156. doi:10.1097/TA.0b013e31802d9700.
- 12. Krishnan S, Moghekar A, Duggal A, Yella J, Narechania S, Ramachandran V, Mehta A, Adhi F, Vijayan AKC, Han X, Wang X, Dong F, Martin C, Guzman J. Radiation Exposure in the Medical ICU: Predictors and Characteristics. *Chest*. 2018;153(5):1160-1168. doi:10.1016/j.chest.2018.01.019.
- 13. Lazarus E, DeBenedectis C, North D, Spencer PK, Mayo-Smith WW. Utilization of Imaging in Pregnant Patients: 10-year Review of 5270 Examinations in 3285 Patients—1997–2006. *Radiology*. 2009;251(2):517-524. doi:10.1148/radiol.2512080736.
- 14. Slovis BH, Shah KH, Yeh DD, Seethala R, Kaafarani HMA, Eikermann M, Raja AS, Lee J. Significant but reasonable radiation exposure from computed tomography-related medical imaging in the ICU.

Emerg Radiol. 2016;23(2):141-146. doi:10.1007/s10140-015-1373-y.

- 15. Woussen S, Lopez-Rendon X, Vanbeckevoort D, Bosmans H, Oyen R, Zanca F. Clinical indications and radiation doses to the conceptus associated with CT imaging in pregnancy: a retrospective study. *Eur Radiol*. 2016;26(4):979-985. doi:10.1007/s00330-015-3924-8.
- 16. Goldberg-Stein S, Liu B, Hahn PF, Lee SI. Body CT during pregnancy: Utilization trends, examination indications, and fetal radiation doses. *Am J Roentgenol*. 2011;196(1):146-151. doi:10.2214/AJR.10.4271.
- 17. Ozbayrak M, Cavdar I, Seven M, Uslu L, Yeyin N, Tanyildizi H, Abuqbeitah M, Acikgoz AS, Tuten A, Demir M. Determining and Managing Fetal Radiation Dose from Diagnostic Radiology Procedures in Turkey. *Korean J Radiol.* 2015;16(6):1276-1282. doi:10.3348/kjr.2015.16.6.1276.
- 18. Rohner DJ, Bennett S, Samaratunga C, Jewell ES, Smith JP, Gaskill-Shipley M, Lisco SJ. Cumulative total effective whole-body radiation dose in critically ill patients. *Chest*. 2013;144(5):1481-1486. doi:10.1378/chest.12-2222.
- 19. Salottolo K, Bar-Or R, Fleishman M, Maruyama G, Slone DS, Mains CW, Bar-Or D. Current utilization and radiation dose from computed tomography in patients with trauma. *Crit Care Med*. 2009;37(4):1336-1340. doi:10.1097/CCM.0b013e31819d6739.
- 20. Sreetharan S, Stoa L, Cybulski ME, Jones DE, Lee AH, Kulesza A V., Tharmalingam S, Boreham DR, Tai TC, Wilson JY. Cardiovascular and growth outcomes of C57Bl/6J mice offspring exposed to maternal stress and ionizing radiation during pregnancy. *Int J Radiat Biol*. March 2019:1-9. doi:10.1080/09553002.2019.1589025.
- 21. Berrington A, Darby SC, Weiss HA, Doll R. 100 years of observation on British radiologists: Mortality from cancer and other causes 1897-1997. *Br J Radiol*. 2001;74(882):507-519. doi:10.1259/bjr.74.882.740507.
- 22. Bolotnikova MG, Koshurnikova NA, Komleva NS, Budushchev EB, Okatenko P V. Mortality from cardiovascular diseases among male workers at the radiochemical plant of the "Mayak" complex. *Sci Total Environ*. 1994;142(1-2):29-31. doi:10.1016/0048-9697(94)90070-1.
- 23. Vrijheid M, Cardis E, Ashmore P, Auvinen A, Bae JM, Engels H, Gilbert E, Gulis G, Habib RR, Howe G, Kurtinaitis J, Malker H, Muirhead CR, Richardson DB, Rodriguez-Artalejo F, Rogel A, Schubauer-Berigan M, Tardy H, Telle-Lamberton M, Usel M, Veress K. Mortality from diseases other than cancer following low doses of ionizing radiation: Results from the 15-Country Study of nuclear industry workers. *Int J Epidemiol*. 2007;36(5):1126-1135. doi:10.1093/ije/dym138.
- 24. Hanania AN, Mainwaring W, Ghebre YT, Hanania NA, Ludwig M. Radiation-Induced Lung Injury: Assessment and Management. *Chest*. 2019;156(1):150-162. doi:10.1016/j.chest.2019.03.033.
- 25. Evans WA, Leucutia T. Intrathoracic changes induced by heavy radiation. *Am J Roentgenol*. 1925;13:203-220.
- 26. Farr KP, Khalil AA, Grau C. Patient-reported lung symptoms and quality of life before and after radiation therapy for non-small cell lung cancer: correlation with radiation pneumonitis and functional imaging. *Acta Oncol (Madr)*. July 2019:1-5. doi:10.1080/0284186x.2019.1634835.
- 27. Maeda Y, Davé V, Whitsett JA. Transcriptional control of lung morphogenesis. *Physiol Rev.* 2007;87(1):219-244. doi:10.1152/physrev.00028.2006.
- 28. Pinkerton KE, Joad JP. The mammalian respiratory system and critical windows of exposure for children's health. *Environ Health Perspect*. 2000;108(SUPPL. 3):457-462. doi:10.1289/ehp.00108s3457.
- 29. Holladay SD, Smialowicz RJ. Development of the murine and human immune system: Differential effects of immunotoxicants depend on time of exposure. *Environ Health Perspect*. 2000;108(SUPPL.

3):463-473.

- 30. Landreth KS. Critical windows in development of the rodent immune system. *Hum Exp Toxicol*. 2002;21(9-10):493-498. doi:10.1191/0960327102ht287oa.
- 31. Barker D. The fetal and infant origins of adult disease. *BMJ Br Med J*. 1990;301(156):1111. doi:10.1136/bmj.301.6761.1111.
- 32. Murphree RL, Pace HB. The Effects of Prenatal Radiation on Postnatal Development in Rats. *Radiat Res.* 1960;12(5):495. doi:10.2307/3570971.
- 33. Fróes SDP, Souza ABF de, Matos NA de, Philips NE, Costa G de P, Talvani A, Cangussú SD, Bezerra FS. Intranasal instillation of distilled water, hypertonic saline and sodium bicarbonate promotes redox imbalance and acute lung inflammation in adult mice. *Respir Physiol Neurobiol*. 2019;266:27-32. doi:10.1016/j.resp.2019.04.013.
- 34. Weber EM, Algers B, Würbel H, Hultgren J, Olsson IAS. Influence of strain and parity on the risk of litter loss in laboratory mice. *Reprod Domest Anim.* 2013;48(2):292-296. doi:10.1111/j.1439-0531.2012.02147.x.
- 35. Jensh RP, Brent RL, Vogel WH. Studies concerning the effects of low level prenatal x-irradiation on postnatal growth and adult behaviour in the wistar rat. *Int J Radiat Biol*. 1986;50(6):1069-1081. doi:10.1080/09553008614551461.
- 36. Nash DJ, Gowen JW. Effects of X-irradiation upon postnatal growth in the mouse. *Biol Bull*. 1962;122(1):115-136. doi:10.2307/1539326.
- 37. Kimler BF, Norton S. Behavioral changes and structural defects in rats irradiated in utero. *Int J Radiat Oncol Biol Phys.* 1988;15(5):1171-1177. doi:10.1016/0360-3016(88)90200-3.
- 38. Jensh RP, Eisenman LM, Brent RL. Postnatal neurophysiologic effects of prenatal x-irradiation. *Int J Radiat Biol*. 1995;67(2):217-227. doi:10.1080/09553009514550271.
- 39. Hossain M, Uma Devi P, Bisht KS. Effect of prenatal gamma irradiation during the late fetal period on the postnatal development of the mouse. *Teratology*. 1999;59(3):133-138. doi:10.1002/(SICI)1096-9926(199903)59:3<133::AID-TERA4>3.0.CO;2-T.
- 40. Wang Q, Du L, Wang Y, Xu C, Sun Z, Fu Y, Yang B, Wang Y, Mu C, Fan S, Cai L, Katsube T, Liu Q. The development and biological characteristics of a novel potentially radioresistant inbred mouse strain. *Mol Med Rep.* 2017;15(2):759-767. doi:10.3892/mmr.2016.6066.
- 41. Kohn HI, Kallman RF. The Influence of Strain on Acute X-Ray Lethality in the Mouse: I. LD 50 and Death Rate Studies. *Radiat Res.* 1956;5(4):309. doi:10.2307/3570420.
- 42. Nunamaker EA, Artwohl JE, Anderson RJ, Fortman JD. Endpoint refinement for total body irradiation of C57BL/6 mice. *Comp Med*. 2013;63(1):22-28.
- Zhang SB, Maguire D, Zhang M, Tian Y, Yang S, Zhang A, Casey-Sawicki K, Han D, Ma J, Yin L, Guo Y, Wang X, Chen C, Litvinchuk A, Zhang Z, Swarts S, Vidyasagar S, Zhang L, Okunieff P. Mitochondrial DNA and functional investigations into the radiosensitivity of four mouse strains. *Int J Cell Biol*. 2014;2014:850460. doi:10.1155/2014/850460.
- 44. Grahn D, Hamilton KF. Genetic Variation in the Acute Lethal Response of Four Inbred Mouse Strains to Whole Body X-Irradiation. *Genetics*. 1957;42(3):189-198.
- 45. Abrams HL. Influence of Age, Body Weight, and Sex on Susceptibility of Mice to the Lethal Effects of X-radiation. *Proc Soc Exp Biol Med*. 1951;76(4):729-732. doi:10.3181/00379727-76-18610.
- 46. Kinnally KW, Peixoto PM, Ryu SY, Dejean LM. Is mPTP the gatekeeper for necrosis, apoptosis, or both? *Biochim Biophys Acta Mol Cell Res.* 2011;1813(4):616-622.

doi:10.1016/j.bbamcr.2010.09.013.

- 47. Ponnaiya B, Cornforth MN, Ullrich RL. Radiation-Induced Chromosomal Instability in BALB/c and C57BL/6 Mice: The Difference Is as Clear as Black and White. *Radiat Res.* 1997;147(2):121. doi:10.2307/3579411.
- 48. Rithidech KN, Udomtanakunchai C, Honikel LM, Whorton EB. No evidence for the in vivo induction of genomic instability by low doses of 137Cs gamma rays in bone marrow cells of BALB/cJ and C57BL/6J MICE. *Dose-Response*. 2012;10(1):11-36. doi:10.2203/dose-response.11-002.Rithidech.
- 49. Okayasu R, Suetomi K, Yu Y, Silver A, Bedford JS, Cox R, Ullrich RL. A deficiency in DNA repair and DNA-PKcs expression in the radiosensitive BALB/c mouse. *Cancer Res.* 2000;60(16):4342-4345.
- 50. Mukherjee S, Sainis KB, Deobagkar DD. F1 hybrids of BALB/c and C57Bl/6 mouse strains respond differently to low-dose ionizing radiation exposure. *J Genet*. 2014;93(3):667-682. doi:10.1007/s12041-014-0422-8.
- 51. Thiesse J, Namati E, Sieren JC, Smith AR, Reinhardt JM, Hoffman EA, McLennan G. Lung structure phenotype variation in inbred mouse strains revealed through in vivo micro-CT imaging. *J Appl Physiol*. 2010;109(6):1960-1968. doi:10.1152/japplphysiol.01322.2009.
- 52. Alm AS, Li K, Chen H, Wang D, Andersson R, Wang X. Variation of lipopolysaccharide-induced acute lung injury in eight strains of mice. *Respir Physiol Neurobiol*. 2010;171(2):157-164. doi:10.1016/j.resp.2010.02.009.
- 53. Card JW, Carey MA, Bradbury JA, DeGraff LM, Morgan DL, Moorman MP, Flake GP, Zeldin DC. Gender Differences in Murine Airway Responsiveness and Lipopolysaccharide-Induced Inflammation. *J Immunol*. 2006;177(1):621-630. doi:10.4049/jimmunol.177.1.621.
- 54. Alm AS, Li K, Yang D, Andersson R, Lu Y, Wang X. Varying susceptibility of pulmonary cytokine production to lipopolysaccharide in mice. *Cytokine*. 2010;49(3):256-263. doi:10.1016/j.cyto.2009.11.007.
- 55. Santos-Silva MA, Pires KMP, Trajano ETL, Martins V, Nesi RT, Benjamin CF, Caetano MS, Sternberg C, Machado MN, Zin WA, Valença SS, Porto LC. Redox Imbalance and Pulmonary Function in Bleomycin-Induced Fibrosis in C57BL/6, DBA/2, and BALB/c Mice. *Toxicol Pathol*. 2012;40(5):731-741. doi:10.1177/0192623312441404.
- 56. Schrier DJ, Kunkel RG, Phan SH. The Role of Strain Variation in Murine Bleomycin-Induced Pulmonary Fibrosis. *Am Rev Respir Dis*. 2015;127(1):63-66. doi:10.1164/arrd.1983.127.1.63.
- 57. Hoyt DG, Lazo JS. Murine strain differences in acute lung injury and activation of poly(ADP-ribose) polymerase by in vitro exposure of lung slices to bleomycin. *Am J Respir Cell Mol Biol*. 1992;7(6):645-651. doi:10.1165/ajrcmb/7.6.645.
- Lasky JA, Ortiz LA, Tonthat B, Hoyle GW, Corti M, Athas G, Lungarella G, Brody A, Friedman M. Connective tissue growth factor mRNA expression is upregulated in bleomycin-induced lung fibrosis. Am J Physiol Cell Mol Physiol. 2017;275(2):L365-L371. doi:10.1152/ajplung.1998.275.2.l365.
- 59. Baecher-Allan CM, Barth RK. PCR analysis of cytokine induction profiles associated with mouse strain variation in susceptibility to pulmonary fibrosis. *Reg Immunol*. 1993;5(3-4):207-20717.
- 60. Bonniaud P, Martin G, Margetts PJ, Ask K, Robertson J, Gauldie J, Kolb M. Connective tissue growth factor is crucial to inducing a profibrotic environment in "fibrosis-resistant" Balb/c mouse lungs. *Am J Respir Cell Mol Biol*. 2004;31(5):510-516. doi:10.1165/rcmb.2004-01580C.
- 61. Van Hove C, Paulissen G, Noël A, Foidart J-M, Cataldo DD, Quesada-Calvo F, Louis R, Tournoy K, Gueders MM, Crahay C, Hacha J. Mouse models of asthma: a comparison between C57BL/6 and BALB/c strains regarding bronchial responsiveness, inflammation, and cytokine production. *Inflamm*

Res. 2009;58(12):845-854. doi:10.1007/s00011-009-0054-2.

- 62. Arenas M, Gil F, Gironella M, Hernández V, Biete A, Piqué JM, Panés J. Time course of antiinflammatory effect of low-dose radiotherapy: Correlation with TGF-β1 expression. *Radiother Oncol.* 2008;86(3):399-406. doi:10.1016/j.radonc.2007.10.032.
- 63. Arenas M, Gil F, Gironella M, Hernández V, Jorcano S, Biete A, Piqué JM, Panés J. Anti-inflammatory effects of low-dose radiotherapy in an experimental model of systemic inflammation in mice. *Int J Radiat Oncol Biol Phys.* 2006;66(2):560-567. doi:10.1016/j.ijrobp.2006.06.004.
- 64. Wunderlich R, Ernst A, Rödel F, Fietkau R, Ott O, Lauber K, Frey B, Gaipl US. Low and moderate doses of ionizing radiation up to 2 Gy modulate transmigration and chemotaxis of activated macrophages, provoke an anti-inflammatory cytokine milieu, but do not impact upon viability and phagocytic function. *Clin Exp Immunol.* 2015;179(1):50-61. doi:10.1111/cei.12344.
- Frischholz B, Wunderlich R, Rühle P-F, Schorn C, Rödel F, Keilholz L, Fietkau R, Gaipl US, Frey B. Reduced secretion of the inflammatory cytokine IL-1β by stimulated peritoneal macrophages of radiosensitive Balb/c mice after exposure to 0.5 or 0.7Gy of ionizing radiation. *Autoimmunity*. 2013;46(5):323-328. doi:10.3109/08916934.2012.747522.
- 66. Nakatsukasa H, Tsukimoto M, Tokunaga A, Kojima S. Repeated Gamma Irradiation Attenuates Collagen-Induced Arthritis via Up-regulation of Regulatory T Cells but not by Damaging Lymphocytes Directly. *Radiat Res.* 2010;174(3):313-324. doi:10.1667/RR2121.1.
- 67. Kern PM, Keilholz L, Forster C, Hallmann R, Herrmann M, Seegenschmiedt M-H. Low-dose radiotherapy selectively reduces adhesion of peripheral blood mononuclear cells to endothelium in vitro. *Radiother Oncol.* 2000;54(3):273-282. doi:10.1016/S0167-8140(00)00141-9.
- 68. Hildebrandt G, Maggiorella L, Rödel F, Rödel V, Willis D, Trott K-R. Mononuclear cell adhesion and cell adhesion molecule liberation after X-irradiation of activated endothelial cells in vitro. *Int J Radiat Biol.* 2002;78(4):315-325. doi:10.1080/09553000110106027.
- 69. Hackam DG, Redelmeier DA. Translation of research evidence from animals to humans. *J Am Med Assoc*. 2006;296(14):1731-1732. doi:10.1001/jama.296.14.1731.
- 70. U.S. Food and Drug Administration. *Challenge and Opportunity on the Critical Path to New Medical Products.*; 2004.
- 71. Hackam DG. Translating animal research into clinical benefit. *Br Med J*. 2007;334(7586):163-164. doi:10.1136/bmj.39104.362951.80.
- 72. Perel P, Roberts I, Sena E, Wheble P, Briscoe C, Sandercock P, Macleod M, Mignini LE, Jayaram P, Khan KS. Comparison of treatment effects between animal experiments and clinical trials: Systematic review. *Br Med J*. 2007;334(7586):197-200. doi:10.1136/bmj.39048.407928.BE.
- 73. Matute-Bello G, Frevert CW, Martin TR. Animal models of acute lung injury. *Am J Physiol Cell Mol Physiol*. 2008;295(3):L379-L399. doi:10.1152/ajplung.00010.2008.
- 74. Bart van der Worp H, Howells DW, Sena ES, Porritt MJ, Rewell S, O'Collins V, Macleod MR. Can animal models of disease reliably inform human studies? *PLoS Med.* 2010;7(3):1-8. doi:10.1371/journal.pmed.1000245.
- 75. Shay T, Jojic V, Zuk O, Rothamel K, Puyraimond-Zemmour D, Feng T, Wakamatsu E, Benoist C, Koller D, Regev A. Conservation and divergence in the transcriptional programs of the human and mouse immune systems. *Proc Natl Acad Sci*. 2013;110(8):2946-2951. doi:10.1073/pnas.1222738110.
- 76. Mestas J, Hughes CCW. Of Mice and Not Men: Differences between Mouse and Human Immunology. *J Immunol*. 2004;172(5):2731-2738. doi:10.4049/jimmunol.172.5.2731.
- 77. Suarez CJ, Dintzis SM, Frevert CW. Respiratory. In: Treuting PM, Dintzis SM, eds. Comparative

Anatomy and Histology. San Diego: Academic Press; 2012:121-134. doi:https://doi.org/10.1016/B978-0-12-381361-9.00009-3.

- 78. Mole RH. The LD50 for uniform low LET irradiation of man. *Br J Radiol*. 1984;57(677):355-369. doi:10.1259/0007-1285-57-677-355.
- 79. Mettler FA, Huda W, Yoshizumi TT, Mahesh M. Effective doses in radiology and diagnostic nuclear medicine: a catalog. *Radiology*. 2008;248(1):254-263. doi:10.1148/radiol.2481071451.
- 80. Australian Institute of Health and Welfare. Australia's Health 2018, Report No 16. Canberra; 2018.
- 81. Hossain M, Chetana M, Uma Devi P. Late effect of prenatal irradiation on the hippocampal histology and brain weight in adult mice. *Int J Dev Neurosci*. 2005;23(4):307-313. doi:10.1016/j.ijdevneu.2005.01.006.
- 82. Hossain M, Uma Devi P. Effect of irradiation at the early foetal stage on adult brain function of mouse: Learning and memory. *Int J Radiat Biol.* 2001;77(5):581-585. doi:10.1080/09553000110037394.
- 83. Hossain M, Uma Devi P. Effect of irradiation at the early fetal stage on adult brain function in the mouse: locomotor activity. *Int J Radiat Biol.* 2000;76(10):1397-1402. doi:10.1080/09553000050151673.
- 84. Minamisawa T, Hirokaga K. Long-Term Changes in Open Field Activity of Male Mice Irradiated with Low Levels of Gamma Rays at Late Stage of Development. *J Radiat Res.* 1996;37(2):117-124. doi:10.1269/jrr.37.117.
- 85. Jensh RP, Brent RL. The Effects of Prenatal X Irradiation on the Appearance of Reflexes and Physiologic Markers in the Neonatal Rat. *Radiat Res.* 1988;116(3):416. doi:10.2307/3577385.
- 86. Minamisawa T, Hirokaga K. Long-Term Effects of Prenatal Exposure to Low Levels of Gamma Rays on Open-Field Activity in Male Mice. *Radiat Res.* 1995;144(2):237. doi:10.2307/3579264.
- 87. Baskar R, Uma Devi P. Influence of gestational age to low-level gamma irradiation on postnatal behavior in mice. *Neurotoxicol Teratol*. 2000;22(4):593-602. doi:10.1016/S0892-0362(00)00076-3.
- Jensh RP, Brent RL. Effects of 0.6-Gy Prenatal X Irradiation on Postnatal Neurophysiologic Development in the Wistar Rat. *Exp Biol Med.* 1986;181(4):611-619. doi:10.3181/00379727-181-42299.
- 89. Nash DJ. Effects of Prenatal X-Irradiation on Development and Postnatal Viability of Inbred and Hybrid Mice. *Biol Bull*. 1971;140(2):230-241. doi:10.2307/1540071.
- 90. Minamisawa T, Hirokaga K, Sasaki S. Gross morphological changes of the mouse brain exposed prenatally to ionizing radiation. *J Radiat Res.* 2008;31(2):214-218. doi:10.1269/jrr.31.214.
- 91. Kokošová N, Tomášová L, Kisková T, Šmajda B. Neuronal Analysis and Behaviour in Prenatally Gamma-Irradiated Rats. *Cell Mol Neurobiol*. 2014;35(1):45-55. doi:10.1007/s10571-014-0144-8.
- 92. Tomášová L, Šmajda B, Ševc J. Effects of prenatal irradiation on behaviour and hippocampal neurogenesis in adult rats. *Acta Physiol Hung*. 2012;99(2):126-132. doi:10.1556/APhysiol.99.2012.2.5.
- 93. Sienkiewicz ZJ, Haylock RGE, Saunders RD. Differential learning impairments produced by prenatal exposure to ionizing radiation in mice. *Int J Radiat Biol.* 1999;75(1):121-127. doi:10.1080/095530099140889.
- 94. Otake M, Schull WJ, Lee S. Threshold for radiation-related severe mental retardation in prenatally exposed A-bomb survivors: A re-analysis. *Int J Radiat Biol.* 1996;70(6):755-763. doi:10.1080/095530096144644.

- 95. Otake M, Schull WJ. Radiation-related small head sizes among prenatally exposed a-bomb survivors. Int J Radiat Biol. 1993;63(2):255-270. doi:10.1080/09553009314550341.
- 96. Nakashima E. Relationship of five anthropometric measurements at age 18 to radiation dose among atomic bomb survivors exposed in utero. *Radiat Res.* 1994;138(1):121-126.
- 97. Otake M, Fujikoshi Y, Schull WJ, Izumi S. A Longitudinal Study of Growth and Development of Stature among Prenatally Exposed Atomic Bomb Survivors. *Radiat Res.* 1993;134(1):94. doi:10.2307/3578506.
- 98. American College of Radiologists. *Practice Parameter for Imaging Pregnant or Potentially Pregnant Adolescents and Women With Ionizing Radiation*. Vol 1076.; 2018.
- 99. Royal College of Radiologists and the College of Radiographers. *Protection of Pregnant Patients during Diagnostic Medical Exposures to Ionising Radiation.*; 2009.
- 100. Royal Australian and New Zealand College of Radiologists. *Diagnostic Radiology and Pregnancy. RANZCR Position Statement.*; 2017.
- 101. American College of Obstetricians and Gynecologists. *Guidelines for Diagnostic Imaging during Pregnancy and Lactation*. Vol 130.; 2017. doi:10.1097/AOG.00000000002355.
- 102. The International Commission on Radiological Protection. *Pregnancy and Medical Radiation. ICRP Publication 84.* Vol 30. Published for the International Commission on Radiological Protection by Pergamon; 2000.
- 103. Preston DL, Cullings H, Suyama A, Funamoto S, Nishi N, Soda M, Mabuchi K, Kodama K, Kasagi F, Shore RE. Solid cancer incidence in atomic bomb survivors exposed in utero or as young children. *J Natl Cancer Inst.* 2008;100(6):428-436. doi:10.1093/jnci/djn045.
- 104. Rajaraman P, Simpson J, Neta G, De Gonzalez AB, Ansell P, Linet MS, Ron E, Roman E. Early life exposure to diagnostic radiation and ultrasound scans and risk of childhood cancer: Case-control study. *BMJ*. 2011;342(7794):424. doi:10.1136/bmj.d472.
- 105. Wakeford R, Little MP. Risk coefficients for childhood cancer after intrauterine irradiation: A review. *Int J Radiat Biol.* 2003;79(5):293-309. doi:10.1080/0955300031000114729.
- 106. Naumburg E, Bellocco R, Cnattingius S, Hall P, Boice JD, Ekbom A. Intrauterine Exposure to Diagnostic X Rays and Risk of Childhood Leukemia Subtypes. *Radiat Res.* 2001;156(6):718-723. doi:10.1667/0033-7587(2001)156[0718:ietdxr]2.0.co;2.
- 107. Delongchamp RR, Mabuchi K, Yoshimoto Y, Preston DL. Cancer mortality among atomic bomb survivors exposed in utero or as young children, October 1950-May 1992. *Radiat Res.* 1997;147(3):385-395.
- 108. Rodvall Y, Pershagen G, Hrubec Z, Ahlbom A, Pedersen NL, Boice JD. Prenatal x-ray exposure and childhood cancer in swedish twins. *Int J Cancer*. 1990;46(3):362-365. doi:10.1002/ijc.2910460304.
- 109. Linet MS, Slovis TL, Miller DL, Kleinerman R, Lee C, Rajaraman P, Berrington de Gonzalez A. Cancer risks associated with external radiation from diagnostic imaging procedures. *CA Cancer J Clin.* 2012;62(2):75-100. doi:10.3322/caac.21132.
- 110. Boice JD, Miller RW. Childhood and adult cancer after intrauterine exposure to ionizing radiation. In: *Teratology*. Vol 59. ; 1999:227-233. doi:10.1002/(SICI)1096-9926(199904)59:4<227::AID-TERA7>3.0.CO;2-E.
- 111. The International Commission on Radiological Protection. The 2007 Recommendations of the International Commission on Radiological Protection. ICRP Publication 103. *Ann ICRP*. 2007;37(2-4). doi:10.1177/ANIB_37_2-4.

- 112. Ohtaki K, Kodama Y, Nakano M, Itoh M, Awa AA, Cologne J, Nakamura N. Human fetuses do not register chromosome damage inflicted by radiation exposure in lymphoid precursor cells except for a small but significant effect at low doses. *Radiat Res.* 2004;161(4):373-379. doi:10.1667/3147.
- 113. Nakano M, Kodama Y, Ohtaki K, Nakashima E, Niwa O, Toyoshima M, Nakamura N. Chromosome aberrations do not persist in the lymphocytes or bone marrow cells of mice irradiated in utero or soon after birth. *Radiat Res.* 2007;167(6):693-702. doi:10.1667/RR0718.1.
- 114. Otis EM, Brent R. Equivalent ages in mouse and human embryos. *Anat Rec.* 1954;120(1):33-63. doi:10.1002/ar.1091200104.
- 115. Krishnan A, Samtani R, Dhanantwari P, Lee E, Yamada S, Shiota K, Donofrio MT, Leatherbury L, Lo CW. A detailed comparison of mouse and human cardiac development. *Pediatr Res.* 2014;76(6):500-507. doi:10.1038/pr.2014.128.
- 116. Honda S, Shibata Y, Mine M, Imamura Y, Tagawa M, Nakane Y, Tomonaga M. Mental health conditions among atomic bomb survivors in Nagasaki. *Psychiatry Clin Neurosci*. 2002;56(5):575-583. doi:10.1046/j.1440-1819.2002.01057.x.
- 117. Yamada M, Izumi S. Psychiatric sequelae in atomic bomb survivors in Hiroshima and Nagasaki two decades after the explosions. *Soc Psychiatry Psychiatr Epidemiol*. 2002;37(9):409-415. doi:10.1007/s00127-002-0572-5.
- 118. Totter JR, MacPherson HG. Do childhood cancers result from prenatal x-rays? *Health Phys.* 1981;40(4):511-524.
- 119. Bosanquet DC, Green G, Bosanquet AJ, Galland RB, Gower-Thomas K, Lewis MH. Doctors' knowledge of radiation A two-centre study and historical comparison. *Clin Radiol*. 2011;66(8):748-751. doi:10.1016/j.crad.2011.03.009.
- 120. Heyer C, Peters S, Lemburg S, Nicolas V. Einschätzung der Strahlenbelastung radiologischer Thorax-Verfahren: Was ist Nichtradiologen bekannt? *RöFo - Fortschritte auf dem Gebiet der Röntgenstrahlen und der Bildgeb Verfahren*. 2007;179(3):261-267. doi:10.1055/s-2006-927366.
- 121. Rice HE, Frush DP, Harker MJ, Farmer D, Waldhausen JH, APSA Education Committee. Peer assessment of pediatric surgeons for potential risks of radiation exposure from computed tomography scans. *J Pediatr Surg*. 2007;42(7):1157-1164. doi:10.1016/j.jpedsurg.2007.02.007.
- 122. Soye JA, Paterson A. A survey of awareness of radiation dose among health professionals in Northern Ireland. *Br J Radiol*. 2008;81(969):725-729. doi:10.1259/bjr/94101717.
- 123. Lin EC. Radiation Risk From Medical Imaging. *Mayo Clin Proc.* 2010;85(12):1142-1146. doi:10.4065/mcp.2010.0260.
- 124. Smith-Bindman R, Aubin C, Bailitz J, Bengiamin RN, Camargo CA, Corbo J, Dean AJ, Goldstein RB, Griffey RT, Jay GD, Kang TL, Kriesel DR, Ma OJ, Mallin M, Manson W, Melnikow J, Miglioretti DL, Miller SK, Mills LD, Miner JR, Moghadassi M, Noble VE, Press GM, Stoller ML, Valencia VE, Wang J, Wang RC, Cummings SR. Ultrasonography versus Computed Tomography for Suspected Nephrolithiasis. N Engl J Med. 2014;371(12):1100-1110. doi:10.1056/NEJMoa1404446.
- 125. Kidwell CS, Chalela JA, Saver JL, Starkman S, Hill MD, Demchuk AM, Butman JA, Patronas N, Alger JR, Latour LL, Luby ML, Baird AE, Leary MC, Tremwel M, Ovbiagele B, Fredieu A, Suzuki S, Villabianca JP, Davis S, Dunn B, Todd JW, Ezzeddine MA, Haymore J, Lynch JK, Davis L, Warach S. Comparison of MRI and CT for detection of acute intracerebral hemorrhage. *J Am Med Assoc*. 2004;292(15):1823-1830. doi:10.1001/jama.292.15.1823.
- 126. Perri A, Riccardi R, Iannotta R, Di Molfetta D V., Arena R, Vento G, Zecca E. Lung ultrasonography score versus chest X-ray score to predict surfactant administration in newborns with respiratory distress syndrome. *Pediatr Pulmonol*. 2018;53(9):1231-1236. doi:10.1002/ppul.24076.

- 127. Thurlbeck WM. Postnatal human lung growth. *Thorax*. 1982;37(8):564-571. doi:10.1136/thx.37.8.564.
- 128. Herring MJ, Putney LF, Wyatt G, Finkbeiner WE, Hyde DM. Growth of alveoli during postnatal development in humans based on stereological estimation. *Am J Physiol Cell Mol Physiol*. 2014;307(4):L338-L344. doi:10.1152/ajplung.00094.2014.
- 129. Narayanan M, Owers-Bradley J, Beardsmore CS, Mada M, Ball I, Garipov R, Panesar KS, Kuehni CE, Spycher BD, Williams SE, Silverman M. Alveolarization continues during childhood and adolescence: New evidence from helium-3 magnetic resonance. *Am J Respir Crit Care Med*. 2012;185(2):186-191. doi:10.1164/rccm.201107-1348OC.
- 130. Berry JD, Lloyd-Jones DM, Van Horn L, Garside DB, Tracy RP, Thomas A, Cai X, Ning H, Greenland P, Dyer A. Lifetime Risks of Cardiovascular Disease. *N Engl J Med.* 2012;366(4):321-329. doi:10.1056/nejmoa1012848.

CHAPTER 7

COTUTELLE PROGRAM EXPERIENCE

A Cotutelle De Theses is a doctoral degree program where a PhD candidate undertakes their studies jointly between two universities. The candidate is jointly supervised from both institutions. The candidate must complete the requirements for both universities, including coursework, yearly milestones and examination processes, and then is conferred a doctorate from both institutions.

7.1 UNIVERSITY INFORMATION

| Home (primary) Institution | Host (secondary) Institution |
|--|---|
| College of Medicine and Public Health (formally Department of Critical Care Medicine, faculty of MNHS) | Department of Biology, Faculty of Science |
| Flinders University, Sturt Rd, Bedford Park, SA, 5048, Australia | McMaster University 1280 Main St W, Hamilton, ON, L8S 4L8, Canada |
| Co-Supervisors | |
| Associate Professor Dani-Louise Dixon BSc, PhD. | Professor Joanna Wilson, BSc, MSc, PhD |
| Supervisory committee | |
| Associate Professor Antony Hooker BBiotech, PhD | Professor Douglas Boreham, BSc, PhD |
| Associate Professor Shailesh Bihari MD, PhD | Professor Grant McClelland, BSc, PhD |
| | 207 |

Study Term

Flinders University: Feb 2016 – May 2017 McMaster University: June 2017 – May 2018 Flinders University: June 2018 – July 2019

7.2 COTUTELLE AGREEMENT

To begin this process, a collaboration was established with Prof. Joanna Wilson and Prof. Douglas Boreham, and McMaster University. A Memorandum of Understanding was negotiated between Flinders University and McMaster University. It was approximately 8 months from first contact with Flinders University Cotutelle liaison until the signing of the final Memorandum of Understanding by both parties, thus my experience in the application process was quite lengthy. After the agreement, enrollment into the McMaster Biology Doctoral Program was completed and the Cotutelle De Thesis PhD was established.

Some main points from the Memorandum of Understanding:

- The schedule of research work will be done in alternating periods at Flinders University and McMaster University with a total of one (1) calendar year completed at the host institute (McMaster University).
- The candidate shall be concurrently enrolled at both institutions for the period of their candidature but shall be exempt from tuition fees at the host institute.
- Any intellectual property rights, data, and information resulting from research activities conducted under this agreement on academic cooperation shall be

jointly owned by the two parties and both parties shall be allowed to use such property.

• The candidate shall be required to fulfil all academic requirements of each institution including required coursework, department graduate workshops, comprehensive examinations, yearly milestones, thesis defence and any other requirements. The thesis must include a separate chapter describing the Cotutelle experience and detail the value added to the degree.

7.3 COTUTELLE EXPERIENCE

7.3.1 Visa and travel

After enrolment was confirmed at both institutions, the next process was gaining the student visa and required documentation to study in Canada. This process was rather straightforward with generous help from the international departments of both institutions. Flinders University also generously supports its Cotutelle students with a Cotutelle travel grant which aided in the travel to Canada.

7.3.2 Learning from a different lab

The Critical Care Lab has been a great source of support, development and emotional release throughout this doctoral journey. This relatively small, close-knit group have pushed me in many professional and personal achievements. The first year with this group changed my communication skills immensely. My PhD experience began with experiments and learning the necessary practical skills however my communication skills developed passively. The lack of awareness around my communication skills was quickly highlighted by this group which led to the development and refinement of intentionality, clarity and expression. In addition, the practical skills learned from this lab enabled me to become the 'expert' of many techniques during my time abroad; which is a powerful but sometimes frustrating position to be in. I gained clinical experience through my audit with patient data and outcomes being the principal results for my 1st and 2nd data chapters. I also learned extensive animal handling and experimental skills through the lab's rodent models which was brought to McMaster University for my 3rd and 4th data chapters. In addition to the physical skills learned, the workup prior to the methods, including the ethics and governance applications, was a smooth process as this lab has dealt with both clinical and animal studies in the past. Without the lab's expertise in both the clinical and animal model studies, I would have had a much harder time and longer process getting these data chapters to completion.

The Wilson lab was very accommodating and inclusive throughout the year in Canada. However, this lab is very different to the Critical Care lab at Flinders University, which allowed for professional and personal growth in many areas. When I entered, the Wilson lab encompassed 21 people from 2nd year undergrads to graduate researchers (Masters and PhD), Research Assistants, Postdoctoral researchers and Joanna herself. This size created a larger scope of academics/researchers than I had been exposed to previously therefore allowing development of communication and group engagement skills. Switching interactions between people from different levels of scientific backgrounds changed: the structure of support, such as receiving support from post docs and giving support to undergrads; and the terminology of communication, including having intellectual debates with experts and giving explanations to lay researchers. In addition, many strong friendships were created with this group that still exist to this date. Overall, this PhD would not have been completed if not from the valuable lessons learned at both universities, both: practical and passive; and professional and personal.

7.3.3 Implications on scientific learning

As mentioned before, the first year, under Assoc. Prof. Dani-Louise Dixon, taught me everything I needed to know to begin my 1st and 2nd data chapter. The clinical attachment of the lab meant that the research completed in the lab could be seen by clinicians and aid in the translational aspect of the research. Having clinician Assoc. Prof. Shailesh Bihari easily assessable allowed for a clearer understanding of what the clinical data meant and thus a better understanding of the results. Assoc. Prof. Antony (Tony) Hooker provided the radiation expertise which implicated all data chapters in this thesis. Discussions with Tony provided clarity on dosimetry in the clinical work and radiobiological responses in the animal work.

A main reason for collaborating with Assoc. Prof. Joanna Wilson and Prof. Douglas Boreham was the animal model they have maintained. This animal model would have taken a significant amount of time to prepare at Flinders University, thus saving research time and costs. Along with this model, a significant variety of skills were present surrounding animal handling/techniques and this model, which have now been learnt. However, the experimental outcomes measurable by the techniques available in the Wilson Lab lacked some desirable interests. Unfortunately, there was no ability to capture functional changes in the lungs of the animals, including respiratory function and gas exchange analysis, which is something obtainable back in the Critical Care Lab. Overall I am content with the science produced in this PhD and would not have been able to

produce these experiments and advance my scientific knowledge without participating in this exchange.

7.3.4 Co-Supervision

It was valuable to see how different supervisors manage their students. I was fortunate to have two extremely helpful supervisors that have different managerial styles. From the co-supervision, I was able to experience times of continuous support vs times of independence, times when I was communicating to the expert vs communicating as the expert, and times of variation in perspectives on ideas, that particularly helped in interpretation of results. I believe that I received the appropriate level of support and guidance from both supervisors in my PhD at the times where I needed them the most. Dani began this PhD experience as the expert who aided in the development of my project and closely observed my progression. Joanna then gave me the space to become the expert and allowed self-development and growth into an independent researcher. Both aided in the final year of this PhD, defining and improving articulation of the final thesis.

7.3.5 Final statement

The Cotutelle program was an interesting and particularly rewarding experience of professional development and personal refinement. In addition to the obvious pros of living abroad in a new community and the experiences attached to a new routine, this thesis and my professional and personal development is solely a product of the scientific, educational format and supervisory differences created through this exchange. I would urge any PhD student who is looking to forward their growth and a variety of skills sets to investigate the Cotutelle program within their university.

THE END